University of South Carolina Scholar Commons

Theses and Dissertations

8-9-2014

How Much Do Groups Still Matter to Politics? An Examination of Group Influences on Public Opinion

Lauren E. Smith *University of South Carolina*

Follow this and additional works at: http://scholarcommons.sc.edu/etd

Recommended Citation

Smith, L. E. (2014). *How Much Do Groups Still Matter to Politics? An Examination of Group Influences on Public Opinion.* (Doctoral dissertation). Retrieved from http://scholarcommons.sc.edu/etd/2913

This Open Access Dissertation is brought to you for free and open access by Scholar Commons. It has been accepted for inclusion in Theses and Dissertations by an authorized administrator of Scholar Commons. For more information, please contact SCHOLARC@mailbox.sc.edu.

How Much Do Groups Still Matter to Politics? An Examination of Group Influences on Public Opinion

by

Lauren E. Smith

Bachelor of Science Clemson University, 2008

Master of Arts University of South Carolina, 2013

Submitted in Partial Fulfillment of the Requirements

For the Degree of Doctor of Philosophy in

Political Science

College of Arts and Sciences

University of South Carolina

2014

Accepted by:

David Darmofal, Major Professor

Charles Finocchiaro, Committee Member

Laura Olson, Committee Member

Lee Walker, Committee Member

Lacy Ford, Vice Provost and Dean of Graduate Studies

© Copyright by Lauren E. Smith, 2014 All Rights Reserved.

Acknowledgements

I would like to express my thanks to David Darmofal, Chuck Finocchiaro, Laura Olson and Lee Walker for guiding me through graduate school, from talks to classes to published papers. They made it the best possible training it could have been. I would also like to thank them for forming my dissertation committee and being willing to work with me to complete the dissertation.

I would also like to thank my family, especially my husband Ryan Smith, for always being behind me 100% and for many long discussions about political science which I know made no sense to them but they listened anyway.

Abstract

Public opinion plays an important role in democracy. A system of government designed specifically to be by the people, for the people and of the people must by necessity listen to the opinions of the people. Accordingly, an important research agenda is determining conditions under which public opinion is listened to and translated into government responsiveness. Most of the public opinion literature answering this question focuses specifically on individual opinion. I argue that this is problematic because politics is ultimately carried out in terms of the collective. Further, I argue that collective opinion is often voiced through groups in society such that groups are an important and often overlooked mediator of public opinion. I present a model of group influences on public opinion, arguing in three parts that groups first shape individual opinion through social identity effects or the desire of individuals to feel connected to others, government is theoretically likely to listen to groups rather than individual or overall opinion because politics is ultimately carried out in terms of the aggregate, but government is only likely to listen to group opinion if the group holds intense preferences and can therefore signal their opinion to individual group members and to government. I test this theory in three separate cases where public opinion is evidenced, political parties, state Supreme Court decision making and ballot initiatives. I find support for my theory in two of the three cases, political parties and state Supreme Court decision making. Overall, I demonstrate the continued importance and role of groups in American politics and to public opinion in particular and show the necessity of testing the breadth and depth of theory.

iv

Table of Contents

Acknowledgements	iii
Abstract	iv
List of Tables	vi
Chapter 1 Introduction	1
Chapter 2: Theory	10
Chapter 3: Partisan Strength and Religious Commitment	
Chapter 4: Judicial Decision Making in state Supreme Courts: Do Justices Respond to Religious Opinion?	61
Chapter 5: Partisan Effects of Ballot Propositions	90
Chapter 6: Conclusion	115
References	130

List of Tables

Table 3.1. Logistic Regression of the Effects of Religious Attachment on Partisan Strength 59
Table 3.2. Predicted Probabilities for the Logistic Regression of theEffects of Religious Attachment on Partisan Strength
Table 4.1. Descriptive Statistics, Method of Judicial Appointment and Death Penalty Cases and Mean Percentage of Religious Groups in the States
Table 4.2. GEE Logistic Regression of the Effects of Religion on State Supreme Courts Death Penalty Decisions
Table 4.3. Marginal Effects for the GEE Logistic Regression of theEffects of Religion on State Supreme CourtsDeath Penalty Decisions89
Table 5.1. Ordinary Least Squares Regression Analysis, of Gay MarriageAmendment States on Turnout in 2004 and Bush Vote Share in 2004111
Table 5.2. Ordinary Least Squares Regression Analysis, Effect of GayMarriage Amendment States and Religion on Turnout in 2004
Table 5.3. Ordinary Least Squares Regression Analysis, of Gay MarriageAmendment States on Bush Vote Share in 2004 and Religion
Table 5.4. Ordinary Least Squares Regression Analysis, of Gay MarriageAmendment States and Religion on Republican Seat Change in StateLegislatures Lower House114

Chapter 1

Introduction

V.O. Key famously defined public opinion as "those opinions held by private persons which government finds it prudent to heed" (1961, 1). Democracy commonly understood as government by the people and for the people, arguably should rest on the opinions of the people. As such, democracy is considered the best form of government since it does give voice to the otherwise voiceless. However, at the same time, democracy is also often considered the most tenuous form of government, and at the extreme, the system of government least likely to succeed. Because it is a system designed specifically to listen to and follow the voice of the people, the ancient Greeks down through the Founding Fathers feared the potential result of misdirected and misguided decision making by a populous unqualified and in many ways unable (in their minds) to make good judgments. They feared the end result would be chaos through mob rule.

To prevent the potential for mob rule, the Founding Fathers compromised by creating not a pure democracy but a democratic republic (Wood 1969, 1991). Rather than literal rule by the people, the Founders crafted a system whereby the people elected leaders to represent them and their opinion in government. Therefore, the people were given at best an indirect voice in government. Accordingly, a key question that faced the Founding Fathers became how much these elected leaders, collectively making up the government, should listen and adhere to public opinion. Are leaders to act as delegates,

consciously taking into consideration the opinions of their constituents, or trustees, free to act as they see fit with little to no regard for the actual opinions of the people?

1.1 Public Opinion: Two Views

At the root of this debate lay two opposing views on public opinion. Among the Founders Hamilton best embodies the skeptic view, arguing that the people are capricious with little knowledge, political in particular, and as a result their opinion should not form the basis of judgment for government. As he puts it, "the people are turbulent and changing; they seldom judge or determine right," though they have been said to have the voice of God "it is not true in fact" (Frisch 1985,108). This view finds substantial support in the literature beginning with Lippmann (1922) and Schumpter (1942) who wrote despairingly of there being any wisdom, relating to governing matters in particular, in the people and consequently their influence on policy should be limited. Beginning with the first public opinion polls, empirical findings supported this claim that the public is ignorant and uninformed about politics providing a basis of support for this skeptical view of public opinion (Campbell et al. 1960; Converse 1964; Butler and Stokes 1969; Luskin 1987, 1990; see more recently Kuklunski et al. 2000 and misinformation). As Delli Carpini and Keeter (1996) note, one scholar goes so far as to say that "America's embarrassing little secret...is that vast numbers of Americans are ignorant" and argues that this lack of basic knowledge about politics undermines the central tenet of democratic government, rule by the people (Blumberg 1990, 1).

However, on the other side of the debate are the optimists who, though acknowledging that citizens are often uninformed, argue that democracy ultimately rests on the will of the people and so listening to and following their opinions is vitally

important especially in maintaining democratic legitimacy. Jefferson best embodies this view among the Founders. For example, in speaking with a friend he argues that he is willing to "freely leave to self government" the masses of the United States (keeping in mind that meant white males) (Padover 1939, 56). Jefferson viewed citizens as more than capable of self-government and believed democracy rested on their input and opinions. Further, optimists argue that the lack of information and knowledge among the public is largely a result of external forces, resulting from defective social and political institutions rather than internal deficiencies (e.g. Delli Carpini and Keeter 1996) and as such these deficiencies can be overcome. They argue that there are systematic biases in America, such as inequality in socioeconomic status, that are a major cause of the ignorance of the populous. Or, in another view, V.O. Key (1961, 557) forcefully puts it "politicians often make of the public a scapegoat for their own shortcomings." He argued that citizens are not as ignorant as often perceived yet politicians can use the fact of the public's deficiencies to their own advantage. At the same time, some optimists also argue that a fully informed electorate is not needed since citizens are able to rely on 'shortcuts' to act as if they are fully informed (Lupia 1994; Lupia and McCubbins 1998; Lau and Redlawsk 2001). Overall, optimists argue that public opinion must be listened to for democracy to maintain its legitimacy, as best stated by V.O Key, (1961,10) "unless mass views have some place in the shaping of policy, all the talk about democracy is nonsense."

1.2 Research Question

Accordingly, though debated, it is clear that the importance of public opinion to democracy remains a vital topic of interest in political science. While of concern to

scholars from the early days of political science (for example Lippman 1922) the study of public opinion exploded with the advent of scientific polling and the consequent ability of scholars to actually measure the opinions of the public. For example, Gallup famously argued that public opinion polling would allow political leaders to find out public opinion and respond to it (Gallup 1938; Gallup and Rae 1940). This view only grew as the ability (both methodologically and financially) to conduct studies of the public increased (for example Campbell et al. 1960 and the subsequent Michigan national election studies).

Consequently, public opinion literature analyzes a wide variety of research questions including whether policy makers actually respond to public opinion (see Shapiro 2011 for a review), do citizens hold political belief systems and have coherent structure in their opinions (Converse 1964; Nie and Anderson 1974; Achen 1975; Luskin 1987), how informed is the citizenry and how informed does it need to be for democracy to survive (Delli Carpini and Keeter 1997; Jerit et al. 2006; Kuklinski, et al. 2000; Hutchings 2003 see the heuristics literature as well Lupia 1994; Lupia and McCubbins 1998; Lau and Redlawsk 2001; Darmofal 2005), how do citizens reach their opinions (Lodge et al. 1989; Zaller 1992), what is the role of the media in opinion formation (Bartels 1993; Mutz 1998; Prior 2007) and many more. In adding to this body of literature I seek to answer the question, what are some of the specific conditions under which public opinion is listened to and translated into government responsiveness. I offer a model of how this process works and test it in three distinct cases.

1.3 Atomistic Actor versus Interpersonal Relations

My research question is premised on a divide I see in the literature between studies that focus (intentional or not) on individuals as more or less atomistic actors in

their opinion and studies that focus on how group interactions affect individual opinion. This divide is evidenced from the beginning of the public opinion literature in the two schools of thought that drove most of the research on public opinion since the 1950s, the Columbia school (Lazarsfeld 1944; Berelson et al. 1954; Katz and Lazarsfeld 1955) and the Michigan school (Campbell et al. 1960).

The Columbia school came first and offered a sociological view of public opinion. These scholars argued that you can think of politics as you think of consumer products where citizen opinion is formed through 'marketing' by the media and other opinion leaders in society. Citizens form opinions based largely on interpersonal interactions with others and look to groups within society (through the opinion leaders of these groups) as a basis for forming judgments and making decisions. In contrast, the Michigan school more heavily emphasized the psychological aspect of opinion. These scholars focused primarily on the individual as in many ways an atomistic actor influenced in their opinion by psychological things such as socioeconomic status (the base of the funnel in their funnel of causality for voting behavior). Sociological factors played a role as well (it is often referred to as the socio-psychological model) but the emphasis was on the individual as in many ways a lone actor forming opinions and judgments.

As with the voting behavior literature, I argue that the majority of the literature on public opinion following these early studies focused on the Michigan school analysis of public opinion as centered on the psychological and considers the individual as more or less an atomistic actor. Because the advent of polling and scientific surveys allowed scholars to some extent 'see inside' people's minds for the first time, it opened a new

avenue of research centered on the individual. Scholars no longer had to rely on aggregate or case study data to form and test theories and hypotheses but could now bring their theories under the scrutiny of actual individual level data. Additionally, the work of Downs (1957) and the resultant rational choice literature also spurred on the focus on the individual and how they form and reach decisions (for example the decision to vote). This is not to say that sociological factors are entirely missing from this literature (see the early media literature for example: Bartels 1993; Iyengar and Kinder 1987). I simply argue that the majority of the literature approached the study of public opinion from a notion of the individual as an atomistic actor centered more heavily on psychological concerns.

However, this began to change in the literature as Huckfeldt (1979; 1983; 1984) and others (Kenny 1992; Huckfeldt et al. 1995; Huckfeldt and Sprague 1995; Huckfeldt et al. 1998; Mcclurg 2003; Huckfeldt, Johnson and Sprague 2004; Walsh 2004; Mutz and Mondak 2006) began reviving the Columbia school focus on the sociological aspects of public opinion. These scholars argued for the role interpersonal relations can have on public opinion and other areas of political behavior. Citizens are not lone actors but are heavily influenced by those around them in their thoughts and opinions.

1.4 Groups Influences

I seek to continue in the tradition of the Columbia school and this growing body of literature by focusing on the role of interpersonal relations, as mediated through groups, on public opinion, looking at the specific question of conditions under which public opinion is listened to by government and translated into government responsiveness. I argue that, while examining the individual is important, it is only

through collective or aggregate opinion that politics is truly carried out, in terms of influencing elected leaders decisions and consequent policy making.

Accordingly, I offer a theory of group influences on public opinion drawing off of the work of the importance of interpersonal relations as argued by Huckfeldt (1979; 1983; 1984) and others (Kenny 1992; Huckfeldt et al. 1995; Huckfeldt and Sprague 1995; Mcclurg 2003; Huckfeldt, Johnson and Sprague 2004; Walsh 2004; Mutz and Mondak 2006). While politics is ultimately carried out in terms of the aggregate, this 'collective' opinion I argue is mediated through groups within society. Citizens spend a great deal of their time associating either loosely or closely with groups (race, gender, religion, social clubs to name a few). I argue that these interpersonal relations form in many ways the basis for their opinions and judgments. Accordingly, one way to study the influence of public opinion on government is to focus on groups and the influence that they wield within society. However, I argue that group influence on public opinion is conditional. This process will only work when a specific condition is present: that is when a group feels intensely about an issue and consequently 'owns' the issue whereby group members are able to clearly perceive the stance of the group and it consequently informs their own opinion and government feels the intensity of the group preference such that they are likely to listen to it as well. Overall, I argue that government listens to group opinion within society when reaching their decisions.

To test this theory of group opinion as influencing governmental decision making, I choose to focus on religious groups in particular. As I will explain in more detail, religious groups I argue offer a good test case for the influence of group opinion on governmental decision making. I then analyze this theory in three distinct areas to show

the breath of the model: political parties, state Supreme Court decision making and ballot initiatives.

Accordingly, in chapter one I explain my theory of group influences on public opinion in detail. It is composed of three parts: social identity theory, or how citizens think in terms of groups, group influences on politics and intensity of preferences held by the group. I next offer an argument for why religious groups in particular present a good test case for my theory. Finally, I explain how my theory relates and will be tested in three distinct analyses, strength of partisan attachment, state Supreme Court decision making and direct democracy focusing on the use of ballot propositions as affecting candidate choice in an election.

Chapters three through five will present my individual analyses. Chapter three focuses on public opinion as viewed through partisanship and strength of partisanship in particular, asking the question, does strength of religious commitment lead to strength of partisan commitment. Chapter four examines public opinion through the lens of one branch of government, the judicial branch, looking at whether religious groups influence state Supreme Court justices in their decision making. Chapter five analyses public opinion as exhibited through direct democracy looking at whether the use of ballot propositions can prime groups in a state, focusing on religious groups, in both voter turnout and vote choice.

Finally, chapter six concludes my dissertation. I review my overall theory of group influences on public opinion and then analyze and evaluate how well it performed in each of the three individual analyses. I offer concluding thoughts on this project and

whether I was able to uncover specific conditions under which public opinion is listened to and translated into government responsiveness.

Chapter 2

Theory

Historically, in American politics, groups within society have been a focal point of scholarly investigation and are arguably central to the maintenance of American democracy. In his classic examination of the success of early American democracy Tocqueville theorized that one of the key reasons the 'experiment' of democracy in America succeeded was a result of the civic life of the American populous (Tocqueville [1835-40] 1969). He noticed that Americans were "joiners," belonging to all types of civic and political associations. As he studied these groups within society he also began to observe that otherwise nonpolitical associations (for example lions clubs) often became mobilized and involved in politics at least at the local level. This greatly surprised Tocqueville as this was not common practice in the European context to which he was accustomed.

Many scholars since Tocqueville also argue for the importance of groups to American democracy. Perhaps the most extended treatment is done by Putnam (1995; 2000, 19) who argues that social capital, defined as "connections among individuals social networks and the norms of reciprocity and trustworthiness that arise from them," is essential for the health of democracy. Social capital enables citizens to come together to work on community concerns as well as engage in the political system. According to Putnam greater amounts of social capital produce an increase in democratic practices.

The decline he sees in group interactions, in the number of 'joiners' in American society is consequently extremely troubling to scholars such as Putnam.

From this observation of the number and importance of groups to American society, early scholars of political science found that even nominal membership in a group carried with it significant political meaning (e.g. Campbell et al. 1960). Simply being a Catholic or a union member or a white southerner (to name only a few) gave scholars clear direction as to their party identification and their political leanings. Campbell et al. (1960) sought to formalize this phenomenon through group reference theory, arguing that citizens look to groups to form their opinions and consequently their political decision making such as voting behavior.

However, with the advent of survey data and the capability (e.g. computers and statistical software) and methodology to carry out and conduct increasingly complex statistical analysis on this newly collected individual level data, the study and importance of groups to American politics declined. At least three things occurred. As I argued earlier, the focus of scholars of public opinion and other subfields became increasingly focused on the individual through the analysis of individual level data. However, at the same time, scholars now had the increasing capability to investigate and analyze beyond the group effect to seek out the causal mechanisms at work in causing these group level effects. Is simply being a white southerner the cause or the mechanism behind Democratic identification or is something else going on? Questions such as these became suddenly accessible to answer from an empirical standpoint. Additionally, American society went through a major transformation beginning in the 1960s such that old group alliances fell away (e.g. white southern, union member to some extent) and new ones

emerged (e.g. evangelical Protestant, Green voters) (see e.g. Levine, Carmines and Huckfeldt 1997).

I argue for the continued importance of examining and understanding group dynamics and their effect on public opinion in particular. As Huckfeldt (2007, 116) argues (for reasons I will elaborate) there is currently an "opportunity to reintroduce the study of groups in political analysis." Accordingly, I offer a theory of group influence on public opinion composed of three parts, social identity, group influence, and intensity of preference or issue ownership, and use this framework to answer the question of specific conditions under which public opinion is listened to and translated into government responsiveness.

2.1 Social Identity or How Citizens 'Think' in Terms of Groups

The foundation for my theory of group influences on public opinion rests on the idea beginning with the Columba school (Lazarsfeld 1944; Berelson et al. 1954; Katz and Lazarsfeld 1955) that citizens are heavily influenced by interpersonal relationships and these relationships are most often formed in the context of groups. As I argued, in contrast to the view of individuals as atomistic actors (whether intentional or not most examinations of public opinion treat individuals that way) other scholars (Huckfeldt 1979; 1983; 1984; Kenny 1992; Huckfeldt et al. 1995; Huckfeldt and Sprague 1995; Mcclurg 2003; Huckfeldt, Johnson and Sprague 2004; Walsh 2004; Mutz and Mondak 2006) argue that politics is most often carried out in terms of interpersonal relationships.

In this view, citizens learn and form opinions not from pure psychological factors or merely associating with a group (nominal membership), but from conversations and interactions with other citizens (see especially Walsh 2004). As Huckfeldt (2007, 116)

succinctly puts it, "it is through these networks that communication and persuasion occur." In the traditional view of this literature this does not have to (and some would argue does not) occur within the context of a group. This renewed focus on interpersonal relationships developed into a robust literature centered on the role of these social networks in political behavior (Huckfeldt 1979; 1983; 1984; Kenny 1992; Huckfeldt et al. 1995; Huckfeldt and Sprague 1995; Mcclurg 2003; Huckfeldt, Johnson and Sprague 2004; Walsh 2004; Mutz and Mondak 2006; see Huckfeldt 2007 for a review of this literature) and has extended to other subfields of political science as well.

As Huckfeldt (2007) in a review of the literature notes, however, social network analysis has to a large extent ignored the potential for continued overarching group dynamics in public opinion. He goes so far as to argue that studying public opinion within the context of these social networks "creates an opportunity to reintroduce the study of groups in political analysis (116)." I agree with Huckfeldt (2007) that the 'reintroduction' of the study of groups to public opinion is necessary and can best be done through social identity theory.

Social identity theory is premised on the idea that there is a human need to feel connected to something, to achieve some level of positive distinctiveness from others (Greene 1999). This theory argues that this occurs through identification with groups. Citizens identify with groups within society and this fulfills a basic need they have to feel connected to others (once again the importance of viewing citizens as social beings rather than individual actors). A corollary of this is that citizens then instinctively categorize the world into dichotomous groupings, being either a part of the in-group or the outgroup. This creates an 'us versus them' mentality that leads to two outcomes, favorable

perceptions of the in-group and increasingly biased perceptions of the out-group and increased loyalty towards the in-group (Greene 1999; Brewer 1979; Druckman 1994). In the context of public opinion, the result is that members of the in-group increasingly become influenced by and adopt the opinions and ideas of the group and groups take on a significant role in the life of the individual.

Social identity theory fits in well with the social network literature in public opinion because I argue interpersonal relationships are at root group based. The most basic social network is the family and an individual's network expands outward from there to friends, work based groups, broader groups within society such as religious institutions, political affiliations and more. Citizens form their opinions in the context of these various networks and in turn, from social identity theory, they develop group based patterns of opinion and behavior as well.

Consequently, for the foundation of my theory of group influences on public opinion I argue that there are clear dynamics whereby groups within society influence individual opinion formation. As with the historical literature, citizens still 'think' in terms of groups and are influenced by groups in what opinions they hold.

2.2 Group Influences on Politics

While I argue that groups still play an important role in shaping individual opinion within society, key to my theory of group influences on public opinion and answering my question of what are the conditions under which public opinion is listened to by government and translated into government responsiveness, is whether there are theoretical reasons to suppose that government is responsive to groups within society in particular.

Historically, at least since Truman (1951), a paradigm in political science by which to conceptualize American politics is through group identity and the related concept of pluralism. Truman (1951) argued that all politics is carried out in terms of groups and that these groups have multiple access points in their ability to influence government, the concept of pluralism. However, this was not a new concept. Madison in *Federalist 10* warned against the mischief of faction, arguing that it would fragment the newly united country. Yet, by the time Alexis de Tocqueville studied American democracy, as seen earlier, he observed that "in no country in the world has the principle of association been more successfully used or applied to a greater multitude of objects than in America" (Tocqueville [1835-40] 1969, 191). While many groups are not inherently political, as they are not organized for political purposes, they often take on political roles and become involved with political issues (Tocqueville [1835-40] 1969; Truman 1951).

The continued importance of groups to American politics is evidenced in the literature at both the mass and elite level. At the mass level, political participation, especially voting behavior, has often been explained in terms of group dynamics and group behavior. As noted, in the seminal study *The American Voter*, Campbell et al. (1960) argue that groups shape the politics of their members as a result of the psychological attachment that members feel for the group (group reference theory). More recent scholarship bears out this proposition. In particular, scholarship on union membership indicates that the stronger the attachment of members to a union the more likely they are to support union backed candidates (Rapoport, Stone and Abramowitz 1991; Clark and Masters 2001). At the elite level, Congressional literature examining the

representational linkage between citizens and government find that Congressmen provide substantive representation in relation to the size or proportion of groups within their geographic or reelection constituencies (Griffin and Newman 2005; Hill, Leighley, and Hinton-Andersson 1995). Accordingly, at least since Pitkin (1967), the concept of representation has been tied to groups. For example, descriptive representation is premised on the idea that various groups in society should have members of their group representing them in Congress.

It is clear in the literature that groups play an important dynamic in politics, yet in the public opinion literature there has been, as noted, less of a focus on the role of groups. Accordingly, while I argue that groups play an important role in shaping individual opinion, I also argue that groups play an important role in influencing government because it is ultimately only though the collective opinion that politics is actually carried out. This contention is clearly seen in the work of Downs (1957) and the classic paradox of the vote. Downs (1957) cleverly shows that an individual vote has little to no effect on changing the outcome of an election. The question he left scholars with became why then does someone actually go to the polls to vote? Applying this logic to public opinion, the question becomes why would (and would it even be possible for) government listen to one voice among the many? In a majority rule system, politicians must strategically cater to the constituency that will win them election (Mayhew 1974; Fenno 1978) and in doing so they must reach the most number of people they can. I argue that one of the primary ways politicians can target citizens is through targeting the groups to which citizens belong. This provides a much more manageable dynamic and ensures politicians are reaching the maximum number of people they can. Accordingly, it then becomes group

opinion and not individual opinion that influences politicians and so government decisions and policymaking.

I argue that politicians' catering to groups within society and consequently listening to group opinion is most likely to occur for dominant groups within society. For a group to be 'dominate' I argue that it must be both large and cohesive in opinion. A group must be large enough for politicians to find it worth their time and effort to target and listen to their opinion. From Tocqueville's early analysis of American democracy to today there are myriad groups within American society, for which it would be impossible for politicians to listen to them all. But a large group provides politicians with an easy way to target their message and gain supporters. It is not just the size that is important however, it is also the cohesiveness of the group. A group must be able to in a sense 'hold in line' the opinions of its members for politicians to be willing and able to listen to their opinion. When a group has cohesive opinion it provides a signal to politicians of the direction of opinion they need to take in order to win the support of the group.

Overall, I argue along with other subfields that group dynamics are an important part of American democracy. With regards to public opinion in particular, I argue that it is impossible for politicians (and consequently government) to listen to individual opinion. However, it is possible and theoretically highly likely that they listen to group opinion and consequently groups within society are able to make their opinions known and so influence government decision making.

2.3 Intensity of Preference and Issue Ownership

However, I further argue that this process of group influence is most likely to occur when a specific condition is present, when a group shows intensity of preference in

their opinion and 'owns' particular issues such that they signal clear direction in opinion both to group members and to politicians. Theoretically, democratic theorist Robert Dahl (1956) argues that politics is carried out with regards to "the relative intensity of preferences." In other words, American democracy institutes safeguards to ensure that the majority does not always win over the minority, especially if the minority holds an intense opinion. According to Dahl (1956), it is not simple majority minority rule which dominates politics, but the intensity with which the majority minority hold their opinions. An intense minority or majority is able to gain access to government to get their voices heard. This fits in with the concept of pluralism as argued by Truman (1951) that even small groups within society have the ability to at a minimum access government through multiple channels such as through the ballot box, through lobbying the presidency or through Supreme Court decisions.

Accordingly, when a group holds intense opinion on particular issues, I argue that they consequently provide a signal to both their own group members and to politicians of the direction of their opinion. In the voting behavior literature, scholars have found that there are partisan stereotypes whereby voters are able to associate positions on issues to either the Republican or Democratic Party. This concept is elaborated in the idea of issue ownership where it is argued that parties hold such a monopoly on a particular issue that they consequently 'own' the issue and it provides a clear signal to voters of the position of individual candidates within the party (Rahn 1993; Petrocik 1996; Egan 2013).

The concept of issue ownership can extend beyond political parties to groups within society. Groups by design often tend to be issue focused (e.g. environmental groups, animal rights groups). Even large groups within society often show clear

direction in terms of their opinion on issues (e.g. religious groups and moral issues). When these group opinions on issues are combined with intensity of preference on the issue, it creates the situation whereby a group can come to 'own' a particular issue. The result as I argued is that it provides a clear signal to individual members of the group of the 'correct' opinion they should hold on the issue (and social identity theory works to ensure that group members follow that opinion as argued earlier) and a clear signal to government of the opinion of the group. For both reasons this is likely to translate into government listening to and being responsive to public opinion as mediated through groups. Without this condition of intensity of opinion present, however, it is not as likely that government will listen to group opinion.

In summary, my theoretical argument has three parts. First, I argue that the importance of interpersonal relationships and social identity theory work to ensure that citizens are still group orientated and are consequently influenced in their individual opinion by groups. The desire to feel connected to others is alleviated through group interactions and in turn individuals develop group based patterns of opinion. Second, I argue that in a large democratic society such as the United States it is virtually impossible for government to listen and be responsive to individual opinion. However, government is likely to listen to collective opinion which is often mediated through dominate groups within society (groups that are both large and have cohesive opinion). Accordingly, government is likely to listen to and be responsive to group opinion rather than individual public opinion. Third, I argue that this group influence is conditional on the group showing intensity of preference and opinion and 'owning' particular issues. This is necessary because it shows individual members of the group what opinion they should

adopt and it signals direction of opinion to government such that politicians have a clear indication of what position they must take to win support and consequently listen to the opinion of the group.

2.4 Focus Group: Religion

To test this theory of group influences on public opinion I choose to focus on religious groups. Religious groups fit all three criteria of the model and should provide a good test for this theory. For my theory and analysis I focus on the largest religious groups in American society, evangelical Protestantism, mainline Protestantism, Catholicism and black Protestantism.

Social Identity

Religious groups offer perhaps the best test of social identity theory and its relationship to group opinion for two reasons. First, social identity theory typically focuses on characteristics of an individual (and so the consequent group it places them in) that are genetic, such as gender, ethnicity and race. Religion, on the other hand, is a group to which familial identification plays a role (I'm a Catholic because my mom was a Catholic) but is also very much the product of individual choice (see especially Fink and Stark 2005 and the idea of the marketplace of religion in the United States). Because it is a group that an individual chooses to be a part of, it is likely that the attachment they feel for the group will run deeper as it took them more effort to go about the task of 'joining' one particular religious group over another (or no religion at all). As a result, even nominal membership in a religious group is likely to have strong social identify effects and thereby produce individual opinion that is shaped by the group. Scholars back up this claim for religion in particular arguing that simply belonging to a religion can

provide a forum in which politics is discussed and where religious leaders especially can (and often do) shape the opinion of individual members of the congregation (Kohut 2000; Smidt et al. 2009; Putnam and Campbell 2010). Of course a corollary of this is that the greater an individual is attached to their religion (religious commitment such as attending church once or more than once a week) the stronger the social identify effect will become.

Second, religion offers perhaps the best test of social identity theory because it arguably best provides the sense of 'connectedness' that is at the heart of the theory (Greene 1999). Citizens look to groups to find purpose and meaning and build relationships with one another and the aim of religion is to provide exactly that. While other groups in American society also provide a sense of meaning and purpose, that is often not their express goal as is the case with religion. While even nominal membership provides this sense of connectedness and purpose, again this is greatly magnified for those who are more committed to their religion. For example, gathering together once or more than once a week provides an ideal basis for the development of interpersonal relationships. For these two reasons, religious groups provide a context in which individual opinion is being shaped and molded by the group.

Group Influences

Religious groups provide an excellent test of the second part of my theory that government listens not to individual opinion but to group opinion. The two criteria I offer for this to take place are for a group to be both large and cohesive in opinion.

Despite predictions of secularization (see Gorski and Altinordu 2008 for a review), religion has persisted in America with the result that America remains one of the

most religious nations in the world. Surveys reveal that nine out of ten Americans claim to identify with a religious group or tradition (Dougherty, Johnson and Polson 2007). While there are many small religious groups within the United States, the four largest groups, evangelical Protestants, mainline Protestants, Catholics and Black Protestants combined make up more than three quarters of the population. Evangelical Protestants a and Catholics each contain roughly a quarter of the population, mainline Protestants a little less than 20 percent of the population and black Protestants about seven percent of the population (Pew Religious Landscape Survey 2007; Wald and Calhoun-Brown 2011). These percentages make religious groups among the largest groups in American society. Rates of religious participation reveal that Americans are very likely to not just claim identification with a religion but actively engage in their faith. For example, eighty percent of Americans report going to church a few times a year or more, with forty percent of Americans reporting weekly (or more) church attendance (Wald and Calhoun-Brown 2011, 11).

Additionally, religious groups, the four largest ones in particular, have come to play an increasingly important role in American politics. This is summed up by Corwin et al. (2009, 1) in the *Oxford Handbook of Religion and American Politics* where they state that over the past three decades in particular "there has been a growing recognition that religion plays a vital role in American politics" with the consequences that it is now a major subfield of examination in American politics. The rise of the Religious Right in particular led scholars to investigate the phenomenon of religion's impact on politics (Wilcox and Larson 2006; Wilcox and Robinson 2011) and they have found profound effects for the influence of religion on politics, especially as related to attachment with

the Republican Party (Domke and Cole 2008). Consequently, it is a group that government is likely to pay attention to and listen to the opinion of when making decisions.

Aside from the size of these four religious groups, they have also come to play an important role in politics because they exhibit cohesiveness as a group (though some more so than others). The cohesiveness of the group is well documented, for example, in the voting behavior literature, where scholars find that religious groups identify with particular political parties (see for example Harris 1999; Layman 2001; Classen and Povtak 2010) and it extends to stances on political issues. For example, black Protestants have consistently been shown to be extremely conservative on social and moral issues, as have evangelicals (Steendland et al. 2000; Campbell and Monson 2008). Therefore, this indicates that religious groups are able to send signals on their preferred policy to government.

Intensity of Preference and Issue Ownership

Finally, religion fits my theory particularly well because not only are religious groups cohesive groups in terms of opinion (as just argued) but they are also groups that hold intense opinions and have come in some ways to 'own' some issues in particular.

As I argued, it is most likely the case that government is responsive to group opinion when the group holds an intense preference about an issue. It benefits government to pay attention to intensity of group opinion in particular because it provides strong signals for government to follow and they can be greatly rewarded for doing so (in terms of votes, financial support etc.). Conversely, it would be costly for government to

ignore intensity of preference and at the extreme it would result in loss of faith in and legitimacy of democracy (Dahl 1956).

Religious groups tend to hold very intense opinions. By design, religious groups most often (especially the four big religious groups under analysis) think of the world in terms of black and white, truth and untruth. Accordingly, they tend to hold strongly to their beliefs (though there are exceptions such as ecumenicalism among Mainline Protestants and Catholics to some extent see Fowler et al. 2010; Wald and Calhoun-Brown 2010). In the context of American politics, these beliefs have most often translated into intense opinion over social and moral issues (though the religious agenda has expanded see Wilcox and Robinson 2011; Smith and Olson 2013). The intensity with which religious groups hold to their opinion on social and moral issues has led scholars to argue that there is a 'culture war' between religious groups and secular society (Hunter 1991; Cook et al. 1992; DiMaggio et al. 1996; Evans 2003; Fiorina 2010; Gay et al. 1996; Green et al. 1996; Layman 1999, 2001; Leege and Kellstedt 1993; Putnam and Campbell 2010). As a consequence of the intensity with which religious groups hold to their opinion, they have come in some ways to 'own' social issues in particular such that individual members of the group know where the group stands and it provides a clear signal to government of direction of opinion, which has most often led to ties with the Republican Party (Green et al. 1996; Kohut et al. 2000; Layman 2001; Layman and Green 2006; Regnerus, Sikkink and Smith 1999; Calfano and Djupe 2009; Layman 2010). Overall, religion provides an excellent group to focus on as a test for my theory of group influences on public opinion.

2.5 Theory Testing: Three Cases

In testing my theory of group influences on public opinion and answering the question of the conditions under which public opinion is listened to and translated into government responsiveness, I conduct three separate analyses examining strength of partisanship, state Supreme Court decision making and ballot initiatives. Each analysis focuses on a different avenue by which public opinion affects government. Conducting three separate analyses allows me to test the breadth and depth of my theory. Each analysis examines a sub-research question but I argue that the overarching theory applies and is tested in each case. Again religion (and for two of the three studies the four largest religious groups in America) is the group I focus on as a test for my theory of the influence of group opinion on governmental decision making

Political Parties: Strength of Partisanship

In this analysis I examine my theory of the influence of group opinion on governmental decision making through the lens of political parties and strength of partisanship. In American democracy, partisanship is arguably the prime way in which public opinion is made known to government (and consequently listened to). While the Founding Fathers and James Madison in particular were worried over what the mischief of faction could do to a new democracy and consequently sought to prevent the formation of political parties (see *Federalist 10*), from the very early days of American democracy a two party system of political parties emerged (Aldrich 1995). Contrary to the decline of party thesis (Burnham 1989; Wattenberg 1996), the literature shows that political parties still play a large and vital role in American democracy (Bartels 1992, 2000; Aldrich 1995; Hetherington 2001). As scholars argue, political parties provide an efficient means

by which mass opinion is aggregated and majority rule is able to be carried out (Key 1942; Committee 1950; Aldrich 1995). As part of the classic normative view of political parties, the responsible party thesis (Committee 1950), parties make a commitment to the public to listen to their opinion and carry out their agenda once in office. Overall, political parties help provide structure to public opinion in American democracy.

However, I argue that it is not partisanship in a nominal sense that affects public opinion but strength of partisanship. If it is true that one way in which public opinion is translated to government is through political parties, then the important question becomes how the parties themselves form their agenda and opinions. In other words, is the opinion and agenda of the parties simply the aggregation of those who affiliate with the party or is it formed by something else? Aldrich (1995) answers this by arguing that parties have always been no more than a tool for politicians, that elite individuals have great power in shaping the opinion and agenda of the parties. Schattschneider (1960, 56) on the other hand argues in part that parties are divided into two entities, "an organized mass of insiders who have effective control of the party" and mass "passive members" who have little say.

I argue along the lines of Schattschneider (1960) that the opinion of the party is most often shaped by those who are strongly attached to the party, strong party identifiers in the party identification scale. The original authors of *The American Voter* (1960) argued for this role of strength of partisanship, identifying it as an important avenue for future research. Scholars find that strength of partisanship leads to greater involvement in government, such as through higher rates of voter turnout (Milbrath and Goel 1977; DeNardo 1980; Abramson and Aldrich 1982; Teixeira 1992), greater participation in

primary elections (Ranney 1972; Abramowitz, McGlennon, Rapoport 1981), and increased loyalty to the political system (Campbell et al. 1960; Converse 1969). Often overlooked in the literature, I argue that strong partisans greatly affect the opinion and agenda of the overall party, thus affecting how public opinion is translated to and listened to by government.

The question then becomes, what causes someone to identify as a strong partisan and consequently have these effects on public opinion? This is the sub-question I ask in my analysis and I argue that my theory of group influences on public opinion provides an answer to this question, focusing specifically on religious groups. More particularly, I ask the question, is there a link between strength of religious attachment and strength of partisan attachment.

First, I argue social identity theory extends to both religious identity and partisanship. These are two groups within society that citizens develop identities with and consequently develop the us versus them mentality and group loyalty whereby individual opinion is shaped by the overall opinion of the group. These social identity effects are likely to be even greater when examining strong partisans and those who are strongly attached to religion in particular. They are self-described as more attached to their party and religion than other members of the group.

Second, as just argued, public opinion is arguably most often translated to government through political parties. The opinion of the party that holds the majority in government is likely to have their opinions (to some degree) focused on and responded to in terms of government policy. However, per my overall argument, it is theoretically likely that political parties are controlled not by aggregate member opinion but by

specific groups within the party. If identifying with a group leads to strength of partisanship, as just seen, that will in turn lead to changes in the overall composition and opinion of the party such that the opinion of the party will then take on the opinion of the group. I argue this is likely to happen through religious groups. However, it is conditional on the third element of my theory, that the religious group holds intense preferences and to some extent 'owns' particular issues. In the context of political parties, it is clear in the literature that this is indeed the case, as religious groups have more and more identified with the Republican Party with their focus on conservative stances on social and moral issues (Green et al. 1996; Kohut et al. 2000; Layman 2001; Layman and Green 2006). Consequently, from my argument, the group opinion influence should only hold for the Republican Party and not for the Democratic Party.

A note for my theory as tested in this analysis (as opposed to the other two) is that I focus on strength of religious attachment rather than individual religious groups. As such I am not providing a complete test of my theory. However, religious attachment only occurs in the context of attachment with a religious group, so it is still group orientated behavior. What is missing is the distinction between different religious groups and their effect on strength of partisanship, here I am simply lumping them all together. Extensions of the analysis will need to look at individual religious group effects.

State Supreme Court Decision Making

In my second analysis, I test my theory of group opinion influences on one of the three branches of government, the Supreme Court as measured at the state level. The Supreme Court presents a good test because it is the branch of government that is most often thought of as insulated from public opinion. It was designed by the Founders to be

to some extent independent from public opinion through the fact that it is the only nonelected branch of government (e.g. see *Federalist* 78). However, at the federal Supreme Court level, scholars are beginning to show that justices do listen to and respond to public opinion in their decision making (Caldeira 1991; Mishler and Sheehan 1993; Flemming and Wood 1997; McGuire and Stimson 2004; Giles et al. 2008). At the state level, the potential for justices listening to public opinion is more apparent since some states do hold judicial elections. And, indeed, scholars find that elected justices do respond to public opinion (Brace and Boyea 2008; Devins and Mansker 2010).

This literature examines the effects of public opinion on judicial decision making through the lens of overall opinion, measured in various ways such as by the 'public mood' (Mishler and Sheehan 1993; Flemming and Wood 1997) overall ideology or by overall public opinion on a specific issue (Brace and Boyea 2008). As such this literature misses the importance of groups in shaping opinion. Accordingly, I ask the question, is it possible that justices respond not to overall opinion but to specific group opinion? In particular, do state Supreme Court justices listen to religious opinion?

Accordingly, I address this question through the lens of my theory of group influences on public opinion focusing on the presence of religious groups, and so religious opinion, in the states. First, the foundation is once again social identity theory. Religious groups in particular provide the sense of connectedness that individuals seek and consequently individual members of the group develop the us versus them mentality whereby they become influenced by the group in their opinions and judgments.

Second, I argue that overall opinion in the states is mediated by groups within society. Again, it is impossible in a democracy for government to listen to individual
opinion. Opinion must be aggregated in some form or fashion for government to be able to listen to it and respond to it in terms of policymaking. This is theoretically likely to occur through groups within society. For state Supreme Court justices, there may be dominant groups within the state that they feel they must listen and respond to in their decision-making (for reasons I elaborate in the analysis, e.g. judicial election, legitimacy). Religious groups in particular, as I argued, are dominant groups within society in terms of both their size and cohesiveness in opinion, and consequently present a good test to see whether justices respond to their opinion.

Finally, I argue that this is only likely to occur when the group holds an intensity of preference on the issue. For my analysis I use the issue of the death penalty as my test case. I examine whether justices change their decision making on death penalty cases (measured as a death penalty reversal) based on the presence of religious groups in the states. The issue of the death penalty is an issue that religious groups feel strongly about (Young 1992; Britt 1998; Pew Research Forum Survey 2007) and so signal clear direction of opinion to their group members and to the state Supreme Court justices. For this analysis I analyze the four largest religious groups in the states, evangelical Protestants, mainline Protestants, Catholics and black Protestants.

Direct Democracy: Ballot Initiatives

For my third and final analysis I test my theory of group opinion influence on government through the lens of direct democracy. Direct democracy occurs when citizens are directly involved in making and forming policy. In American democracy this most often takes place in individual states through the use of ballot initiatives, referendums and recall elections. As Smith et al. (2008, 102) note, analysis of the effects

of direct democracy was ignored by political science and legal scholars for much of the twentieth century but has received increased attention in the past few decades such that it is now "in vogue" and a burgeoning literature (e.g. Bell 1978; Smith and Tolbert 2001; Bowler and Donovan 2002; Nicholson 2005; Smith, DeSantis, and Kassel 2006; Smith and Tolbert 2007). These scholars examine the effects of direct democracy on a variety of different things including political efficacy among citizens (Bowler and Donovan 2002) levels of political knowledge (Smith 2001) and voter participation (Smith 2001; Tolbert, Grummel, and Smith 2001) among others.

Focusing on public opinion in particular, from a theoretical standpoint, direct democracy should result in public opinion being clearly translated into government policymaking (since 'the public' is the one making the policy). Indeed, scholars find that states with initiatives and referenda are more responsive to public opinion than those without (Arceneaux 2002; Burden 2005). However, at the same time, scholars call into question the ability of direct democracy to actually reflect the underlying structure of mass preferences (Riker 1982; Broder 2000). These scholars warn that a rosy picture of direct democracy where direct democracy is equated with perfect public opinion responsiveness may not happen in actual practice.

I argue that my theory of group influences on public opinion can account for this discrepancy that scholars find. If my theory is correct in the assumption that public opinion is mediated through dominant groups within society, such that public opinion is best thought of in terms of group opinion, then the effect of direct democracy should be that public opinion as a whole is not followed but public opinion of groups within the state.

Accordingly, I test my theory and this claim analyzing one specific aspect of direct democracy, candidate choice, asking the question, do ballot propositions affect candidate choice. More specifically (to my theory), I examine whether religious ballot propositions affect candidate choice in elections by priming religious individuals (religious groups) in who to vote for. Here I am looking at the 'spillover' effects direct democracy can have on shaping opinion through candidate choice.

Once again the foundation for my theory is the presence of social identity effects. For ballot propositions in particular, public opinion is reflected through citizens showing up at the polls and voting on the policy at hand. For the spillover effects to occur and so public opinion to be reflected in the choice of candidates, it is also dependent on citizens' first showing up at the polls to vote. Social identity, in the context of religious groups, therefore works in two ways in this analysis. First, the voting behavior literature clearly shows a positive relationship between religious affiliation and voting and this relationship grows stronger when religious commitment is taken into account (Macaluso and Wanat 1979; Hougland and Christenson 1983; Hill and Cassel 1984; Huges and Peek 1986; Peterson 1990; Rosenstone and Hanson 1993; Verba et al. 1995; Smidt et al. 2007). While scholars provide different explanations for this finding, I argue that social identity applies. The social identity effects of a group are not isolated to opinion formation but can extend to behavior. Whereas I have focused primarily on opinion effects, that being a part of the group creates the us versus them mentality and as a result causes individual members to be shaped by and take on the opinion of the group, the group can also shape patterns of behavior. If the group highly values participation in elections, individual members will conform to this behavior and vote in elections. Indeed with regards to

religious groups in particular scholars are recognizing this possibility and the consequent 'habit' of political participation that is formed (see Smith and Walker 2013). Second, as just seen, social identity applies because individual members of the religious groups are shaped in their opinion by the group and therefore in the direction of candidate choice.

As noted, some scholars are skeptical of the link between direct democracy and public opinion, arguing that direct democracy does not truly reflect the mass distribution of public opinion (Riker 1982; Broder 2000). The second part of my theory can help account for this. By conceptualizing public opinion in terms not of overall opinion but group opinion, it may be the case that opinion is accurately reflected through dominant groups. I argue in my specific analysis that ballot propositions target (intentionally or not) specific groups within a state to turn out to vote and as a corollary to vote for a specific candidate. Consequently, the overall result of the ballot proposition will reflect group opinion and the overall vote for candidates will reflect group opinion as well. However, this is again only likely to occur when the specific condition of group intensity of preference is met. The ballot initiative I examine is gay marriage amendments in the 2004 election. This is an issue on which religious groups hold very intense preferences and can be considered to 'own' it along with other key moral issues such as abortion (Abramowitz 1995; Domke and Coe 2008; Wilcox and Robinson 2011). For evangelical Protestants the media claimed that this particular issue was used to specifically target increased evangelical turnout for President Bush (e.g. Dao 2004). Accordingly, this issue meets the condition of group intensity of preference and so it is highly likely that group opinion is reflected rather than overall opinion.

2.6 Summary

In summary, I offer three distinct analyses of my theory of group influences on public opinion: political parties, arguably the primary way in which public opinion is translated to government, state Supreme Courts, focusing on the branch of government least likely to be responsive to public opinion, and direct democracy through the use of ballot initiatives, arguing for the effect of public opinion on candidate choice through spillover effects of ballot initiatives. Analyzing these three distinct areas will help me test the breadth and depth of my theory.

Accordingly, in chapters three through five, I present these analyses. Each contains a sub-research question and theory as I showed and as such will be presented as standalone analyses. However, I argue they all fit within the framework of my overall theory and help to answer my overall research question of finding the specific conditions under which public opinion is listened to and translated into government responsiveness. In chapter six I will examine specifically how well my theory performed for each analysis and offer concluding thoughts.

Chapter 3

Partisan Strength and Religious Commitment

"Most Americans have this sense of attachment with one party or the other. And for the individual who does, the strength and direction of party identification are facts of central importance in accounting for attitude and behavior."

- Campbell et al. *The American Voter* (1960)

Party identification, as noted by Green and Palmquist (1990), is the single concept most extensively discussed and analyzed in the field of American political behavior. First identified and elaborated by Campbell et al. (1960) it remains an important and enduring concept. It is found to play a central role in such things as individual vote choice, evaluations of government and judgments about political leaders and political groups.

Recent literature stresses that contrary to the "decline of parties thesis" (see Burnham 1989; Wattenberg 1996) the impact of party identification is substantial (and possibly increasingly so) on the political behavior of citizens (Bartels 1992, 2000; Hetherington 2001). This is stated perhaps the most forcefully by Bartels (2000,35) as he finds that partisan loyalties had as much impact on the voting behavior of citizens at the presidential level in the 1980s as in the 1950s and perhaps more impact on voting behavior in the 1990s. Therefore he concludes that "the 'decline of parties' in American politics is badly outdated." This literature suggests that party identification should and will continue to receive attention by scholars of political behavior.

The great majority of the literature on party identification revolves around a few central debates. The first is the role of partisanship in shaping and influencing political behavior, including, among other things, the voting behavior of citizens. As just evidenced, this literature waxes and wanes over time as the importance of partisanship to citizens' behavior is debated (Bartels 2000). A second debate looks at trends in partisanship at the macro level, arguing for its stability (Green and Palmquist 1990; Green, Palmquist and Schickler 2002) or instability (Mackuen, Erickson and Stimson 1989; Erickson, MacKuen and Stimson 2002) over time. Yet a third debate in this literature focuses on the definition of party identification, whether it is an exogenous variable as conceptualized by Campbell et al. (1960) as a highly stable orientation developed at a young age, or whether it is an endogenous variable subject to short-term influences such as evaluations of presidential performance (Fiorina 1981; Page and Jones 1979; Brody and Rosenberg 1988) and candidate ratings (Markus and Converse 1979; Page and Jones 1979) among other things.

A central theme running through these studies of party identification is analyzing the direction of partisanship (including a focus on independents and independent leaners) without necessarily considering the magnitude or strength of partisanship. The majority of the literature (except see Ranney 1972; Milbrath and Goel 1977; Holbrook and McGlurg 2005) is concerned with understanding how citizens acquire their partisanship (regardless of which party they identify with) and how that affects their political behavior. However, as the opening quote suggests, both the direction and the strength of party identification are of central importance to fully understanding the role that party identification plays in citizens' lives.

Accordingly, I argue that understanding the strength and not just the direction of partisanship is a vital research agenda. There are many normative reasons why strong party identification is appealing, yet the literature provides little theoretical or empirical studies with this as their focus. I seek to more fully answer the question of what influences the strength to which citizens are attached to their political party.

In particular, I argue that there is a psychological aspect of partisan strength that has gone overlooked within the political science literature. I argue that attachments individuals feel towards two identities are related: attachment to their political identity and attachment to their religious identity, that is, individuals identifying as a member of a group is an important part of their conception of self. The assumption behind this is that those individuals who value their identity within a group within one realm of life may also find it desirable to do so in the political realm. As such, I frame my argument within the social identity theory literature that has been developed within other social science fields and seek to expand its importance to the area of political science.

Social identity theory suggests that individuals become highly attached to a number of different groups and identities. Accordingly there are a number of different identities that may affect one's political identity other than religion. However, I argue that religious identity, since it is more fluid (more of a choice involved, which religion, how deeply committed etc.) rather than a fixed entity (gender, race, ethnicity, etc.) is arguably the best identity that may affect one's choice of a political identity. Therefore, in this chapter the key question I address is whether the degree of loyalty (partisan strength) citizens feel towards political parties can be explained by their loyalties to other

identities, in particular their loyalty to their religious organization. In other words, does greater religious commitment lead to greater partisan strength?

In answering this question I control for a number of factors that may affect partisan strength including personal factors such as age, level of educational attainment, political factors such as ideological strength, issue saliency, political knowledge, interest in politics, and levels of social capital which have been shown to affect one's partisan strength. I also break the analysis apart by political party since it may be the case given the literature (Green et al. 1996; Kohut et al. 2000; Layman 2001; Layman and Green 2006) that religious identity may affect Republican identifiers more than Democratic identifiers.

Overall, I find support for my theory that there is a psychological link between partisan strength and religious commitment and it is stronger for Republicans than Democrats. Accordingly, this analysis provides another explanation for differentiating levels of partisan strength and does so using a theory that is relatively absent within the political science literature, though it figures quite prominently in other areas of social science research. However, I still find a lot of variation in partisan strength unaccounted for leaving this as an important research agenda for the future.

3.1 Strength of Partisanship: Normative Concerns

While the normative implications of party identification are probed in the literature, that, among other things, it provides a cue for voters (Lupia 1994) and as a result allows otherwise uninformed citizens to act as if they have some amount of political knowledge (desirable from a democratic theory standpoint), the normative concerns regarding strength of partisanship are no less attractive. In fact there are a

number of political implications as well as broader implications for democracy regarding strength of partisanship.

The literature, both theoretical and empirical, is clear that strength of partisanship is a strong predictor of political participation and voter turnout in particular. Campbell et al. (1960) initially made the claim that strong party identifiers are more likely to be engaged in politics. They argue that strong feelings of party identification help contribute to a psychological involvement in politics (see also Milbrath and Goel 1977). Later scholars trace the decline in voter turnout in the mid to late 20th century to declining strength of party identification among citizens (DeNardo 1980; Abramson and Aldrich 1982; Teixeira 1992). These studies demonstrate that, as with the pluralist chorus (Schattschneider 1960), the American electorate sings with a specific accent—in this case, a strong partisan accent. The fact that strength of partisanship leads to increased political participation begs the question of discovering what causes individuals to hold strongly to a particular party. If there are ways to increase the likelihood that individuals strongly identify with a party then the normatively desirable result will be increased citizen involvement in government.

Similarly, strength of partisanship significantly influences the likelihood of voting in a primary election (Ranney 1972; Abramowitz, McGlennon and Rapoport 1981). As a result, strong partisans have an inordinate influence on the selection of candidates to run for political office. Though not as extreme in ideology as political activists (see Aldrich 1983 for a discussion of activists) the fact that they vote disproportionately compared to other groups (weak partisans and independents) likely results in the selection of more extreme candidates. This could be part of the reason behind the failure of spatial models

which predict the emergence of more moderate or centrist candidates (Downs 1957). The 2010 midterm elections and 2012 Republican presidential nomination contest present a recent case study of this phenomenon, in the selection of more extremist candidates in primary elections across the United States.

Strength of partisanship also has broader normative implications. Scholars argue that strong parties make for stronger democracies. Campbell et al. (1960) and Converse (1969) argue that partisan loyalty begets loyalty to the political system. When individuals become attached to a particular party it indicates that they are willing to work within the political system that is established, thereby enhancing the overall stability of the government. It is not partisanship itself that leads to these normatively desirable outcomes, but partisan loyalty, in the words of Campbell et al. (1960) and Converse (1969).

Strength of partisanship has important political (political participation) and governmental (democratic stability) implications. For all of these reasons, it is evident that it is vitally important to identify the mechanisms at work causing individuals to identify as strong partisans, the question that I address in this analysis.

3.2 Previous Research

The traditional measure of party identification is a seven point scale which includes categories for strong partisans, weak partisans, independent leaners and pure independents (Campbell et al. 1960). There is debate over whether this scale is an accurate measure of party identification (see for example Weisberg 1980 and Franklin and Jackson 1983 on the dimensionality and dynamics of the traditional party identification measure). Yet it is clear that each category behaves in a distinct fashion on

a number of political variables including voter turnout and vote choice. This indicates that it is tapping some level of individual decision making with regard to how citizens conceive of their party identification.

While much literature exists examining the independent voter (e.g. Burnham 1970; Abramson 1976; Norpoth and Rusk 1982; Clark and Suzuki 1994), less research is conducted focusing on the strong partisans of the party identification scale. Converse (1969; 1976) offers the first concentrated theoretical effort at answering the question of what causes voters to identify as strong partisans. He argues that strength of partisanship is a predictor of political stability (as seen earlier). The biggest influences on partisan strength, he argues, are social learning and the intergenerational transmission of party ties. Social learning comes with age. As individuals' age, they are more likely to identify strongly with a political party because they identify to a greater degree with the political system and have participated in the political system. The second influence on partisan strength is the transmission of party identification elaborated by Campbell et al. (1960), the funnel of causality, citizens' strength of partisanship is heavily influenced by that of their parents.

However, his theory, especially his assertion that as individuals age they become more loyal to their political party, is heavily criticized. In particular, scholars argue that is not individual life-cycle changes that effect partisan strength, but generational changes (Abramson 1976; 1979). These scholars argue that the decline of parties evidenced in the 1960s and 1970s (except see Keith et al. 1992) resulted in younger generations having weaker political ties than their parents (Burnham 1970; Niemi et al. 1985). Scholars also

quibble with the argument that strength of partisanship is influenced by the transmission of intergenerational party ties. They argue that intergenerational effects result in the transmission of the direction of partisanship but not the strength or intensity of it (Claggett 1981).

Scholars after Converse (1969; 1976) identify other potential causes of partisan strength, including political participation, in the form of voting because it solidifies party attachment (Cassell 1993), length of party attachment (Tilly 2003), and high socioeconomic status (Campbell et al. 1960). However, it is clear that there has been little theoretical development on the causes of the strength of partisanship since Converse (1969). In part, this could be the case because of the perceived elegance of Converse's model (see Shively 1972; 1990; Niemi 1986).

3.3 Religious Attachment and Hypotheses

However, there may be other potential mechanisms at work causing citizens to self-identify as strong partisans. In particular, I argue that there is a relationship between strength of party identification and strength of religious attachment (or religiosity). The more committed an individual is to their religion the more likely that person is to identify as a strong partisan. In other words, commitment to religion should affect levels of commitment in other areas of citizens' lives. Past research has tended to look at these potential effects through the lens of social capital theory and I test for this possibility. However, I argue that a second mechanism as yet unidentified is at work, social identity theory.¹

¹ There is the potential that religious commitment and strength of partisanship are endogenous. However, theoretically, I argue, based on the conceptualization of party identification of Campbell et al. (1960), that religious commitment comes first. As a

Scholars demonstrate that citizens hold a certain amount of social capital, which is defined as "connections among individuals - social networks and the norms of reciprocity and trustworthiness that arise from them" (Putnam 2000: 19). Stores of social capital are related to participatory attitudes. The greater the amount of social capital the more likely citizens are to engage in politics (and other forms of group activity such as organizational meetings). Scholars make clear that one component of social capital is religious affiliation and religious commitment (Verba et al. 1995; Putnam 2000). Citizens learn skills in religious settings that prepare them and enable them to participate in politics. These scholars lament the decline of social capital in America, as they fear it will result in less citizen participation in government as well as in other areas of life including school boards and PTA meetings (see especially Putnam 2000).²

I argue that the connection between social capital and participatory attitudes can be extended to include group attachments and partisan attachments in particular. Citizens who hold high stores of social capital are more likely to not only participate in politics to

result, the relationship is exogenous. An individual's party identification develops though the process of political socialization, part of which includes characteristics of the individual and their parents including religious commitment. Therefore, religious commitment is likely to be a primary influence on other areas of the individual's life. ^{2} Wielhouwer (2009) argues that there are three distinct approaches identified in the literature within the broader social capital argument of the link between religion and political participation: (1) the type of individuals who are likely to engage in one form of social involvement, such as church attendance, are likely to engage in other forms of social involvement, such as voter turnout (Rosenstone and Hanson 1993; Jones-Correa and Leal 2001); (2) religious settings operate as "centers for electoral mobilization" where members are exposed to messages about social and political issues and are often targets for mobilization efforts (Cassell 1999; Harris 1999; Guth et al. 2002); (3) religious settings are venues for political communication (Wald, Owen, and Hill 1988, 1990; Huckfeldt, Plutzer, and Sprague 1993) as well as "crucibles of democratic skills" where members gain skills that can be easily applied to the political or civic arena (Verba et al. 1995; Smidt 1999; 2003).

a greater extent but also to hold stronger partisan views. This is the case because, I argue, seeing a greater need for participation in the political system (the result of social capital) corresponds with holding a particular view of what actions government should take. As citizens become engaged in politics to a greater extent they see the necessity of actively promoting a particular worldview, which, in the American system, means (most often) promoting one of the two political parties, thereby becoming more likely to identify as strong partisans. Following this logic, religiosity (a main source of social capital as noted) leads to strength of party identification.

A second theoretical reason that religiosity may presumably lead to strength of partisanship is related to social identity theory. Campbell et al. (1960) first argued that party identification can be understood by group reference theory, the idea that as people identify with racial, ethnic or religious groups, so too do they identify with political parties. It posits that party identification can and should be treated as another fundamental category or group to which citizens are psychologically attached (see also Greene 2004; Green, Palmquist and Schickler 2002 for extensions).

Social identity theory expounds this idea, stating that identification with a group is motivated by a human need to feel connected to something, to achieve some level of positive distinctiveness (Greene 1999). Social identity theory posits that humans instinctively categorize the world into dichotomous groupings, being either a part of the in-group or the out-group. This creates an 'us versus them' mentality that leads to favorable perceptions of the in-group and increasingly biased perceptions of the outgroup, leading in even the most minimal conditions to the emergence of in-group favoritism (Greene 1999; Brewer 1979). When this favoritism is coupled with the

increased biased view of the out-group, social identity theory seems to be a key factor in intergroup conflict (Tajfel 1981; Turner, Hogg, Oakes, Reicher & Wetherell 1987). Accordingly, intergroup conflict can spur a group on towards collective action (Huddy 2001), which in the case of politics most often happens by first identifying with a political party. Social identity theory also posits that a consequence of intergroup conflict is increased loyalty to ones' own in-group, for the purposes of this study, strength of religion and strength of partisanship (Druckman 1994).

European studies have found considerable evidence that social identity theory does have a sizable impact on partisanship (Kelly 1988; 1989; 1990a; 1990b). While most of the literature on social identity theory focuses on nature-based groupings (race, ethnicity, gender etc.) I argue that this theory applies to religious identity as well. Accordingly, I argue that social identity is not limited to one aspect of an individual's life (gender for example) and consequently, individuals who feel a need to be connected in one area of their life (religion) might also reasonably feel a similar need in other areas of their life (politics). I argue that social identity can and does have a spillover effect such that levels of commitment to one identity lead to higher levels of commitment to another identity. Therefore the level of commitment citizens hold to their religious group affects the level of commitment they hold to their political party (and so strength of partisan attachment).

Hypothesis 1: Strength of religious commitment (religiosity) is positively related to strength of partisanship.

There is good reason to suspect, however, that this hypothesized effect of religious commitment on partisanship strength might well be mitigated by partisans'

perceptions that their religious commitment has some meaning in relation to their political party's platform, ideas and leaders. Therefore, it is likely that as partisans' judge their political party in terms of their party's religious attachment, it may lead to different outcomes in how religious commitment affects Republicans and Democrats. More specifically, if religiously committed Democrats fail to think of the Democratic Party as itself being religiously committed, then the us versus them mentality central to social identity theory cannot take root and consequently there may not be this strong link between religious commitment and partisan commitment. On the flip side if Republicans view the Republican Party as being particularly well attached to religion, it would encourage an us versus them mentality to take hold. In other words, the perception that people have regarding the political party's level of commitment to the same group (in this case religion being the group) is central to the application of social identity theory.

Previously research shows that the Republican Party is becoming increasingly religious whereas the Democratic Party is becoming increasingly secular (Green et al. 1996; Kohut et al. 2000; Layman 2001; Layman and Green 2006) and that there is an increasing "religion gap" with regard to religious practice, or religiosity, between the two parties (Kohut et al. 2000; Layman 2001; Green and Silk 2003; Fiorina 2005; Olsen and Green 2006; Layman et al. 2006). As Domke and Cole (2008) argue, since Ronald Reagan's 1980 nomination acceptance speech that brought religion to the forefront of political campaigns, religion is now somewhat of a political test for Republican politicians. Overall, this literature argues that religion appears to matter to a greater extent to Republican Party identifiers than to Democratic Party identifiers. This

conclusion is bolstered by findings in the literature that the Religious Right has been very influential in mobilization campaigns in support of Republican Party candidates (Regnerus, Sikkink and Smith 1999; Calfano and Djupe 2009) and that religious activists are heavily concentrated in and represented by the Republican Party (Layman 2010). Accordingly, I hypothesize that the linkage between religious commitment and partisan strength is conditional upon the particular political party.

Hypothesis 2: The effects of religious attachment on partisan strength may manifest at different levels for Republicans and Democrats. Specifically, the effects of religious attachment will be greater for Republican partisans than Democratic partisans.

3.4 The 2008 Election

In order to test the theory that religious attachment affects partisan strength, I choose to focus on the recent 2008 presidential election. Bartels (2000) and other scholars are finding the effect of partisanship on elections has increased over the past several years, making the most recent presidential election ripe for analysis. Using a recent presidential election cycle also avoids some of the biases that are present during non-presidential election cycles (systematic differences regarding interest, knowledge, participation, etc.). Additionally, because both parties' presidential candidates were viewed skeptically when it came to religion in 2008 there is less of a chance that respondents will be identifying with the candidates' religious attachments rather than the parties' as a whole. Another reason for choosing the 2008 election is that the existing research looking at the question of what causes strength of partisanship in the United States analyzes data through the 1992 election (Cassel 1993). Therefore focusing on later

elections provides an updated analysis and can see if there are differences or changes in the influences on partisanship strength in recent elections.

The choice of analyzing data surrounding presidential elections is also a strategic choice, as I argue that theoretically speaking the importance of partisanship and partisanship strength in particular is found in its relation to elections. What is of interest to scholars is discovering how partisanship affects politics. One primary aspect of this is elections, as partisanship affects levels of turnout as well as voter choice and so is extremely consequential in determining election outcomes. Furthermore, many of the normative implications of partisanship are only apparent with regards to elections as they deal primarily with voter turnout. Literature also suggests that the concept of party identification as a stable and long lasting construct (Campbell 1960) continues to be the case in politics today (see Green and Palmquist 1990; Green et al. 2002), with the implication that there should be little difference in reported partisanship for election and non election years.³

3.5 Data

Accordingly, I use the Henry Institute's 2008 pre-election (April & May 2008) and post-election (November 2008) data sets from their Religion & Civic Engagement

³ As Cassel (1993) notes, the revisionist theories of party identification which argue that party identification (focusing primarily on direction but the arguments can be extended to include intensity) is affected by short term influences poses a challenge to identifying the causes of partisanship strength (see Franklin and Jackson 1983; Mackuen, Erickson and Stimson 1989; Erickson, MacKuen, and Stimson 2002). However, she notes that these studies show that short term influences have only a marginal substantive impact on party identification and other scholars, notably Green and Palmquist (1990), argue that even these findings are incorrect when measurement error in taken into account. Therefore, enough doubt is cast in the literature to argue that party identification is not subject to short term influences and consequently I have chosen not to account for these influences in my analysis.

Project. This data set, despite having a much smaller n than studies such as the National Annenberg Election Surveys (NAES), were judged to be superior for the purposes of answering this research question since the data sets included questions that better get at a sense of religious attachment.⁴

Dependent Variable

As discussed, the research question that I am interested in is determining what mechanisms are at work influencing the strength of party identification of citizens—in particular, whether religious commitment affects strength of partisanship. Therefore, my dependent variable is partisan strength. Party identification is traditionally measured as a seven-point scale (Campbell et al. 1960) including categories for strong and weak Republicans and Democrats, independents who lean toward either the Republican or Democratic Party, and pure independents. Though there is debate in the literature concerning whether this measure of party identification is an accurate portrayal of citizen partisanship (see Weisberg 1980 and the dimensionality of the measure), the general consensus is that it is an accurate measure and as a consequence it is the one traditionally used by scholars (see Cassel 1993).

For the present analysis, however, I collapse the seven-point scale of party identification into a dummy variable encompassing pure independents, independent leaners, weak partisans (Republican or Democrat), and strong partisans (Republican or Democrat).⁵ Thus, for the present project I test three dependent variables, 1) strong partisans versus

⁴ I do test my theory using the 2000, 2004, 2008 NAES data. The proxy for religious attachment included in those surveys is religious attendance. Using this variable, I do find support for both of my hypotheses.

⁵ I exclude respondents who identify with third party candidates (only 2-3 percent of the data). I also exclude responses of don't know or know response (also roughly 2-3 percent of the data).

everyone else (weak partisans and independents), 2) strong Republicans versus weak Republicans, independents leaning Republican and pure independents and 3) strong Democrats versus weak Democrats, independents leaning Democratic and pure independents.⁶

Religious Attachment

The key independent variable of interest is religious attachment or religiosity. To measure this concept, I built a composite score using four different questions from the Henry Institute data. First, respondents were asked whether they believe it best to keep one's religious beliefs private; answers were provided on a five-point scale and those who strongly agreed with the statement were coded as being less religiously attached (=1) than those who strongly disagreed (=5). Second, respondents were asked whether there are absolute standards of right and wrong; again, answers were provided on a five-point scale, only this time those who strongly agreed were coded as being more religiously attached (=5) than those who strongly disagreed (=1). Third, respondents were asked whether they agreed with the statement that all the great religions are equally good and true; on the five-point scale those who strongly agreed were coded as being the least religiously attached (=1) and those that strongly disagreed were coded as being the most religiously attached (=5). Finally, respondents were asked how often they attend religious services; again, this question was coded along a five-point scale wherein those who never attend were coded as the least attached to religion (=1) and those who attend more than once a week were coded as the most

⁶ All variable coding is available upon request.

religiously attached (=5).⁷ I combine these four questions to create a composite score of religious attachment.

Ideology

Although social identity theory is a growing body of literature in the social sciences, there are some who remain skeptical about its explanatory value. Contrary to proponents of social identity theory, Abramowitz and Saunders (2006) suggest that it is merely political ideology that leads voters to develop attachments to political parties, not their social identities as defined by their group memberships. In such an account, issues clearly matter more than social identification. I have aimed to counteract this claim by including a control variable for political ideology that is built using three separate questions within the Henry Institute data. The use of three separate questions is necessary since the data set does not actually include a measure of ideology. Therefore questions measuring respondents' preferences regarding the environment, poverty and immigration are used to build a composite score measuring respondents' ideological persuasion. Noticeably absent from this composite score are issues that are religious in nature since I distinctly try not to confound variables and the inclusion of largely religious issues like abortion and gay marriage might do just that.⁸

⁷ In between the "never" and the "more than once a week" options were those who attend a few times a year (coded=2), those who attend once or twice a month (=3) and those who attend each week (=4).

⁸ I did include a composite of religious issues in an earlier model (abortion, gay marriage and public displays of the Ten Commandments) but the results were robust so I kept it out of the final model to keep it more parsimonious (Achen 2002). Also, although the non-religious issues that I chose may be related to the teachings of a religious doctrine, I do not believe these to be inherently religious issues, nor do I believe the general public forms their preferences on these issues out of devotion to a particular religious doctrine.

Age and Previous Political Participation

Beginning with Converse (1969; 1976), previous research identifies two key variables that affect partisan strength, age and previous political participation. As citizens age, they develop loyalty to the nation and so loyalty to their political party making them more likely to identify as strong partisans. Similarly, as citizens engage in the political process, especially through voting in elections, they become "bound" to the political system and so to a political party again making them more likely to identify as strong partisans. As Cassel (1993) notes, Converse (1969; 1976) argues that the "binding in" effect of voting in elections comes from voting for a particular party not from the experience of voting per se. Accordingly, I control for these two possibilities by including a variable for the respondent's age and for voter turnout in the previous election, in this case 2004. For the Republican and Democratic only models I include vote choice in 2004 rather than voter turnout.

Additional Independent Variables

To ensure that the results are not spurious, I include the following independent variables in the model: marital status, education, income level, political knowledge, political interest, and social capital variables—volunteering, length of residence in their community, and discussion of politics.⁹ Increases in age, as conceptualized by Converse (1969) result in greater acceptance of the political system and hence greater attachment to the political parties. Therefore, those higher in age should exhibit greater attachments to the political parties and so greater amounts of partisan strength (since this is a cross sectional study). Finally, Campbell et al. (1960) and later scholars (Wolfinger and Rosenstone 1980) find that

⁹ For all of the independent variables, including religious attachment, I exclude responses of don't know or no response. I treat these responses as missing data.

income and education are strongly related to voter turnout. This is at least in part because these citizens have greater amounts of social capital and have a greater stake in the political system. This line of reasoning can be extended to strength of partisanship, as in this case those high in income and education should be more likely to identify strongly with a particular party.¹⁰

Statistical Method

Because I collapse the normal 7-point party identification scale into a dummy variable, where respondents are coded as either being strong partisans or not (weak partisans and independents), the analysis is presented using logistic regression. I estimate three models corresponding to the three dependent variables: partisan strength with both parties included, one examining just Republicans and one examining just Democrats. Since MLE coefficients are not directly interpretable, I estimate predicted probabilities in order to understand the substantive effects of the independent variables, in particular religious attachment, on partisan strength.

3.6 Results

As reported in Table 3.1, the results provide support for both of the hypotheses. First, when examining all respondents regardless of party affiliation, I find that religious attachment is positively related to strength of partisan attachment. As respondents become more attached to their religion they increase their likelihood of identifying as a strong partisan. I argue that this provides support for the social identity theory argument,

¹⁰ One important variable that is missing from the model is the partisanship of the respondent's father. Unfortunately, this question was not asked in either the NAES, the NES or the Henry Institute surveys. It was only asked in the NES surveys through 1980. Cassell (1993) finds, in examining strength of partisanship, that her model fit drops dramatically after 1980 from previous presidential elections when she can no longer include father's partisanship (because the NES ceased asking that question).

that as citizens become more attached to their religion they develop an us vs. them mentality that has a spillover effect positively affecting their level of partisan attachment.

As reported in Table 3.2, the actual substantive effect of religious attachment on partisan strength is quite large. Moving from the minimum level of religious attachment (no attachment) to the maximum level of religious attachment (high attachment) holding everything else at the mean leads to a 17 percent change in a respondent's level of partisan strength.

This result for religious attachment is robust given the control variables. As expected, previous voter experience is positively related to partisan strength and it produces the largest substantive effect, a 22 percent change moving from not voting in 2004 to voting in 2004. Interestingly, I find that variables of political interest are positively related to strength of partisanship while social capital variables are insignificant. Both talking about politics and expressing an interest in politics increases the likelihood that a respondent will identify as a strong partisan. Strikingly given the literature, in the overall model, I find that both age and ideology are insignificant predictors of partisan strength.

However, the results of the overall model come into greater focus when I examine Republican and Democratic respondents separately (columns 2 and 3 in Table 3.1 and Table 3.2). The second hypothesis is confirmed that the effect of strength of religious attachment on strength of partisanship is conditional upon the perceptions respondents' hold of the two political parties. As strength of religious attachment increases, the likelihood of identifying as a strong Republican increases but it does not affect the strength of partisanship for Democratic respondents. The substantive effect is quite large.

Moving from no religious attachment to the highest level of religious attachment leads to a 41 percent change in the likelihood of identifying as a strong Republican. This is the largest substantive effect produced by any of the models. This result provides support for the literature which argues that the Republican Party is the party that is 'friendlier' to religion. It appears that the perception of the Republican Party being more attuned to religious interests is clear in the minds of respondents. This result may indicate that the Republican Party may be able to play a role in using social identity to their favor by continuing to frame their party as the party that is friendly to religion and so continue to secure the religious base.

Again these results for religious attachment are robust given the control variables. Interestingly, ideology is a significant predictor of partisan strength when the two parties are analyzed separately. Ideology influences both Republicans and Democrats, with an increase in ideology increasing the likelihood of identifying as a strong partisan. Another interesting finding is that political interest variables matter for Democrats but not for Republicans. Talking about politics and following politics has no affect on strength of Republican Party identification but increases the likelihood of identifying as a strong Democrat. As in the overall model, social capital variables are not related to partisan strength.

Overall, these findings suggest that religious attachment, via group attachment theory and social identity theory, can have an effect on partisan strength but not always since perception (of the party) appears to be a mitigating circumstance. Thus, future extensions of this analysis would need to find a way to measure this perception. For the current analysis, the proxies available within the Henry Institute data are not entirely

satisfactory and so they are not included. Also, this explanation might help explain why Democratic respondents relied on more cerebral aspects of politics—political knowledge and follow politics variables held significant for Democrats but not Republicans—since their psyches would not be tapped by the emotional effects of group attachment theory and social identity theory if they do not link their own religious attachments to those of their party in general.

3.7 Conclusion

The authors of *The American Voter* make clear that both direction and strength of party attachment are important predictors of the political attitudes and behavior of citizens. Strength of partisanship in particular is associated with normatively desirable outcomes, including increased voter participation and increased attachment to the political system (leading to political stability). However, despite the critical implications of partisan strength, a great deal less research is conducted identifying the causes of the strength of citizens' attachment to political parties than is spent understanding the development and direction of party identification.

Scholars investigating this question find two principle causes of partisan strength, the intergenerational transmission of partisan ties and the length of attachment to the political system, including how long citizens have lived under the system (their age) as well as their levels of political participation and engagement with the political system. In the current analysis, I identify another principle influence on partisanship strength, religious attachment. I find that religious attachment positively affects the probability that Republicans will identify as a strong partisan, but not Democrats. I argue that this provides support for the idea that the Republican Party is perceived as being 'friendlier'

towards religion. They have been successful at using social identity to their advantage to portray themselves as the party for religion. However, further research is necessary to fully understand why religious strength affects Republican Party identification and not Democratic Party identification. I suspect that variables measuring respondents' perceptions of the parties' religious attachments would be instrumental in better understanding this divide. Perceptions are an important part in the way group attachment theory and social identity theory take root (or fail to take root) and due to the social capital variables' lack of significance, furthering the research that specifically pertains to these explanations is likely to yield the most interesting and significant results in the future.

Further, over time analysis is important to see if there are changes in the influences of partisan strength, especially with regards to religious attachment, as events in the past several decades, notably the rise of the Religious Right, may have made religion more salient to the political lives of citizens. Relatedly, there may be different effects of religious attachment to strength of partisanship across the religious groups in America. In particular, the Religious Right targets conservative and evangelical Christians, indicating that religion may be the most salient to the lives (political and non-political) of these citizens and so they may identify more strongly with a political party (primarily the Republican Party see Layman 2001).

The desirability of an electorate that is strongly tied to political parties is clear, resulting in increased citizen involvement in government and increased political stability. Therefore, identifying the causes of partisan strength is and will continue to be an

important research agenda for scholars of political behavior and of party identification in particular.

	All Respondents Model	Republican Model Coefficient (SE)	Democrat Model Coefficient (SE)
Variable Religious Attachment	$\frac{\text{Coefficient (SE)}}{0.046 (0.012) ***}$	0 135 (0 018)***	-0.008 (0.017)
Kengious Attachment	0.040 (0.012)	0.133 (0.018)	-0.008(0.017)
Age	0.006 (0.003)	0.011 (0.005)*	0.007 (0.004)
Vote 2004*	0.958 (0.154)***	1.079 (0.260)***	1.018 (0.193)***
Ideology	0.001 (0.015)	-0.140 (0.022)***	0.127 (0.021)***
Political Knowledge	-0.024 (0.047)	0.070 (0.069)	-0.188 (0.064)**
Talk Politics	0.125 (0.043)**	0.112 (0.069)	0.116 (0.057)*
Follow Politics	0.309 (0.070)***	0.105 (0.103)	0.315 (0.093)***
Volunteer	0.275 (0.094)	0.104 (0.136)	-0.055 (0.126)
Length of Residence	0.003 (0.003)	0.003 (0003)	0.003 (0.003)
Marital Status	-0.179 (0.091)*	-0.031 (0.131)	-0.277 (0.127)*
Education	-0.017 (0.034)	0.016 (0.048)	-0.046 (0.046)
Income	0.027 (0.027)	0.056 (0.038)	-0.010 (0.038)
Constant	-3.458 (0.353)***	-4.217 (0.545)***	-3.587 (0.453)***
Log Likelihood	-1554.269	-768.409	-868.596
$LR \chi^2$	180.78***	231.11***	134.39***
Pseudo R^2	0.055	0.131	0.072
Number of	2393	1332	1386
Observations			
Reduction in Error	11.81%	20.00%	9.79%

 Table 3.1. Logistic Regression of the Effects of Religious Attachment on Partisan

 Strength

Note: Dependent variable for All Respondents Model is Party Identification 1= strong partisan 0= weak, leaner, independent; For Republican only Respondents 1=strong Republican 0=weak Republican, independent leaner, independent; For Democratic only Model 1=strong Democrat 0=weak Democrat, independent leaner, independent; SE= Standard Error

*Vote2004 is voter turnout for All Respondents model and vote choice for Republican and Democrat Models

* p-value < .05 ** p-value<.01 *** p-value<.001

	All Respondents	Republican	Democrat
Variable	Model %	Model %	Model %
Religious	.176	.417	
Attachment			
Age		.189	
Vote 2004	.217	.208	.218
Ideology		356	.338
Political Knowledge			138
Talk Politics	.120		.111
Follow Politics	.214		.205
Volunteer			
Length of Residence			
Marital Status	044		066
Education			
T			

Table 3.2. Predicted Probabilities for the Logistic Regression of the Effects ofReligious Attachment on Partisan Strength

Income --- --- ---Note: Predicted Probabilities are calculated by setting the variables at their mean and moving the variable of interest from its minimum value to its maximum value

Chapter 4

Judicial Decision Making in state Supreme Courts: Do Justices Respond to Religious Opinion?

The concept of judicial independence is firmly entrenched in American Democracy. The judiciary at the federal level is a nonelected body, being the branch of government most removed from the influence of the public. The judicial branch was designed to be an impartial arbiter of the law, yet as recent work indicates, justices, especially at the level of the US Supreme Court, often vote their ideological preferences (Segal and Spath 1993, 2002). At the same time, other work is examining a key question regarding the nature of judicial decision making, whether justices respond to public opinion when making their decisions. Though this seems counterintuitive given the nature and design of the judicial branch, scholars are beginning to show at the federal level that justices do respond to and reflect changes in public opinion (Caldeira 1991; Mishler and Sheehan 1993; Flemming and Wood 1997; McGuire and Stimson 2004; Giles et al. 2008).

At the state level, the potential influence of public opinion on judicial decision making is more apparent. The practice of electing judges to state Supreme Courts indicates that the same direct representational link between Congress and the public exists for justices as well. As a result, elected judges may take public opinion into account when making decisions, as scholars have indeed found to be the case in the area of the death penalty (Brace and Boyea 2008).

However, a related question that has not been examined by the literature is whether justices are influenced by specific groups within the public and by religious groups in particular. Congressional literature examining the representational linkage finds that Congressmen provide substantive representation in relation to the size or proportion of groups within their geographic or reelection constituencies (Griffin and Newman 2005; Hill, Leighley, and Hinton-Andersson 1995). In particular, Smith et al. (2010) finds that Congressmen provide substantive representation for religious constituents within their state depending upon the proportion of the religious groups within the state. While work on state Supreme Court judicial decision making finds that the religion of justices affects their decisions (Songer and Tabrizi 2000; see also Tate 1981 for the US Supreme Court) it has not examined whether religious groups within states affects judicial decisions, in states where justices are either elected or nonelected. Therefore, this analysis seeks to answer the question, does the presence of religious groups in states influence the decision making of justices on state Supreme Courts?

4.1 Public Opinion and Judicial Decision Making

The majority of literature that examines the linkage between public opinion and judicial decision making focuses on the US Supreme Court. However, recent developments in measuring public opinion in the states (see Erickson, Wright, McIver 1993; Berry et al. 2000) resulted in scholarly work examining this linkage at the state Supreme Court level as well. Collectively, this literature identifies three potential casual pathways for a direct link between public opinion and judicial decision making in state Supreme Courts: strategic behavior to ensure implementation of decisions and to protect

judicial legitimacy, attitudinal change reflecting changing public mood and the direct election of justices.

The first mechanism at work causing justices to respond to public opinion is the strategic behavior of justices to both ensure proper implementation of its decisions and to protect the perceived legitimacy of the court (McGuire and Stimson 2004; Giles et al. 2008). Hamilton wrote in *Federalist* 78 that the judicial branch was the "least dangerous" branch of government, because it has "neither force nor will, but merely judgment" and it has "no influence over either the sword or the purse." As a result of its inability to enforce its decisions, the court looks to public opinion to ensure proper implementation. They do this indirectly, through the implementation of its decisions by the other branches of government and directly, through maintaining legitimacy in the eyes of the public.

At least since Murphy (1964: 171), scholars have theorized that the court must act strategically, anticipating that its decisions will not "stir a political reaction" thereby potentially threatening the implementation of its policy decision. Building on the concept of strategic anticipation, McGurie and Stimson (2004, 1022) note that "the Court requires the cooperation of legislative and executive officials, many of whom are themselves careful auditors of mass opinion." Therefore, the court takes into consideration whether a given policy will be adequately implemented by the other branches of government, framing the policy in such a way to ensure the cooperation of the other branches.

However, as McGurie and Stimson (2004) note, the primary consideration of the other branches is whether their constituency would agree with the Supreme Court decision and so whether their chances of reelection would be harmed. As a result, public

opinion is indirectly linked to judicial decision making, through the linkage of the public to the officials who the court depends upon to enforce its decisions. At the same time, there is also a direct link between public opinion and judicial decision making in the desire and necessity of the court to be perceived as legitimate in the eyes of the public in order for its decisions to be faithfully implemented and adhered to. Legitimacy theory argues that in order for the judiciary to remain effective, it must be viewed as legitimate in the eyes of the public, with judges acting as impartial arbiters of the law (Friedman 1998; Gibson 2008). If the court is no longer perceived as legitimate, citizens will be unwilling to listen to and follow the decisions that it renders. The necessity and desire of justices for institutional legitimacy is especially evident in a cross national perspective (see Haynie 2003; Widner 2001). Therefore, justices act strategically to protect their legitimacy, especially knowing that their policies have the greatest effect when popular support is on their side (Rosenberg 1991).

The second casual pathway linking public opinion with judicial decision making is an attitudinal change explanation (Giles et al. 2008) whereby justices reflect overall changes in public mood. In part, this is tied in with the above description of judicial legitimacy. A component of legitimacy theory is the idea that judges cannot stray too far from the opinion and norms of the public, either as reflected in the composition of the court (see feminist legal theory Wilson 1990 for example) or in the opinions and judgments that are rendered (Friedman 1998). If the court does stray too far from societal norms, their legitimacy, the notion that they are impartial and so can adequately ensure that justice is served, can potentially be harmed. Therefore, justices are likely to follow overall changes in public opinion and public mood.

However, a more recent explication of this mechanism comes from Giles et al. (2008) in the notion that the preferences of justices, like other actors, may be shaped and molded by societal forces occurring in their environment (also see Mishler and Sheehan 1996; Flemming and Wood 1997). The idea that judicial attitudes are not permanent but can shift with changes in society is supported in the literature, with the finding that some Supreme Court justices have exhibited a substantial shift in attitude during their tenure on the court (Baum 1988; Ulmer 1973). More recently, scholars are arguing that shifts in judicial attitudes are far more common than is supposed under the attitudinal model (Epstein et al. 1998; Epstein et al. 2007). These shifts in judicial attitudes provide an indication that forces outside of the justices may be causing them to change opinion. As Giles et al. (2008: 295) write, "the attitudinal change explanation suggests that the observed direct linkage between public opinion and the behavior of justices arises from the force of mutually experienced events and ideas in shaping and reshaping the preferences of both the public and the justices." Justices are humans like everyone else and are subject to the same societal forces and changes as the public.

The third mechanism by which public opinion is reflected in judicial decision making is specific to state Supreme Courts, the method of replacing justices by election. Justices who face election (and reelection) may "have their eyes on the ballot box" responding to changes in public opinion to ensure that they remain in office (Brace and Boyea 2008: 361; Huber and Gordon 2004). This view is widely held among observes of state courts, but has received relatively little scholarly attention, largely a result of the lack of adequate measures of public opinion in the states (Brace and Boyea 2008).
More recently, judicial elections are increasingly of concern to scholars and observers of state Supreme Courts alike as the visibility of judicial elections has increased over the last two decades. As well, the US Supreme Court ruled in 2005 (*Republican Party of Minnesota v White*) that judges have free-speech rights allowing them to declare their policy positions during campaigns, something they were prohibited from engaging in before in many states (the commit clause). This set off concern that judicial legitimacy would be harmed by the ability of justices to declare how they would decide in cases as well as the ability to engage in negative advertising. Gibson (2008), in a survey in Kentucky using hypothetical situations where justices do engage in such practices as negative advertising, finds initial evidence that the legitimacy of the court is harmed by justices engaging in these campaign tactics.

As a result of these recent developments as well as the development of accurate and reliable measures of public opinion in the states, more recent research has begun to examine the link between judicial elections and judicial decision making. Empirically, scholars are finding evidence that elected justices are swayed by public opinion, specifically in the issues of sentencing and the death penalty (Huber and Gordon 2004; Brace and Boyea 2008). Further, the literature is providing theoretical clarifications on the linkage, searching for conditions under which elected justices pay attention to and reflect public opinion. In particular, Brace and Boyea (2008) argue that the linkage can be informed by the literature on elite responsiveness to public opinion. These theories argue that when the issues are complex, or when they are reflective of party cleavages (opinion among the public is divided) public opinion may be influenced by elites, policy or other events. In these situations, elites are unlikely to follow public opinion, as it gives

them no clear direction or advantage among the population. However, when issues are simple, there is convergent opinion cutting across party lines, elites are more likely to be responsive to public opinion (Hurley and Hill 2003).

4.2 Group Influence Model of Judicial Decision Making

Through these three mechanisms, it is clear in the literature that justices listen to and reflect public opinion, though it is often unclear as to which mechanism is at work (see especially Giles et al. 2008 for a discussion). However, what has been unexplored in the literature is whether justices may listen to specific group opinion rather than public opinion as a whole. I seek to elaborate on this linkage, looking at it both in general terms as well as focusing on the particular case of religious opinion.

At least since Truman (1951), a paradigm in political science by which to conceptualize American politics is through group identity and the related concept of pluralism. Truman (1951) argued that all politics is carried out in terms of groups and that these groups have multiple access points in their ability to influence government, the concept of pluralism. However, this was not a new concept. Madison in *Federalist 10* warned against the mischief of faction, arguing that it would fragment the newly united country. Yet, by the time Alexis de Tocqueville studied American democracy, he observed that "in no country in the world has the principle of association been more successfully used or applied to a greater multitude of objects than in America" (191). While many groups are not inherently political, as they are not organized for political purposes, they often take on political roles and become involved with political issues (Truman 1951).

The continued importance of groups to American politics is evidenced in the literature at both the mass and elite level. At the mass level, political participation, especially voting behavior, has often been explained in terms of group dynamics and group behavior. Going back to the seminal study *The American Voter*, Campbell et al. (1960) argue that groups shape the politics of their members as a result of the psychological attachment that members feel for the group. More recent scholarship bears out this proposition. In particular, scholarship on union membership indicates that the stronger the attachment of members to a union the more likely they are to support union backed candidates (Rapoport, Stone and Abramowitz 1991; Clark and Masters 2001). More recently, Lewis-Beck et al. (2008) employ a model of group influence to explain mass voting behavior in the 2000 and 2004 election in their revisitation of the Campbell et al. (1960) work.

The importance of groups to American politics is also evidenced at the elite level. As noted earlier, Congressional literature examining the representational linkage between citizens and government find that Congressmen provide substantive representation in relation to the size or proportion of groups within their geographic or reelection constituencies (Griffin and Newman 2005; Hill, Leighley, and Hinton-Andersson 1995). Accordingly, at least since Pitkin (1967), the concept of representation has been tied to groups. For example, descriptive representation is premised on the idea that various groups in society should have members of their group representing them in Congress.

While it is evident that groups influence politics and public opinion at the mass level and the elite level, specifically concerning the behavior of Congressmen, I argue that a model of group influence can also be applied to the decision making behavior of

state Supreme Court justices. I further argue that this occurs in three ways, through the election mechanism, through the presence of dominant groups in a state and through the intensity of political beliefs held by groups in a state.

The first mechanism by which groups may influence the decision making of state Supreme Court justices is through the mechanism of judicial elections. As seen earlier, justices who are elected to office respond to public opinion on specific issues, especially the death penalty (Brace and Boyea 2008). Borrowing from Congressional literature, justices who are elected to office will cater to the group or groups that are needed for reelection (see especially Fenno 1978). As Brace and Boyea (2008) state, "elites fear losing office" including justices to the Supreme Court. More recent Congressional literature notes the linkage between substantive representation and the composition of a Congressmen's district or state. Congressmen actively seek to provide representation to groups within their constituency to ensure reelection (Griffin and Newman 2005; Hill, Leighley, and Hinton-Andersson 1995). I hypothesize that similar behavior occurs among justices who are elected to office.

The second mechanism by which groups may influence judicial decision making is through the presence of dominant groups in a state who in turn influence the societal norms of the state. As noted earlier, one pathway by which justices respond to public opinion is through "attitudinal change" (Giles et al. 2008) or changes in societal norms and opinions. Mishler and Sheehan (1996; 175) argue that the attitudes of justices may change as a response to either "fundamental, long-term shifts in the public mood or to the societal forces that underlie them." I argue that a potential clarification is needed on this theory. It may be that justices are indeed influenced by societal changes but that overall

societal changes are mediated by dominant groups within society. In other words, a dominant group within society may be the actual influence behind a shift in public mood.

Accordingly, the third mechanism by which judicial decision making may be influenced by group opinion is the idea that justices may listen to groups in a state that hold intense opinions. Democratic theorist Robert Dahl (1956) argues that politics is carried out with regards to "the relative intensity of preferences." In other words, American democracy institutes safeguards to ensure that the majority does not always win over the minority, especially if the minority holds an intense opinion. According to Dahl (1956), it is not simple majority minority rule which dominates politics, but the intensity with which the majority minority hold their opinions. An intense minority or majority is able to gain access to government to get their voices heard.

I argue that this theory of democracy is applicable to judicial decision making. Justices may listen to groups in society that exhibit intensity in their opinions. A motivating factor to do so is that the potential consequence of failing to consider intensity of preference is loss of legitimacy in the eyes of the group. A prime example of judicial decision making taking into consideration intensity of preference is the issue of desegregation and the Supreme Court case *Brown v Board of Education* (1954).

4.3 Judicial Response to Religious Opinion?

It is evident that, in general terms, judicial decision making on state Supreme Courts may be influenced not simply by overall public opinion but by specific group opinion through the three mechanisms outlined above. However, I further argue that justices are likely to be responsive to religious opinion in particular when making their decisions, as religious groups fit each of the three criteria just described. Religious

groups, therefore, present a vital test of a group influence model on judicial decision making for three main reasons.

First, religious groups, notably the four Christian religions, evangelical Protestants, mainline Protestants, black Protestants and Catholics, are predominant groups in American society.¹¹ Added together, over three quarters of Americans identify with one of these four Christian religions. Aside from the size of these four religious groups, they also exhibit cohesiveness as a group (though some more so than others). The cohesiveness of the groups is well documented, for example, in the voting behavior literature, where scholars find that religious groups identify with particular political parties (see for example Harris 1995; Layman 2001; Classen and Povtak 2010) and it extends to stances on political issues. For example, black Protestants have consistently been shown to be extremely conservative on social and moral issues, as have evangelicals (Steendland et al. 2000; Campbell and Monson 2008). Therefore, this indicates that religious groups are able to send signals on their preferred policy to government. The literature reveals that Congressmen provide substantive representation for religious groups residing in their states, in part to ensure reelection (Smith et al. 2010). Accordingly, religious groups, as a result of their size and cohesiveness, are a potential

¹¹ Evangelical Protestants are distinguished by their strict interpretation of scripture, their emphasis on the need for each individual to accept Jesus Christ as personal savior, and their active efforts to spread their faith through evangelism to others. Mainline Protestants are more theologically—and increasingly, politically—liberal than their evangelical counterparts. Black Protestants, as noted, identify politically with the Democratic Party yet are extremely conservative on social and moral issues, opposing, for example, abortion and gay marriage. Catholics, though historically identifying with the Democratic Party, are increasingly moving towards the Republican Party, in part because of their conservative views on social and moral issues.

constituency that justices facing reelection may provide representation for in the direction of their decisions.

Secondly, also as a result of the size and cohesiveness of religious groups, it may be that they influence societal norms in individual states. Therefore, if justices follow societal norms in their decision making (Caldeira 1991; Mishler and Sheehan 1993; Flemming and Wood 1997; McGuire and Stimson 2004; Giles et al. 2008) then they may be following the norms of religious groups in their state and so make decisions following religious opinion.

Thirdly, for particular issues, notably social and moral issues, the literature indicates that religious groups hold very intense opinions. The Religious Right was formed in large part to give voice to the conservative stance of religious groups, notably evangelical Protestants, on social and moral issues (see Wilcox and Larson 2006; Wilcox and Robinson 2011). Therefore, religious groups are a group that justices may listen too due to the intensity of their preferences on particular issues.¹²

4.4 Hypotheses

I propose to examine an issue on which religious opinion should be likely to influence judicial decision making on state Supreme Courts, the death penalty. I chose to look at death penalty decisions for two primary reasons. First, this is an issue on which religious groups have coherent and cohesive opinion and hold their opinion very strongly.

¹² Another component of the strength of religious preferences is seen in the great variation across states in the intensity with which individuals adhere to their religion. In states where religious intensity is higher, where more individuals attend church for example, justices may feel more pressure to make decisions in line with religious opinion. While this is a distinct possibly, I hold off testing it due to lack of available data on the religious intensity of individuals within the states.

Second, previous research has found that public opinion affects judicial decision making on death penalty cases (see Brace and Boyea 2008). Therefore, this allows me to test my theory of group influence, that specific groups, in this case religious groups, are potentially part of the mechanism at work behind the influence of public opinion on judicial decision making. This allows for greater clarity and nuance than previous literature in understanding the role of public opinion in judicial decision making.

I focus primarily on the effects of the four Christian religions on judicial decision making on death penalty cases. The issue of the death penalty presents an instance of divided opinion among the religious groups. Evangelical Protestants are the most in favor of the death penalty as an acceptable form of punishment (Britt 1998). Both evangelical clergy as well as members overwhelmingly support the use of the death penalty. However, mainline Protestants and Catholics are less supportive of the death penalty than evangelicals. A number of mainline denominations, including the Episcopal Church and the Methodist Church, have made explicit statements against the death penalty as a form of punishment. As well, the Catholic Church does not support the use of the death penalty. Accordingly, a 2007 survey of mainline Protestant clergy found that that 66% of mainline clergy oppose the death penalty. However, for both mainline Protestants and Catholics, the rank and file members overwhelmingly support the death penalty, though showing less support than evangelicals (see the Pew Research Forum Survey 2007). Black Protestants are the least supportive of the death penalty as a form of punishment, partly because black men are disproportionately given the death penalty sentence (Young 1992). Because black Protestants are the most liberal on the issue of the

death penalty among the religious groups, they comprise the excluded category for the analysis.

Hypothesis 1: I expect a negative relationship to exist between evangelical Protestants, mainline Protestants and Catholics and the decision to reverse a death penalty sentence by state Supreme Court judges.

As noted, one potential pathway for groups, and religious groups in particular, to influence judicial decision making is through judicial elections. Justices that are elected to office may take greater heed of religious opinion as they "have their eyes on the ballot box."

Hypothesis 2: I expect a negative relationship to exist between evangelical Protestants, mainline Protestants and Catholics and the decision to reverse a death penalty sentence for state Supreme Court judges who are elected to office.

4.5 Research Design

Data, Case Selection and Dependent Variable

The primary data I use for this analysis is the State Supreme Court Data Archive (SSCDA) complied by Brace and Hall for the years 1995, 1996, 1997 and 1998.¹³ The case I will use to analyze whether state Supreme Court justices listen to religious opinion in the states is the death penalty. Scholarly work already indicates that elected justices deciding on death penalty decisions are responsive to state public opinion in general (Brace and Boyea 2008). However, religious groups, as noted above, offer a clear direction of opinion on this issue to which state Supreme Court justices can listen.

¹³ The SSCDA project was supported by the National Science Foundation grants SBR 9617190 SBR 9616891, SES-051849, SES-0516409 and SES-0516600. For additional information on the SSCDA data, visit the website http://www.ruf.rice.edu/~pbrace/statecourt/.

Therefore, the effect of public opinion on judicial decision making on death penalty cases may be mediated by the presence of religious groups in the state. The death penalty also presents a good case to test the hypothesis that the presence of religious groups affects judicial decision making because the four Christian religions diverge in their opinion on this issue. Therefore, this case should present a relatively easy test of the theory that state Supreme Court justices respond to religious opinion.

For judicial decisions on death penalty cases, a liberal decision will be coded as one if the death penalty if overturned and a conservative decision will be coded as zero if the death penalty is upheld. More specifically, this measure is constructed in the following way. A liberal decision occurs when the appellant in the death penalty case is an individual and the justices vote to overturn the decision (thereby overturning the death penalty sentence) and a conservative decision occurs when the appellant is an individual and the justices uphold the decision (thereby upholding the death penalty sentence). *State Religion Data and Method of Judicial Appointment*

In order to test for the effect of state religious populations on judicial decision making, I employ a measure of religious affiliation complied by Green (2007) for the year 2004. His religious data is generated from numerous statewide surveys and is generally regarded as one of the most precise measures of religious affiliation in the fifty states that exists (Smith et al. 2010). Although it is measured seven years after the last year in the SSCDA data, religious affiliation, like party identification, is a stable and long lasting construct, with few individuals shifting from one religious tradition to another.

I collapse the traditional seven category measure of religion (see Steensland et al. 2000) into five categories, evangelical Protestant, mainline Protestant, black Protestant,

Catholic, and other faith. I do this for two reasons. First, my theory of group, and in this case religious, influence on judicial decision making necessitates that there is a relatively large population for justices to listen to. If a group comprises only a small percentage of the population, say less than two percent as, for example, is typical for the Jewish or Muslim population, justices have little incentive to listen to them and so be responsive in their decision making to their opinion on particular issues. As the groups get larger, however, they gain in influence according to my theory. As a result, I keep the religious categories separate that do have relatively large populations across the United States (evangelical Protestant, mainline Protestant, Catholic and black Protestant as discussed earlier). The second reason that I chose to collapse religious categories is to have a more parsimonious model (Achen 2005). Therefore, I chose to collapse the categories of Jewish, Muslim, other Christian (which includes Mormons) and no religion into a single category of other faith. While this creates a less meaningful category, the advantages associated with collapsing the category outweigh this particular disadvantage.

The second main independent variable of interest is a measure of judicial method of appointment. Following Brace and Boyea (2008) I focus on methods of judicial retention rather than judicial selection. Justices who have been in office and face a retention election may, as argued earlier, "have their eyes on the ballot box" more so than justices who are facing election for the first time. They have developed a tract record on which voters and groups may base their decisions, either favorably or unfavorably. Therefore, using the judge level dataset added to the Brace and Hall data, I look at two types of retention mechanisms, whether justices were appointed (by either governor or state legislature), or whether justices were elected (either partisan, nonpartisan or

retention election). This is coded as 1 for either election or retention and 0 for justices that were appointed to office.

Control Variables

To ensure that the results are not spurious, I include a number of control variables that may affect judicial decision making. These variables are measures of citizen ideology, ideology of the court, individual judge characteristics and case and state specific factors.

The ideology of the citizens of the states as well as the ideology of the justices of the states must be controlled for, as judicial decision making could simply be reflecting the overall ideology of the state or of the ideology of the judges (in the framework of the attitudinal model). Citizen and justice ideology are obviously connected to judicial decision making, justice ideology through the attitudinal model and citizen ideology as argued in this paper. To capture state Supreme Court judge ideology, I use the PAJID scores developed by Brace, Langer and Hall (2000). These scores integrate the partisan affiliation of the judges with the citizen ideology at the time of their selection (using the Berry et al. (2000) scores for citizen ideology I use the Berry et al. (2000) scores. These measures. As a measure of citizen ideology I use the Berry et al. (2000) scores. These measures are coded from 0-100 for judge ideology and 0-100 for citizen ideology with 0 being extremely conservative and 100 being extremely liberal.¹⁴

¹⁴ These two measures, though there is some overlap in how the measures are constructed, are correlated at less than .6. Though this is slightly high, it is not correlated enough to present problems with multicollinearity. I can substitute the Berry et al. (2000) citizen ideology scores with the Erickson, Wright McIver (1993) scores. However, this is not as adequate a solution, as not all states are included in the EWI scores. When running an analysis with the EWI scores similar results are obtained. There is the further consideration that using the judge ideology scores and the citizen ideology scores in the

The second set of variables I control for are judge specific characteristics that affect their decision making. Recent work reveals that judge gender and age can affect the decisions that they reach across a wide variety of issues (see for example Songer and Crews-Meyer 2000 for a discussion of the effect of gender on state Supreme Court decisions). Gender is coded as a dummy variable with one being a male and zero being a female and age is left as a ratio level variable.

The final set of variables I control for are case and state specific variables. Following Brace and Boyea (2008) I include a control for whether there is a public defender in the death penalty case. The quality of council is a major issue in many death penalty cases, with the public defender being less resourced to defend their client. Therefore, it is likely that in cases with a public defender judges may find greater room for reversible error in the case.¹⁵ This variable is coded as one if a public defender is present and zero otherwise. I also control for state specific characteristics though my modeling strategy detailed below.¹⁶

same model is problematic since the judge ideology measure incorporates the Berry et al. (2000) citizen ideology scores as part of its construction. However, the two variables are correlated at .38, therefore not necessarily presenting a problem of multicollinearity.

¹⁵ I also include a variable for whether the state enacted a commit clause or not, a proxy for campaign restrictions. The hypothesis here is that where restrictions on the speech of justices exist, they should be more insulated from the effects of public opinion (see Brace and Boyea 2008). In all of my models it does not reach statistical significance and does not change the overall model. For the sake of parsimony I exclude it from the models that are presented in this analysis.

¹⁶ An important control variable that the literature is missing is closeness or proximity to the election. To my knowledge, none of the literature on judicial elections control for how close a justice is to reelection. There is strong Congressional literature to suggest that Congressmen change their behavior when an election nears in hopes of shoring up constituent support. I also do not control for this for lack of data. However, I will collect this to include it in future analysis on this topic and research question.

Methodology

Because the outcome variable is dichotomous (judge vote) and the data includes both pooled cross-sectional and time-series elements, I use binomial generalized estimating equation (GEE) analysis with logit as the link function to estimate my models (see also Brace and Boyea 2008). More specifically, due to the cross-sectional nature of the data, there is possible interdependence between justices when they make their decisions that is unaccounted for in the model as specified. Accordingly, in order to account for this possible interdependence I use GEE clustered on the court case. GEE allows for more precise estimation of data involving this type of conditional interdependence (Zorn 2006; Brace and Boyea 2008). Additionally, in order to control for any possible temporal effects (the data covers four years) I control for the decision year in the GEE analysis.

In analyzing the effects of religion on state Supreme Court decision making, I conduct three models. First, I look at the effects of the percentage of religious traditions in the states controlling for the variables described above, excluding the measure of judicial retention, with black Protestant as the excluded category. Second, I include the measure of judicial retention to the first model. Third, I include interaction effects between judicial retention and state ideology and the religious groups that are statistically significant in the first two models (evangelical and mainline Protestant) in order to test the hypothesis that the influence of religious groups on judicial decisions making on death penalty cases is found in the mechanism of judicial election.

4.6 Results

Table 4.1 presents descriptive statistics of the number of death penalty cases that fall within each of the three judicial retention mechanisms for the years 1995-1998 as well as showing the percentage of the religious groups and their range residing in the 50 states. The overwhelming majority of death penalty cases were decided in states that have either judicial elections or retention elections as their method of retention for Supreme Court justices. As evidenced, only a small minority, five percent, of death penalty cases were decided in states where justices are appointed to office. The other striking observation is that the rest of death penalty cases are split fairly evenly between justices that are retained through elections and through retention elections.¹⁷

Looking at the percentage of religious groups in the states, evangelical Protestants comprise the largest group (26 percent) followed by Catholics (22 percent) and then mainline Protestants (21.5 percent). In all three cases, these religious traditions make up individually over one fifth of the population in the states, confirming the assertion that they represent large segments of the population in the 50 states and as a result may be groups that state Supreme Court justices are likely to pay attention to when deciding

¹⁷ Previous research uses a dichotomy of appointed and non-appointed when examining the effect of judicial elections on judicial decision making. However, it may be that there is a difference in decision making between justices who are retained through elections against a challenger (though a challenger may not always be present) and being retained through retention elections where they do not face a challenger. The literature finds that there are substantial differences between judicial elections and retention elections, especially that retention elections are not as insulated from the public and outside forces as often assumed (Hall 2001) though it has not been looked at with regard to public opinion and religion. In future research I plan to develop a theoretical expectation for why there would be a difference between elected and retained justices and the influence of public opinion and religious opinion in particular on their decision making and test it empirically.

cases. However, while it is clear that religious traditions represent large groups within the population of the states, it remains to be seen whether their presence actually affects judicial decision making, specifically in cases involving the death penalty.

Table 4.2 presents the results of the GEE logistic regression analysis of the effect of religion on judicial decision making in death penalty cases. The first column looks at the effect of religion on death penalty decisions without controlling for the mechanism of judicial retention. Here the results show a negative relationship as hypothesized between the religious traditions and the decision to reverse a death penalty sentence. In other words, the presence of evangelical and mainline Protestants in the states, compared to black Protestants, does significantly affect the decision of justices to vote in a conservative fashion on death penalty decisions (a vote to uphold the death penalty sentence). Looking at the control variables reveals that more liberal justices are more likely to vote to reverse death penalty decisions (confirming the literature), the presence of a public defender also influences justices to vote to reverse death penalty decisions, indicating that justices may indeed find more room for reversible error when a public defender is present, and male justices are more likely to vote to reverse death penalty decisions than females. An interesting non-finding here is that citizen ideology fails to reach statistical significance.

Including a control for judicial retention (column 2) is not statistically significant and it does not change the sign or significance of the variables in the model. However, the true test of whether the mechanism of judicial elections influences votes to reverse death penalty decisions is found in the interaction of judicial retention with citizen ideology and the religious traditions.

Accordingly, column 3 presents the results of the effects of religious tradition and judicial retention, with interactions between judicial retention and citizen ideology and religious tradition, on the vote of justices to reverse or uphold the death penalty. First, examining the control variables reveals that liberal justices, male justices and public defenders significantly influence the likelihood of a vote to reverse the death penalty (consistent with before). Second, a surprising finding is that the interaction between citizen ideology and judicial retention is significant and negative. This indicates that in states where justices are elected to office the ideology of the state (from conservative to liberal) results in a greater likelihood to uphold the death penalty (a conservative decision). Third, the interaction between judicial retention and the percentage of evangelical Protestants in the states is also significant and negative, indicating that in states where justices are elected, as the percentage of evangelicals increases, justices are more likely to vote to uphold the death penalty, the conservative decision.

While the statistical analysis thus far reveals some support for my hypotheses that the percentage of religious groups residing in states affect judicial decision making on death penalty cases and that this relationship exists in particular in states where justices are elected to office, it is important to understand the magnitude of the effects. Accordingly, table 4.3 presents the marginal effects of the significant variables from table 2 on judicial decision making in death penalty cases.

Focusing on the 3rd column, the results reveal that the marginal effects of all of the significant variables are small. This indicates that the variables do not produce large substantive effects on judicial decision making. Looking at the control variables, the presence of a public defender in the case increases the probability of a judge reversal by 2

percent and male, as compared to female, justices increase the probability of a reversal by 3.7 percent, holding everything else constant, from the baseline probability of 23 percent probability of a judge vote of reversal on the death penalty. A one unit increase in judge ideology, from conservative to liberal, results in a .002 increase in the probability of a judge reversal, or in other words, a ten percent increase in judge ideology (the scale of 0-100) results in a 2 percent increase in the probability of a judge reversal.

Examining the marginal effects of the judicial retention, citizen ideology and the religion variables reveals that the interaction of judicial retention with citizen ideology has the largest substantive impact on votes to reverse the death penalty. For states where justices are elected to office, a one unit increase in the citizen ideology of the state (from conservative to liberal) decreases the probability of a death penalty reversal by .018 percent. Put another way, a ten percent increase in citizen ideology (the scale of 0-100) results in a decrease in the probability of a death penalty reversal by 18 percent. The interaction of the percentage of evangelical Protestants residing in the state with judicial retention produces a much smaller substantive effect. Here, in states where justices are elected to office, a one unit increase in the percentage of evangelicals in the state decreases the probability of a judge reversal by .003 percent. Accordingly, a ten percent increase in the percentage of the evangelical population results in a decrease in the probability of a judge vote to reverse the death penalty by 3 percent.

Overall, the results confirm existing literature as well as provide support for my hypotheses regarding the effect of religion on judicial decision making in death penalty cases. The results confirm the literature that case characteristics matter, in particular the presence of a public defender results in more death penalty reversals (Brace and Boyea

2008) as well as the literature that specific judge characteristics matter, that justices are influenced by their own ideology when voting (Segal and Spaeth 2002) and their gender (Songer and Crews-Meyer 2000). The results also confirm the literature which argues that elected justices respond to public opinion when deciding cases, but they show the opposite effect, that as states become more liberal justices are more likely to vote to uphold the death penalty rather than reverse it. This finding could be due to the rough measure of citizen ideology that is used compared to previous scholars who are able to capture actual citizen support for the death penalty (Brace and Boyea 2008). Finally, my hypotheses are supported that justices and elected justices in particular do respond to religious opinion, primarily to evangelical Protestants. As the percentage of evangelical Protestants increases in states where justices are elected to office, the justices are more likely to vote in a conservative direction on the death penalty.

4.7 Discussion, Conclusion and Limitations

Overall, these results reveal that justices do listen to religious opinion when deciding death penalty cases, thus providing some support for my hypotheses. Without controlling for judicial elections, the results show that justices are more likely to vote in a conservative direction on the death penalty as the percentage of evangelical and mainline Protestants increases (compared to black Protestants). When controlling for the mechanism of judicial elections, the results reveal that only the presence of evangelical Protestants serves to influence the decision of justices who are elected to office to vote in a conservative direction on the death penalty. Given these results it is reasonable to argue that state Supreme Court justices may indeed respond to specific group opinion in the states as well as to overall public opinion, especially justices that are elected to office.

Religious groups in particular present justices with coherent and cohesive opinions on issues, providing them with an easy segment of the public to satisfy with the direction of their rulings.

While the issue of the death penalty presents one example of justices responding to religious opinion, the assumption is that justices are likely to respond to religious opinion on other issues on which religious opinion is united, social and moral issues in particular such as abortion and gay rights. These privacy issues should present an even easier test of the theory that religious groups are likely to influence state judicial decision making.¹⁸

However, there may be broader implications of these findings. The rise of the Religious Right has made religion increasingly important to politics (see Wilcox and Larson 2006; Wilcox and Robinson 2011). As a result, it may be that judges feel the increased political presence and power of the religious groups in their states and respond to their opinion not just in the domain of social and moral issues but in other areas of the law as well. One extension of this project would be to look at judicial decision making over time, to see whether justices increasingly responded to religious opinion with the rise of the Religious Right and the perceived importance of religious groups to politics.

Overall, the results of this paper that justices respond to religious opinion and findings of previous research that justices respond to public opinion have two opposing normative implications. On the one hand, these findings indicate the justices do listen to the will of the people, an inherently democratic value. They indicate that the judicial branch is not as removed and insulated from the people as it often appears to be. On the

¹⁸ In future research, I plan on testing my theory on a number of privacy issues.

other hand, these findings may reflect the fact that judges are not impartial when it comes to the law and that they are influenced not just by internal forces, their own political ideology, but also by external forces. By extension, it may be that it is simply impossible for judges to fulfill their role as impartial arbiters of the law. Accordingly, it is essential to identify other potential external forces that may influence judicial decision making as well as probe the normative implications of the increasing number of empirical findings which argue that judicial decision making is not a cut and dry process but is influenced by many different factors.

Judicial Retention Mechanism	Number of Death Penalty	Death Penalty Percent of Death Penalty	
	Cases	Cases	
Appointed (Governor/state Legislature)	411	5.44	
Judicial Retention Election	4292	51.53	
Judicial Election (Partisan/Non-Partisan)	3584	43.03	
Total	8329	100.00	
Religious Groups	Mean Percentage in	Range of Percentage in	
	States	States	
Evangelical Protestant	26.49	6.70 - 47.00	
Mainline Protestant	21.46	6.20 - 33.8.0	
Catholic	22.05	4.70 - 51.00	
Black Protestant	8.71	0.10 - 30.60	
Other Faith	21.29	9.80 - 74.90	
Total	100.00		

 Table 4.1. Descriptive Statistics, Method of Judicial Appointment and Death Penalty

 Cases and Mean Percentage of Religious Groups in the States

		Religion and	Religion and
		Judicial	Judicial Retention
	Religion Only	Retention	Interactions
Variable	Coefficient (SE)	Coefficient (SE)	Coefficient (SE)
Citizen Ideology	-0.004 (0.004)	-0.003 (0.004)	0.100 (0.020)*
Evangelical Protestant	-0.018 (0.007)*	-0.020 (0.007)*	0.028 (0.019)
Mainline Protestant	-0.059 (0.010)*	-0.061 (0.011)*	-0.120 (0.033)*
Catholic	0.003 (0.007)	0.001 (0.007)	0.010 (0.007)
Other Faith	-0.154 (0.025)*	-0.149 (0.025)*	-0.137 (0.024)*
Judge Ideology	0.010 (0.001)*	0.010 (0.001)*	0.009 (0.001)*
Gender	0.234 (0.080)*	0.230 (0.080)*	0.211 (0.081)*
Judge Age	0.004 (0.004)	0.004 (0.003)	0.012 (0.315)
Public Defender	0.181 (0.057)*	0.188 (0.058)*	0.126 (0.060)*
Judicial Retention		0.213 (0.133)	5.479 (0.960)*
Judicial Retention*Citizen			-0.102 (0.019)*
lacology			-0.020 (0.007)*
Judicial Retention*Evangelical			
Judicial Retention*Mainline			0.017 (0.010)
Constant	0.275 (0.474)	0.167 (0.478)	-4.423 (0.980)
Number of Observations	7988	7988	7988
Number of Groups	209	209	209
Wald χ^2	225.20*	227.31*	261.14*

Table 4.2. GEE Logistic Regression of the Effects of Religion on State Supreme Courts Death Penalty Decisions

Note: Dependent variable is individual judge decision on death penalty cases, 1= liberal decision 0= conservative decision. The excluded religious category is Black Protestant. Judicial Retention is coded as 1=elective method of retention 0=appointed. SE= Standard Error

*p<.01

	Religion Only	Religion and Judicial Retention	Religion and Judicial Retention
Variable	Marginal Effects	Marginal Effects	Marginal Effects
Citizen Ideology			0.018
Evangelical Protestant	-0.003	-0.004	
Mainline Protestant	-0.0005	-0.011	-0.022
Catholic			
Other Faith	-0.028	-0.027	-0.025
Judge Ideology	0.002	0.002	0.002
Gender	0.040	0.040	0.037
Judge Age			
Public Defender	0.033	0.034	0.023
Judicial Retention			0.288
Judicial Retention*Citizen			-0.018
lucology			-0.004
Judicial			
Retention*Evangelical			
Judicial Retention*Mainline			
Baseline Probability	0.236	0.236	0.236

Table 4.3. Marginal Effects for the GEE Logistic Regression of the Effects ofReligion on State Supreme CourtsDeath Penalty Decisions

Note: Marginal effects for the significant variables from the GEE logistic regression reported in table 4.2.

Chapter 5

Partisan Effects of Ballot Propositions

Early voting behavior literature found that voters have long time "standing decisions" that influence their vote choice long before an election cycle begins (Key 1949; Berelson, Lazerfeld and McPhee 1954; Campbell et al. 1960). The literature argued that campaigns had only minimal effects on vote choice. For example, Finkel (1993) found that for presidential elections in the 1980s, "the effect of the general election campaign in shifting the aggregate vote distribution was small, and the maximum overall net impact of the campaign was about 3%" (14). Scholars argued that campaign effects are too minimal to change electoral outcomes and as a result are not significant. However, recent scholarly work has begun to demonstrate that campaigns do matter, having a greater impact on vote choice than early literature found (Holbrook 1996; Shaw 1999).

A new area of research looking at campaign effects examines the effect of ballot propositions on vote choice. The dramatic increase of ballot measures has caused scholars to begin examining their impact on political behavior. For example, in the 2000 election alone, there were 204 statewide measures on the ballot, including 71 that were popular initiatives and referenda (Initiatives and Referendum Institute 2000). Scholars have found a number of significant effects of the use of direct democracy, including promotion of political efficacy among citizens (Bowler and Donovan 2002) increased levels of political knowledge (Smith 2001) and increased voter participation (Smith

2001; Tolbert, Grummel, and Smith 2001). Cumulatively, the literature suggests that direct democracy has meaningful impacts on political behavior and the political life of citizens.

Recently, scholars have sought to examine whether there is a link between voting on a ballot proposition and voting for a candidate. This question came of interest after scholars began finding that voting on ballot measures is highly informed by an individual's partisan identification (Smith and Tolbert 2001; Smith and Tolbert 2007). Literature investigating this link between initiatives and candidates is only just beginning to emerge, with findings suggesting a positive relationship does exist between voting on a ballot proposition and voting for a particular candidate (Nicholson 2005; Smith, DeSantis, and Kassel 2006; Donovan, Tolbert and Smith 2008). For example, Nicholson (2005) finds that the issue of Nuclear Freeze in the 1982 elections appearing on ballots across the states had an effect on national candidates. I seek to expand on this literature, looking at gay marriage ballot propositions in 2004 and their possible spillover effects to vote choice for presidential and lower level elective offices.

5.1 Agenda Setting Theory of Ballot Propositions

Nicholson (2005) offers the most well developed theory on how and why ballot propositions have spillover effects on candidate evaluations. He argues that ballot propositions have agenda setting effects, whereby they shape the political agenda, making certain issues more salient in an election than otherwise would be the case. In doing so, the ballot propositions then serve to "prime voters to evaluate candidates" based on that issue (15). In this theory, ballot propositions serve as another means by which

issues are brought before the public, apart from the effects of the campaigns themselves or the media.

In order for the link between ballot propositions and candidate evaluations to work, however, voters must ascribe positions on the issues to the candidates themselves. As Nicholson (2005) notes, there are partisan stereotypes, whereby voters associate positions on issues to either the Republican or the Democratic Party. Rahn (1993) finds these stereotypes to be so powerful that even if a candidate holds a position different from that of the party, citizens still associate the candidate with the position of the party because of the use of stereotypes. Accordingly, scholars have found that there are certain issues that parties "own", an example being the Republicans and moral issues such as abortion. Petrocik (1996) deems this "issue ownership", finding that if the important issues in a given election are owned by a particular party, that party will be advantaged, even if the candidates themselves do not discuss the issues or take opposing sides on the issues.

From this theory, ballot propositions can have a powerful effect on candidate evaluations, independent of what candidates themselves do or say. As a result, Nicholson (2005) finds evidence that candidates and parties actively seek to use ballot propositions to their advantage. They do so by using ballot propositions as a wedge issue, which is an issue that can "divide supporters of the opposing candidate, either persuading them to switch or to just sit out the election" (Baer 1995, 58). Nicholson found that the Republican Party successfully used anti-affirmative action initiatives in California as a wedge issue (Nicholson 2005 Chapter 6; see also Bowler, Nicholson and Segura 2006).

A key point of this theory linking ballot propositions to candidate evaluations is that it not only affects high profile elections, such as the Presidency, but it can affect all elections in a given electoral cycle. Ballot propositions affect views of the party, as demonstrated, thereby priming evaluations of all the individual candidates associated with that party. Therefore, they have the potential to be far reaching, affecting multiple campaigns and candidates. For example, Nicholson found in looking at nuclear freeze ballot initiatives, that voters in states with nuclear freeze initiatives were "likely to consider the freeze issue when making voting decisions for a given office or across offices, which voters in states without them were not likely to do so" (2005: 87). However, this aspect of the spillover effect of ballot propositions to candidate evaluations across multiple elective offices has received limited scholarly attention (though see Nicholson 2005; Ensley and Bucy 2010).

Donovan, Tolbert and Smith (2008) offer an important clarification to the agenda setting theory of ballot initiatives. They argue that the effect of ballot propositions is primarily a priming effect, whereby citizens are persuaded to think of a particular issue as highly salient and important in their evaluation of candidates. Theories of cognitive behavior treat citizen recall of memory as a largely passive process. They argue that when answering questions or making decisions about vote choice for example, citizens use "considerations that are immediately salient" to them (Zaller 1992, 49; Bartels 2003). In other words, citizens often make decisions based on what is at the top of their head, what has recently been stored in their memory and on their mind. Accordingly, ballot propositions serve to bring particular issues to the attention of citizens. Therefore, when evaluating candidates citizens are primed by ballot propositions to consider particular

issues important to their vote choice, and so vote for the candidate who is more in line with their opinion on that issue.

A final consideration towards a theory of the agenda setting nature of ballot propositions is the possibility that these propositions can prime subsets of individuals to vote for specific candidates. Ballot propositions have the possibility of priming individuals who feel intensely about the issue, rather than priming the entire electorate in the state. This can especially be the case when candidates and campaigns target specific groups with the propositions. Campbell and Monson (2008) test for this possibility with the gay marriage ballot propositions in 2004. They argue that no previous research had actually tested for the Republican campaign strategists plan in 2004, which was to "shore up Bush's support among evangelical Protestants" (402). Part of this strategy was to use the ballot propositions to their advantage among that particular group of voters.

Clearly, in theory, ballot propositions can have substantial and far reaching effects on election outcomes. They can serve as a mechanism by which individuals evaluate and vote for specific candidates. In the next section I examine the extant literature on the gay marriage propositions in the 2004 election which I use as a test for the theory of the agenda setting nature of ballot propositions.

5.2 Gay Marriage and the 2004 Election

In the last few decades, the issue of gay rights has been a common feature in state and local politics, and has also increasingly become a prominent feature in national politics (Haider-Markel and Meier 1996; Haider-Markel, Querze and Lindaman 2007). One aspect of gay rights in particular that has received a great deal of attention in the last few years is gay marriage. In 1996 the United States Congress passed the Defense of

Marriage Act, which defined marriage for the purposes of federal law, as a union between one man and one woman. However, beginning in 2003, a number of states took action to redefine marriage for purposes of individual state law. Massachusetts became the first state to allow gay marriage when the state Supreme Court ruled that a proposed civil union bill was unconstitutional as it denied equal access to marriage for same sex couples. Additionally, in 2003 San Francisco, California and Portland, Oregon began allowing same sex marriage, though their efforts were blocked by their respective courts. This led Presidential Bush in his State of the Union address on January 20, 2004 to mention the prospect of a federal constitutional amendment banning gay marriage. As a consequence of these events, numerous states added constitutional amendments to their November ballots, such that in the 2004 election thirteen states had gay marriage amendments, eleven of which were on the November ballot.¹⁹

Donovan, Smith and Tolbert (2008) note that the 2004 gay marriage amendments received significant media coverage. Tracking the media coverage on gay marriage through Google Trends, they find that, "nationally, media attention to gay marriage peaked in late February", declined through the summer, and "peaked again in late October" (Donovan, Smith and Tolbert 2008, 1220). Additionally, there was a clear distinction between the political parties and between the candidates on this issue. The Republican Party favored a constitutional ban on gay marriage, with Bush having suggested a constitutional amendment to define marriage between a man and a woman (2004 Republican Platform). The Democratic Party in contrast, was against a

¹⁹ The amendment states were Arkansas, Georgia, Kentucky, Louisiana, Michigan, Mississippi, Missouri, Montana, North Dakota, Ohio, Oklahoma and Oregon. Louisiana and Missouri were the two states that did not place the ballot measure on the November ballot, but instead included it on the ballot in the primary election.

constitutional amendment and in favor of letting the states decide the issue (2004 Democratic Platform). Senator John Kerry also favored letting the states decide but did argue for the legality of civil unions.

This was an issue that received a great deal of attention and an issue that the parties were divergent on, and so had the potential to set the electoral agenda and prime voters when going to the polls. This possibility was noticed by political pundits both before and after the election. Commentators from the *New York Times* to the *Wall Street Journal* speculated that Bush won reelection based on the presence of these issues on the ballot, especially in the swing state of Ohio (see, for example, Dao 2004).

As a result of the prominence and speculation of the importance of this issue to the 2004 election outcome, the issue of gay marriage provides a prime test for the agenda setting theory of ballot propositions. Scholars have investigated the effect of the gay marriage amendments on the 2004 election at both the individual and the aggregate level, with mixed results. At the individual level, some scholars conclude that "moral value" issues, including gay marriage, were not a central factor in the reelection of President Bush (Hillygus and Shields 2005; Lewis 2005). Instead, they found that the war on terror and the war in Iraq were more salient issues bringing individuals to the polls (Hillygus and Shields 2005). However, other scholars have reached opposite conclusions. They find that the presence of gay marriage amendments primed voters to support Bush (Donovan, Tolbert and Smith 2008) and it also affected some voters more than others to support Bush, most notably evangelical Protestants (Campbell and Monson 2008). Studies looking at this issue at the aggregate level are also mixed. Using aggregate state wide and national data, Abramowitz (2004) and Burden (2004) found that states with the

gay marriage amendments did not have increased levels of support for Bush or increased turnout compared to states without the amendments. Conversely, using county data for Ohio and Michigan, Smith, DeSantis and Kassel (2006) found that those who supported the gay marriage amendments were more likely to vote for Bush, confirming individual level findings.

I improve on this literature in a number of ways. A problem of the above mentioned aggregate level studies is their failure to look at the effect of the amendments on religious populations. As Campbell and Monson (2008) put it, they have not tested what the Republican campaign strategists said they were planning on doing, which was "shoring up support among evangelical Protestants" (402). Therefore, I will reexamine aggregate level data including measures of religion in the states. I will focus my analysis on the three Christian religions, evangelical Protestantism, mainline Protestantism and Catholicism, as they are the subgroup of voters most likely to be mobilized by the presence of gay marriage amendments (Campbell and Monson 2008).

Aside from the work of Ensley and Bucy (2010) scholars have not examined whether there were spillover ffects of the gay marriage amendments to lower level elective offices. The research thus far on the gay marriage amendments has focused on presidential vote choice. However, as Nicholson (2005) found, the agenda setting power of ballot propositions goes beyond a single elective office or campaign. Accordingly, Ensley and Bucy (2010) are the first to examine whether the presence of gay marriage amendments affect elective offices other than the Presidency. They focus on gubernatorial elections, finding that Republican governors benefited from the presence of the amendment when their position on the issue diverged from that of their challenger. I

contribute to this literature by examining the spillover effects of the gay marriage amendments on state legislative elections, examining whether Republican legislators benefited from the presence of these amendments.

5.3 Hypotheses

A consistent finding in the literature is that the presence of ballot initiatives can lead to higher levels of turnout (M. Smith 2001; Tolbert, Grummel, and Smith 2001). Individuals who would otherwise fail to vote are mobilized by the presence of these issues on the ballot. Scholars have found that states that allow for the use of ballot initiatives, especially the western states, report consistently higher turnout than states that do not allow their use (see Tolbert, Grummel, and Smith 2001).

Hypothesis 1: States with gay marriage amendments will report higher levels of turnout than states without the amendments.

As noted above, scholars have just begun investigating the partisan effects of ballot propositions. Ballot propositions can serve as a means of setting the electoral agenda, by priming voters to focus on specific issues in the campaign. When positions on these issues are tied to parties and candidates, voters can use that information in determining their vote choice. In the present analysis, the Republican Party and President Bush in particular were clearly tied to support for the gay marriage amendments.

Hypothesis 2: States with gay marriage amendments will have increased levels of turnout for the Republican presidential candidate, President Bush, as compared to states without the amendments.

One of the conditions under which ballot propositions can have partial effects is by mobilizing subsets of voters. Ballot propositions have the potential of mobilizing

individuals who feel intensely about the issue rather than mobilizing the entire statewide electorate. Scholars have found that there are issue publics within the American electorate. For a variety of reasons citizens may feel strongly about specific issues (Krosnick 1990). Ballot propositions are, by definition, issue focused. As a result, the issue public that feels intensely about the proposition, either for or against, may be the voters who are mobilized to vote, both for the proposition and for the candidate who supports their position on the issue. In the present study, evangelical Protestants, and identifiers of the three Christian religions more generally, felt strongly about the issue of gay marriage.²⁰

Hypothesis 3: There should be a spillover effect of gay marriage amendment states interacted with the three Christian religions to support the presidential Republican candidate, President Bush: specifically, the percentage of the three Christian religions in gay marriage amendment states, evangelical Protestantism in particular, should report increased levels of support for the Republican candidate, as compared to states without the amendments.

The partisan effects of ballot propositions should not stop at the highest elective office on the ballot. Voters associate issue positions on the propositions with a political party, not just with a specific candidate (Rahn 1993). As a result, spillover effects of

²⁰ There is also the possibility that subsets of voters who feel strongly against ballot propositions will be mobilized to turn out to vote. In the present analysis, gay rights activists and possibly liberals could be mobilized by the gay marriage amendments to vote for the Democratic candidate, John Kerry. Another possibility is that subsets of voters are demobilized by the presence of ballot propositions. In the case of the gay marriage amendments, Campbell and Monson (2008) find that secular Americans (those who do not identify with a religion) were demobilized both in terms of voter turnout and in vote choice for the Democratic candidate, John Kerry. However, though these are interesting propositions, I do not test for these possibilities in this analysis, holding them for future research.

ballot propositions to lower level offices should be seen, as voters are primed by the proposition to support a particular political party. This was clearly the case with gay marriage amendments, as the Republican Party took a clear position on the issue, so that all Republican candidates should benefit from the presence of the proposition. Two potential mechanisms could work to benefit lower level Republican candidates. First, the ballot propositions may mobilize voters who would otherwise have failed to vote, but who would vote for the Republican Party anyway. Second, the ballot propositions may cause voters to switch parties, giving them a greater likelihood of voting for Republican candidates.

Hypothesis 4: The effect of the gay marriage amendment should carry over to lower level elective offices; specifically, there should be seat gain for the Republican Party in state legislatures in states with gay marriage amendment initiatives.

As with Hypothesis 3, ballot propositions have the potential of mobilizing subsets of voters. Specifically, the gay marriage amendments may mobilize identifiers of the three Christian religions, evangelical Protestants in particular, to vote for the Republican presidential candidate. They may be mobilized to vote for other Republican candidates on the ballot, indicating the potential of spillover effects of the propositions to lower level offices.

Hypothesis 5: There should be a spillover effect of gay marriage amendment states interacted with the three Christian religions to lower level elective offices: specifically, the percentage of the three Christian religions in gay marriage amendment states, evangelical Protestantism in particular, should report increased

levels of support for the Republican Party in state legislatures, as compared to states without the amendments.

5.4 Data and Methods

As already noted, this study is an aggregate level analysis of the effect of gay marriage amendments on the 2004 election. As such, the unit of analysis is the individual state, and therefore I will be examining whether there were statewide effects of the gay marriage amendments. This is potentially a stricter test for the theory than individual level analysis, as the magnitude of the effect of gay marriage amendments must be stronger to have a statewide effect rather than having an effect on individuals. It is also theoretically important to look at state effects of ballot propositions. In the Electoral College, only states have an effect on the outcomes of elections. Therefore, if individual level effects of ballot propositions do not carry over to have statewide implications, it indicates that ballot propositions have only marginal effects in the political system. *Dependent Variables*

I test the effects of gay marriage amendments on three dependent variables. The first is a measure of overall state turnout in the 2004 election. This was calculated using total number of votes cast by state, dividing the number of votes cast in each state by the total population of each state. The second dependent variable is a measure of turnout for President Bush. This was also calculated using total number of state votes, dividing the number of votes cast for President Bush in each state by the total population of each state. The second dependent variable is a measure of turnout for President Bush. This was also calculated using total number of state votes, dividing the number of votes cast for President Bush in each state by the total population of each state.²¹ The third and final dependent variable is a measure of the change in the number

²¹ State population data comes from the US census population estimate for 2004. Turnout data for 2004 comes from David Leip's website <u>http://www.uselectionatlas.org</u>.
of Republican seats held in the lower house of the state legislatures from the previous election. I focus my analysis on the lower house in state legislatures because every member is up for reelection, rather than only examining part of the legislative body which is up for reelection in 2004. This was calculated by measuring the difference between the number of seats held by Republicans in the 2004 election from the previous election in each state, such that positive numbers indicate seat gain for Republicans and negative numbers indicate seat gain by Democrats.²² Nebraska was excluded from the analysis due to its unicameral nonpartisan legislature and Virginia was also excluded from the analysis because it holds off year state legislative elections.

Independent Variables

There are two main independent variables of interest in the analysis, whether the state had a gay marriage amendment in the 2004 election and the size of the religious population in each state. There were eleven states that had gay marriage amendments on their ballot in the 2004 November election: Arkansas, Georgia, Kentucky, Michigan, Mississippi, Montana, North Dakota, Ohio, Oklahoma, Oregon and Utah. The variable is a dichotomous variable, coded as one if a gay marriage amendment was present in the state and zero otherwise.

The second main independent variable of interest is a measure of the religious affiliations of citizens in the fifty states. Recent scholarship in religion and politics has established the necessity of using a more nuanced measure of religious affiliation than

His turnout data is collected directly from each state's election board or Secretary of State and so is an accurate measure of turnout figures in each state.

²² The data for Republican seat change comes from the State Dataset compiled by Stefanie A. Lindquist. It is located on the State Politics and Policy Quarterly website, which can be accessed at <u>http://academic.udayton.edu/SPPQ-TPR/tpr_data_sets.html</u>.

was previously used in the literature (Fastnow, Grant and Rudolph 1999; Layman 2001; Steensland et al. 2000). Specifically, Steensland and his colleagues (2000) created a religious measure with seven categories: evangelical Protestantism, mainline Protestantism, Catholicism, Judaism, Mormonism, African American Protestantism and other faith. In the current analysis, I employ data recently compiled by Green (2007) measuring the statewide population of each of the religious traditions. He compiled this data from numerous statewide surveys, and it is currently the most precise estimate of religious affiliations in the fifty states. It is measured as a percentage of the state population identifying with each religious affiliation. For the current analysis I focus on the three Christian religious categories, evangelical Protestantism, mainline Protestantism and Catholicism.²³

To test the hypothesis that gay marriage mobilized religious individuals and evangelical Protestants in particular to vote, an interaction term will be used between the percentage of each religious affiliation in the states and gay marriage amendment states. This will examine the subgroup that both live in a gay marriage amendment state and are affiliated with evangelical Protestantism, mainline Protestantism and Catholicism respectively. The main interaction of interest is with evangelical Protestants, but all three religious groups are examined.

²³ African American Protestants are an interesting case. They are socially conservative yet identify with the Democratic Party for historical reasons (Civil Rights for example) (see Steensland et al. 2000 for a full discussion of African American Protestants). Since my focus is on the Republican candidates, I chose to exclude them from the current analysis. However, in future work I will look at the effect of the gay marriage amendments on turnout for the Democratic candidate, John Kerry, to see if African American Protestants were mobilized by the propositions to vote for him.

In order to control for competing explanations for overall levels of turnout and turnout for President Bush in 2004, I follow previous aggregate level studies examining the effect of gay marriage amendments and include a control for the previous election turnout and vote share for President Bush (Abramowitz 2004; Burden 2004). Including a control for turnout and turnout for President Bush in 2000 allows for state demographic controls on turnout. The two variables are calculated in the same manner as described for turnout and turnout for Bush in 2004. In examining Republican state lower house legislative seat gain, I include a control for legislative composition in the previous election, measured as the number of Republicans in the lower house.²⁴ I also control for citizen ideology and party competition, which would affect Republican seat gain in the state legislature. More competitive states have a greater number of seats, which could potentially change party control compared to less competitive states. This could be a potential cause of Republican seat change, as states pick up seats partly because the state is more competitive, with more of a chance of winning seats. Party competition is measured using the Ranney index of party competition. Ideology is also a potential confounding variable with Republican seat change in that more conservative ideologies may be the causal factor behind gains in Republican seats. This is measured using the Berry et al. (1998) measure of citizen ideology, which is a score of ideology ranging from 0 to a 100 with higher scores indicating increased liberalism.²⁵

²⁴ Turnout data for 2000 again comes from David Leip's website <u>http://www.uselectionatlas.org</u>. Legislative composition in the lower house again comes from State Dataset compiled by Stefanie A. Lindquist. It is located on the State Politics and Policy Quarterly website, which can be accessed at <u>http://academic.udayton.edu/SPPQ-TPR/tpr_data_sets.html</u>.

²⁵ Both the Ranney index and the citizen ideology measure come from the State Dataset compiled by Stefanie A. Lindquist. It is located on the State Politics and Policy

The analysis is estimated using Ordinary Least Square regression. OLS is used because the three dependent variables are continuous variables, indicating that OLS is the most appropriate method. I also check for potential violations of OLS, specifically examining the data for potential problems with non-constant error (heteroskedasticity) and normality.

5.5 Results

I begin by examining the overall effect of gay marriage amendments on turnout and turnout for President Bush controlling for the previous election year results. Table 5.1 presents the results. In both cases, the gay marriage amendment variable is not statistically significant, indicating that the presence of these ballot measures did not increase overall levels of turnout or increase levels of support for President Bush in the states which had amendments versus the states that did not. ²⁶

However, as noted, the literature at the individual level has found that the presence of the gay marriage amendments mobilized a subgroup of voters, evangelical Protestants, to turn out to vote and to cast their ballot for President Bush (Campbell and Monson 2008). I begin testing this hypothesis looking at overall levels of turnout in the 2004 election.

Table 5.2 presents two models of turnout in the 2004 election, focusing specifically on the effect of religion. The first model is a base model examining the effects of the three Christian religions on overall levels of turnout in the 2004 election.

Quarterly website, which can be accessed at <u>http://academic.udayton.edu/SPPQ-TPR/tpr_data_sets.html</u>.

²⁶ There are only fifty observations, leading to a potential problem with normality. However, a quorm test for normality revealed that the data are normally distributed. A test for heteroskediscity reveals that the errors are uncorrelated in all of the models.

The results reveal that states with higher percentages of mainline Protestants exhibit significantly higher rates of turnout in the 2004 election controlling for turnout in 2000, while states with higher percentages of evangelical Protestants and Catholics do not. Model 2 looks at the interaction of the percentage of evangelicals, mainline Protestants and Catholics residing in the states with gay marriage amendment states. It examines whether states with higher percentages of the three religions and with gay marriage amendments on the ballot have higher levels of turnout in the 2004 election. However, the results largely disprove my hypotheses. States with higher percentages of evangelical Protestants with a gay marriage amendment on the ballot do not have higher levels of turnout. The only significant finding is that states with higher percentages of mainline Protestants still exhibit higher levels of turnout in the 2004 election.

Table 5.3 presents the same two models looking at the effect of religion on turnout for President Bush in the 2004 election. The base model, model 1, examining the effects of the three Christian religions on turnout for President Bush in the states, reveals that controlling for levels of support for Bush in 2000 none of the three Christian religions is statistically significant. Model 2, examining whether higher percentages of all three Christian religions interacted with gay marriage amendment states results in increased turnout of President Bush, shows that here as well there is no statistically significant effect of gay marriage amendment states and religion.

Table 5.4 presents the final set of results looking at whether the effect of gay marriage amendments carries over to other Republican candidates, in this case specifically looking at the lower house of the state legislatures. The first column shows results just including whether gay marriage amendment states is a positive contributor to

Republican legislative lower house seat gain. The Ranney index of party competition is positive with a large effect as anticipated. Citizen ideology, while not significant at the traditional .05 level (.064), is also in the expected direction having a negative coefficient indicating that a one unit increase in citizen ideology, going from conservative to liberal, results in roughly a one seat lose for the Republican Party. Republican control in the previous election is also significant. However, here again the presence of gay marriage amendment states has no effect on Republican legislative lower house seat gain.

I next look at the interactive effect of religion and gay marriage amendments on Republican seat gain, examining the same three models as before. Model 1 examines whether the three Christian religions have a significant effect on Republican seat gain. The results reveal that they do not. However, model 2, including all three Christian religions interacted with gay marriage amendment states, reveals some interesting results. The percentage of mainline Protestants residing in the states interacted with gay marriage amendment states is significant, indicating that as the percentage of mainline Protestants residing in amendment states increases by one unit, Republicans gain slightly over one seat. While not significant at the .05 level, the percentage of Catholics residing in the states interacted with gay marriage amendment states (.066) exhibit the reverse effect of mainline Protestants, causing a decline of roughly one Republican seat. This effect of Catholics is possibly the result of Catholics being mobilized to vote in legislative races by the gay marriage amendment yet voting for the Democratic candidate, as they have historically identified with the Democratic Party.

5.6 Discussion and Conclusion

Overall, the findings largely fail to confirm my hypotheses, as I find that the gay marriage amendments did not increase levels of turnout in the election, they did not increase support for President Bush, they did not affect Republican seat gain in the lower house of state legislatures and the amendments did not interact with the percentage of evangelical Protestants residing in the states to effect any of the three dependent variables.

However, the findings do produce two notable and highly interesting results. First, the percentage of mainline Protestants residing in the states is significantly related to higher turnout. This confirms the religion and politics literature that finds that mainline Protestants have traditionally been the most civically engaged of the three Christian religions (Wuthnow 1999). Second, I did find that higher percentages of mainline Protestants residing in gay marriage amendment states does increase Republican seat gain in the lower house of the state legislatures, while the opposite effect was found for Catholics, and no effect was found for evangelical Protestants. As I speculated earlier, the negative relationship for the percentage of Catholics residing in the states may be a product of the fact that they have historically identified with the Democratic Party. Therefore, they are mobilized by the gay marriage amendments yet vote for the Democratic candidate.

The difference between mainline and evangelical Protestants presents a different story. The fact that the percentage of evangelical Protestants residing in the states did not affect any of the dependent variables indicates that they were already mobilized to turnout and, in turning out, to vote for the Republican candidates on the ballot. This

comports with recent religion and politics literature. Evangelicals have become an established voting bloc, with well over three quarters of evangelicals identifying with the Republican Party (Green 2007). They were mobilized to vote with the rise of the Religious Right in the 1980s and have become habitual voters, identifying with and voting for the Republican Party (Wilcox and Larson 2006; Gerber, Green and Shachar 2003; Smith and Walker 2013). Mainline Protestants, on the other hand, are more liberal than evangelicals, and as a result do not identify as strongly as a group with the Republican Party.²⁷ Therefore, they are more amenable to being swayed to vote for Republican candidates as a result of the gay marriage amendments. However, I argue that the fact that many of my hypotheses were rejected, that I found largely no effect for gay marriage amendments on the 2004 election at the state level, is puzzling for two reasons.

First, these results reveal a disconnect between the way the media and political analysts after the election viewed the importance of the gay marriage amendments to the outcome and empirical evidence on their actual effect. As noted, commentators from the *New York Times* to the *Wall Street Journal* declared that the gay marriage amendments had been a key force in the victory of President Bush in 2004, garnering him votes in swing states, most notably Ohio, which allowed him to win. For example, the *New York Times*, stated that the amendment in Ohio "helped turn out thousands of conservative voters on Election Day" and that this support by conservative voters was "widely viewed as having been crucial to President Bush's narrow victory in that swing state" (Dao 2004, A-28). In contrast to these claims by the media, the results here clearly show that the gay

²⁷ See Steensland et al. (2000) for a thorough description of the differences between mainline and evangelical Protestants.

marriage amendments did not change electoral outcomes and so were not the cause of President Bush's victory. This leaves an interesting question as to why the media was so persuaded that the gay marriage issue made the difference in the election. This disconnect also points to the importance of empirical investigation into political phenomenon, as it serves to clarify assumptions about political events.

The second reason that these results present a puzzle is the discrepancy between the individual and the aggregate level findings on the effect of the gay marriage amendments. At the individual level, scholars found that the gay marriage amendments did increase support for President Bush among white evangelical Protestants living in amendment states (Campbell and Monson 2008). However, the findings here suggest that this increased support did not carry over to have an aggregate level effect. While gay marriage amendments might have mobilized evangelical Protestants in support of President Bush, the effect was not strong enough to have electoral implications. This discrepancy clearly highlights the danger of the ecological fallacy and the need to conduct multiple levels of analysis on the same phenomenon. A finding at one level of analysis does not necessitate the existence of that finding at another level of analysis.

Finally, these findings point to the need to refine the agenda setting theory of ballot initiatives. These results, along with the scholarly work on this topic, indicate that there may be certain conditions under which ballot propositions are effective. The present findings seem to suggest that the effect of ballot propositions is not large enough to change electoral outcomes, but does have a substantial impact on individual voting decisions. Further refinement is needed to sort out the conditions under which this theory and the effects of ballot propositions hold.

Table 5.1. Ordinary Least Squares Regression Analysis, of Gay MarriageAmendment States on Turnout in 2004 and Bush Vote Share in 2004

	2004 Turnout	2004 Bush Vote Share	
Variable	Coefficient (SE)	Coefficient (SE)	
Turnout in 2000	1.024 (0.042)***		
Bush Vote Share 2004		1.014 (0.035)***	
Gay Marriage States	0.008 (0.005)	0.004 (0.003)	
	0.005 (0.017)*		
Constant	0.036 (0.017)*	0.033 (0.007)***	
Observations	50	50	
R^2	0.92	0.95	

Note: Dependent Variable is the percent turnout in 2004 and the percent vote share for President Bush in 2004 measured from 0 to 1. SE= Standard Error *p<.05 **p<.01 ***p<.001

	Model 1	Model 2
Variable	Coefficient (SE)	Coefficient (SE)
Turnout in 2000	0.942 (0.054)***	0.948 (0.055)***
Gay Marriage States		0.066 (0.055)
Evangelical Protestant	0.000 (0.000)	0.000 (0.000)
Mainline Protestant	0.01 (0.000)**	0.001 (0.001)*
Catholic	-0.000 (0.000)	0.000 (0.000)
Evangelical*Gay Marriage		-0.001 (0.001)
Mainline*Gay Marriage		0.000 (0.002)
Catholic*Gay Marriage		-0.002 (0.001)
Constant	0.040 (0.020)	0.032 (0.021)
Observations	50	50
R^2	0.94	0.94

Table 5.2. Ordinary Least Squares Regression Analysis, Effect of Gay MarriageAmendment States and Religion on Turnout in 2004

Note: Dependent Variable is the percent turnout in 2004 measured from 0 to 1. SE= Standard Error *p<.05 **p<.01 ***p<.001

	Model 1	Model 2	
Variable	Coefficient (SE)	Coefficient (SE)	
Turnout in 2000	0.958 (0.042)***	0.965 (0.045)***	
Gay Marriage States		-0.018 (0.037)	
Evangelical Protestant	0.000 (0.000)	0.000 (0.000)	
Mainline Protestant	0.000 (0.000)	0.000 (0.000)	
Catholic	-0.000 (0.000)	-0.000 (0.000)	
Evangelical*Gay Marriage		0.000 (0.001)	
Mainline*Gay Marriage		0.001 (0.001)	
Catholic*Gay Marriage		-0.000 (0.000)	
Constant	0.037 (0.011)**	0.037 (0.012)**	
Observations	50	50	
\mathbf{R}^2	0.96	0.96	

Table 5.3. Ordinary Least Squares Regression Analysis, of Gay Marriage Amendment States on Bush Vote Share in 2004 and Religion

Note: Dependent Variable is the percent vote share for President Bush in 2004 measured from 0 to 1. SE= Standard Error *p<.05 **p<.01 ***p<.001

	Gay Marriage				
	States	Model 1	Model 2		
Variable	Coefficient (SE)	Coefficient (SE)	Coefficient (SE)		
Republican Seats	0.097 (0.024)***	0.101 (0.025)***	0.101 (0.024)***		
Previous Election					
Citizen Ideology	-0.120 (0.058)	-0.159 (0.091)	-0.148 (0.090)		
Party Competition	14.966 (6.218)*	20.145 (8.241)	18.650 (8.564)*		
Gay Marriage States	-1.670 (3.970)		0.633 (20.173)		
Evangelical Protestant		-0.164 (0.150)	-0.390 (0.155)		
Mainline Protestant		0.133 (0.178)	-0.028 (0.192)		
Catholic		-0.043 (0.125)	-0.013 (0.126)		
Evangelical*Gay Marriage			-0.256 (0.396)		
Mainline*Gay Marriage			1.333 (0.573)*		
Catholic*Gay Marriage			-0.965 (0.510)		
Constant	-4.193 (3.970)	-2.158 (6.255)	-2.319 (6.066)		
Observations	48	48	48		
R^2	0.33	0.34	0.46		
Note: Dependent Variable is the change in Republican seats in the Lower House of the					

Table 5.4. Ordinary Least Squares Regression Analysis, of Gay MarriageAmendment States and Religion on Republican Seat Change in State LegislaturesLower House

Note: Dependent Variable is the change in Republican seats in the Lower House of the States Legislatures in 2004 from the previous election, measured as positive for Republican gain. SE= Standard Error *p<.05 **p<.01 ***p<.001

Chapter 6

Conclusion

"Unless mass views have some place in the shaping of policy, all the talk about democracy is nonsense."

- V.O. Key (1961)

Democracy should arguably rest on the opinions of the people. A government of the people, by the people and for the people would be a contradiction in terms if it did not listen to the voice of the people and take into account their opinions and desires in governing. This is perhaps especially the case in a representative democracy where citizens do not directly vote on policy but their views are translated through elected officials. The study of public opinion then, determining whether and under what conditions government listens to the voice of the people, presents an important area of research.

I argue that most of the literature examining the role of public opinion in American democracy tends to focus on the opinion of individual citizens and typically views these citizens as atomistic actors. Questions such as whether citizens have clear and structured preferences (Converse 1969), how informed is the citizenry and how informed should it be (Delli Carpini and Keeter 1996), and are there ways in which citizens can 'get by' with less information (Lupia 1994) have long dominated the attention of scholars. I add to the literature by bringing back a renewed focus on groups within American society and examine how they shape, form and affect public opinion. In doing so I build on the more recent work of scholars that have moved from examining public opinion from the starting point of citizens as atomistic actors to focusing on the role of interpersonal relationships in shaping and forming opinion (Huckfeldt 1979; 1983; 1984; Huckfeldt et al. 1995; Huckfeldt and Sprague 1995; Huckfeldt et al. 1998; Huckfeldt, Johnson and Sprague 2004; Walsh 2004; Mutz and Mondak 2006). In particular, I focus on the role of interpersonal relationships in public opinion as they are played out through groups within society. Through this I address the question, what are some of the specific conditions under which public opinion is listened to and translated into government responsiveness.

To answer this question I present a theory of group influences on public opinion. My theory has three parts. First, I argue that the importance of interpersonal relationships and social identity theory in particular work to ensure that citizens are group orientated and are influenced in their individual opinion through groups. Citizens have a strong desire to feel connected to others (social identity theory) which is alleviated through group interactions and these interactions in turn shape and mold individual opinion. Second, in a large democratic society it is virtually impossible for government to listen and be responsive to individual opinion. Instead, government listens to collective opinion which, I argue, is often mediated through dominant groups within society (groups that are large and cohesive in opinion). Finally, group influence on public opinion is conditional on the group showing intensity of preference and opinion and 'owning' particular issues. This allows individual group members to know and adopt the opinion of the group and

signals to government the group opinion such that government has a clear indication of opinion to follow.

6.1 Theory Testing: How Well did it Perform?

I test this theory in three specific cases, political parties, state Supreme Court decision making and ballot initiatives. Each analysis examines a sub-research question as the focus of attention. However, as I explained in the introduction, my overall theory of group influences on public opinion applies in each case. While I went over why and how the theory applies in the introduction, here I will examine how well my theory actually performed.

6.2 Political Parties

The sub-research question under consideration for my first analysis is whether strength of religious attachment leads to strength of partisan attachment. I argue that in the American two party system of government, one of the primary, if not the primary way, in which public opinion is listened to and translated into government decision making is through partisanship. Accordingly, the platform and opinions of the political parties take on a significant role in translating public opinion to government. Scholars argue that the overall opinion of the political parties tends to be formed through the elites in government (Aldrich 1995) and through those who are strongly attached to the party (Schattschneider 1960). Identifying the causes behind someone claiming to be a strong partisan, therefore, has significant public opinion consequences.

I argue in the analysis that there are spillover effects of social identity theory, such that individuals who are strongly attached to their religious identity are likely to be strongly attached to their political party. This is premised on the idea that the need to feel

connected to others in one area of life leads to a need to feel connected in other areas of life as well. However, it is likely that this relationship is conditional upon partisans' perceptions that their religious commitment has some meaning in relation to their political party's platform, ideas and leaders. Overall, I find in the analysis that religious attachment is positively related to strength of partisan attachment. As respondents become more attached to their religion they increase their likelihood of identifying as a strong partisan. However, I do find that this is conditional on the political party. When analyzing Republicans and Democrats separately, the effect only holds for Republican identifiers.

Accordingly, I argue that these results provide support for my overall theory of group influences on public opinion. First, the social identity effects are clear. Increased attachment to religion leads to increased attachment to a political party. Religion, as per social identity theory, is therefore informing the opinion and behavior of its individual members. This confirms the potential for group based patterns of opinion. However, a note here is that I am examining religious attachment as opposed to specific religious groups. I would expect to find this effect looking at individual religious groups, though there will likely be variation in the actual substantive effect.

Second, I argue that as a result of this confirmed link between strength of religious attachment and strength of partisanship, the implication is that religion is impacting the political parties and so the opinions that they hold. Again, it is arguably the strong partisans who most affect the agenda and opinions (e.g. stances on specific issues) of the overall political party, so the fact that more religiously attached individuals are more likely to be strong partisans indicates that they have more influence over the

agenda and opinions of the party. I argue that this confirms my theory of group based influence on public opinion, because religious groups are influencing the overall opinion of the political party (though note here again further tests are needed looking at specific religious groups to fully confirm this point).

At the same time, I find that this group based influence on public opinion through political parties is conditional on the group holding intensity of preferences, the third point of my theory. This is clearly born out because I only see affects of strength of religious attachment on strength of partisan attachment for Republicans and not for Democrats. As I argue, religious groups have formed very intense preferences about social issues in particular and the Republican Party is thought of as the party that is 'friendlier' toward religion. Accordingly, the intensity to which religious groups hold their opinion signals clear direction to their own members to attach to the Republican Party and it signals to the Republican Party the continued direction of opinion they should follow.

Overall, I find some support for my theory of group influences on public opinion in this analysis. However, it is not a complete test as clarifications are needed. As presented, I provide only an indirect test of group influences. For a direct test, I need to look specifically at religious groups rather than religion as a whole. Similarly, I argue from a theoretical standpoint (and from the literature) that it is the religious group influencing the opinions of the overall political party (the Republican Party in particular). However, this needs to be more directly tested through over time analysis to see whether the opinions of the party truly change as religious groups show intensity in their preferences and opinion on specific issues.

6.3 State Supreme Court Decision Making

For this analysis, the sub-research question under consideration is whether state Supreme Court justices respond not just to overall public opinion but to specific group opinion, religious groups in particular, in their decision making. Here I focus on a specific branch of government, the courts, to see how well they listen to public opinion and so answer my overall research question of conditions under which public opinion is listened to and translated into government policy. This is the branch of government theoretically least likely to be influenced by public opinion as evidenced in the fact that scholars did not spend much time until recently studying public opinion influences on the judicial branch. However, more and more research is showing that the national Supreme Court and state Supreme Court justices are indeed responsive to public opinion (Caldeira 1991; Mishler and Sheehan 1993; Flemming and Wood 1997; McGuire and Stimson 2004; Giles et al. 2008).

I argue that it is theoretically likely that justices respond not just to overall public opinion but to specific group opinion, religious opinion in particular, for three reasons. First, in many states, state Supreme Court justices are elected to office. They, along with other elites, 'fear losing office' (Mayhew 1974; Brace and Boyea 2008) and so will likely cater towards the groups within the state that will ensure their election or reelection. Religious groups are some of the largest groups in American society and hold cohesive opinion on some issues in particular so they may present a valuable election group for justices. Second, justices may listen to public opinion because they themselves are changing as a result of the changing norms and opinions of the state (Giles et al. 2008). For example, as a state becomes more conservative, justices may be themselves

influenced and vote in a more conservative fashion. I argue that changes in societal norms are theoretically likely to be caused by dominate groups in a state. Religious groups are again both large and cohesive in opinion and so may influence the overall ideology of a state and so the ideology of justices. Third, justices may respond to group opinion if a group holds very intense preferences (Dahl 1956). Justices may feel the need to respond to intense opinion to ensure legitimacy in the eyes of the group (see the importance of legitimacy for the courts, McGurie and Stimson 2004) as well as be protectors of minority rights. The fact that religious groups tend to hold very intense preferences makes them a group justices are likely to respond to. Accordingly, I expect to find, examining death penalty cases, that state Supreme Court justices both elected and nonelected are responsive to religious opinion looking at the four largest religious groups.

Overall, I find that justices do respond to religious opinion on the death penalty. Without examining the election mechanism, justices are more likely to uphold the death penalty, the conservative opinion, as the percentage of evangelical Protestants and mainline Protestants in a state increases (as compared to black Protestants). In states where justices are elected, I find that as the percentage of evangelical Protestants increases, justices are more likely to uphold the death penalty. However, the substantive effect of these results is small, indicating that while a change in opinion is produced, it is not very strong.

These results provide support for my overall theory of group influences on public opinion. While not directly addressed in the analysis, the foundation for my results rests on social identity theory. Citizens join groups to satisfy a need to feel connected to others and are consequently shaped in their opinion and behavior by the group. My

finding that state Supreme Court justices are responsive to religious groups on the issue of the death penalty is premised on the fact that religious groups hold clear and cohesive direction of opinion such that justices can respond to it, a direct implication of social identity theory. Further, my theory is confirmed that there are clear group influences on public opinion. It is important to not only examine the opinion of individuals or public opinion as a whole, but to examine group opinion. As I find in this analysis, it is the case that government listens to and responds to group opinion (i.e. State Supreme Court justices do not just listen to overall opinion but to religious opinion). Finally, for my theory of group influences on public opinion to hold, the condition of the group holding an intensity of opinion must be met. I find support for this in my analysis. My results primarily point to evangelical Protestants as influencing state Supreme Court justices in their decision making on the issue of the death penalty. The literature shows that evangelical Protestants are the most conservative on the issue of the death penalty (Britt 1998) (and therefore hold the most intense opinion) so it upholds my theory that I primarily find results for them. The other religious groups (mainline Protestants and Catholics) are more divided in their opinion on the death penalty (less intense) and so it is less likely that justices would respond to them, as I find to be the case.

Overall, I once again find support for my theory of group influences on public opinion. An increase in the presence of evangelical Protestants in states in particular results in state Supreme Court justices making the conservative decision to uphold the death penalty. Justices are indeed swayed by group opinion. However, it is important to broaden this analysis to other policy areas, social and moral issues and beyond, to more fully test how well the group influence theory of public opinion works.

6.4 Direct Democracy: Ballot Propositions

For my final analysis, the sub-research question under consideration is whether ballot propositions affect candidate choice. In other words, I examine whether there are spillover effects of ballot propositions such that voters are primed by the initiative in their choice to cast a vote for one candidate over another when they go to the polls. In a democracy, mechanisms such as ballot propositions are the most 'direct' that public opinion can be translated to government because it involves individuals actually voting on policy. As noted, scholars find that states with initiatives and referendums are more responsive to public opinion than those without (Arceneaux 2002; Burden 2005). For my analysis, I focus specifically on how public opinion is voiced through candidate choice, which can in turn be affected by the use of direct democracy. Elected officials must take into consideration the opinion of those who vote for fear of not getting elected or losing office later (Mayhew 1974).

I argue that ballot propositions have agenda setting effects whereby they shape the political agenda making certain issues more salient in an election than otherwise would have been the case. In doing so, ballot propositions serve to prime voters to appraise candidates based on their position on the issue and consequently vote for the candidate that holds their own opinion. This only works, however, if parties "own" particular issues (Petrocik 1996) such that voters can then ascribe the position on the issue to individual candidates regardless of whether that candidate actually adheres to the position. Additionally, candidates can use ballot propositions to their advantage by identifying them as a wedge issue (Baer 1995) that divides the supporters of the opposing candidate. In particular to my theory of group influences on public opinion, ballot

propositions have the potential to target and so prime subsets of voters and groups more specifically in who to vote for in the election. I examine the gay marriage amendments in the 2004 election, an issue that received a great deal of attention and had the potential to prime religious groups in voting for the Republican candidates in the election (President Bush on down the ballot). I examine aggregate level effects (the state being the unit of analysis) rather than individual opinion. For my theory of group influences on opinion, individual level opinion and opinion change is important (social identity) but I argue that it is only through the aggregate that politics is ultimately carried out so examining aggregate level effects is vitally important.

Overall, in my analysis I find no result for ballot propositions increasing overall levels of turnout, turnout for President Bush or turnout for other lower level Republican officials in the fifty states. Examining religious groups in particular, I only find a result for mainline Protestants, that they report a higher rate of voter participation in states that have gay marriage amendments and they increase Republican seat gain slightly. I find, in essence, almost no support for my theory of spillover effects of ballot propositions. It is particularly striking that I find no result for evangelical Protestants, the group most likely to have been primed by the gay marriage amendments to increase turnout for President Bush and other Republican officials. These results are at odds with individual level findings (Campbell and Monson 2008) and as such show the importance of examining both individual and aggregate level effects and the danger of the ecological fallacy.

This analysis largely fails to confirm my overall theory of group influence on opinion as displayed in the case of ballot propositions. First, I argue for social identity effects to be seen both in terms of groups being primed by the initiative to show up at the

polls (if the group as a whole values increased participation then individual members will conform to that behavior) and to effect who they choose to vote for (members will conform to the opinion of the group to vote for a specific candidate in this case religious groups and the Republican candidates). However, I find support for neither contention. Second, I argue that by conceptualizing public opinion in terms not of overall opinion but group opinion, it may be the case that opinion is accurately reflected through dominant groups. Ballot propositions may target specific groups such that these groups are more likely to express their opinion (i.e. vote and vote for a specific candidate) and so group opinion is the influence behind public opinion. However, again I find no effect for the gay marriage amendments priming religious groups in voter turnout or vote choice. Finally, I argue that my theory holds only when the condition of intensity of preference is met. The group must feel strongly in their opinion for it to influence individual members of the group and for it to signal to government the group's preferred stance. Gay marriage amendments were an issue that religious groups and evangelicals in particular felt very passionately about. So the third condition was met yet I still did not find support for my theory.

Overall, this analysis produces a number of interesting conclusions. First, it shows the discrepancy, as noted, between individual and aggregate level effects of ballot propositions. Individual level findings did show gay marriage ballot propositions affecting evangelicals in particular in their vote choice (Campbell and Monson 2008) yet I do not find that this holds at the aggregate level when examining overall state results. For my theory of group influences on opinion, I argue that it is at the aggregate level that we truly see opinion mattering to government. This is a case in point. Though individual

evangelical opinion might have been changed by these propositions, at the aggregate level this was not the case and so it had no effect on overall opinion and so overall election outcomes. Second, this analysis shows a limitation on my theory. While I found support for my theory in the case of political parties and state Supreme Court decision making, I do not find support in the case of ballot propositions. It would be a good idea to confirm this non-finding with ballot initiatives focusing on other issue areas and with other groups. Finally, this points to the importance of testing the breadth and depth of theory to see conditions under which it holds and does not hold.

6.5 Concluding Thoughts

America both historically and currently is a nation of 'joiners.' Americans show themselves willing to form and join a wide variety of groups within society and these groups often take on political meaning either intentionally or not (Tocqueville [1835-40] 1969; Truman 1951). Accordingly, groups and group dynamics play a large role in American politics, one I argue that is often overlooked. While scholars investigate the role of groups in areas such as voter turnout (Campbell et al. 1960; Lewis-Beck et al. 2008) and the development of social capital (Putnam 2000), the public opinion literature has long been lacking in a focus on the role groups can and do play in the study of public opinion (Huckfeldt 2007).

I seek to address this gap in the literature focusing on the question of finding specific conditions under which public opinion is listened to and translated into government responsiveness. As an answer to this question, I argue that it is through groups that government can and does listen to public opinion and so be responsive to opinion in terms of policy and decision making. Put another way, I argue that groups are

an important and often overlooked mediator of public opinion. More specifically, I offer a theory of group influences on public opinion, arguing that groups first shape individual opinion through social identity effects, government is theoretically likely to listen to groups rather than individual or overall opinion because politics is ultimately carried out in terms of the aggregate, but government is only likely to listen to group opinion if the group holds intense preferences and can therefore signal opinion to individual group members and to government.

In testing this theory, I find some support. I find that increases in religious attachment leads to increases in partisan attachment. In terms of public opinion and my overall research question, this means that religion (and by extension I argue religious groups though this still must be tested directly) is influencing the opinion of the political parties and thus of elected officials making government, I argue, responsive to group opinion. I also find that state Supreme Court justices listen and respond to religious groups (evangelical Protestants in particular) in their decision making on death penalty cases. Hence, I find that the branch of government least theoretically likely to listen to public opinion does in fact listen and respond to group opinion more specifically. These results show that groups are indeed an important mediator of public opinion and as such deserve attention in the public opinion literature.

However, I also find potential limitations of my theory of group influences on public opinion. In examining whether ballot initiatives have spillover effects on candidate choice in elections, I find at the aggregate level that gay marriage amendments had no effect on turnout or turnout for Republican officials (including President Bush) among religious groups. Religious groups were arguably the groups most likely to be

primed by these initiatives and so show group influence on opinion yet I did not find this to be the case. This finding can be read in two different ways. It shows either a limitation to my theory of group influence on opinion or it shows that direct democracy may not in fact 'add' anything to democracy in terms of public opinion. Citizens and religious groups in particular were already likely to voice their opinion with or without the mechanism of ballot initiatives to get them to the polls and prime them in who to vote for. Further tests are needed to confirm which story is the correct one.

Beyond demonstrating the continued significance of groups to American politics and to public opinion in particular, my results also highlight the importance of examining both individual and aggregate level data. As I argued in the introduction and theory, most studies of public opinion tend to focus on the individual. Yet, while the study of individual opinion and opinion formation is important, it is only in the aggregate that politics is ultimately carried out. In almost every case, government cannot listen and respond to individual opinion but must respond to the opinion of the 'collective'. Though the collective can be conceived of in different ways, I argue that one way to do so is through groups in American society. Consequently, I argue that it is important to look not just at individual public opinion but public opinion at the aggregate level. Accordingly, for two of my three analyses (state Supreme Court decision making and ballot initiatives) I focus on the effect of the aggregate. For the case of ballot initiatives in particular, I find discrepancies between individual level analyses (e.g. evangelicals did increase their turnout for President Bush in gay marriage amendment states) and my aggregate level findings (e.g. no effect for evangelicals increasing turnout for President Bush). This demonstrates, I argue, that individual opinion and aggregate or collective

opinion can and often do behave in two distinct ways. It highlights the danger of the ecological fallacy that scholars have long noted. It also highlights the need for more aggregate level opinion studies if it is indeed true that politics is ultimately carried out in terms of the aggregate.

Finally, the fact that my theory holds for two out of the three analyses shows that future research is needed to refine where and when group influences on public opinion applies. By examining three separate aspects of how public opinion can be listened to and translated into government decision making, I demonstrate the importance, and arguably the necessity, of scholars testing the breadth and depth of their theories in their research.

References

- Abramowitz, Alan. 1995. "It's Abortion, Stupid: Policy Voting in the 1992 Presidential Election." *Journal of Politics* 57: 176-186.
- Abramowitz, Alan. 2004. "Terrorism, Gay Marriage, and Incumbency: Explaining the Republican Victory in the 2004 Presidential Election." *The Forum* 2(4).
- Abramowitz, Alan, John McGlennon, and Ronald Rapoport. 1981. "A Note on Strategic Voting In a Primary Election." *The Journal of Politics* 43(3): 899-904.
- Abramowitz, Alan I., and Kyle L. Saunders. 2006. "Exploring the Bases of Partisanship in the American Electorate: Social Identity vs. Ideology." *Political Research Quarterly* 59(2): 175-87.
- Abramson, Paul R. 1976. "Generational Change and the Decline of Party Identification in America, 1952-1974." *American Political Science Review* 70: 469-78.
- Abramson, Paul R. 1979. "Developing Party Identification: A Further Examination of Life-cycle, Generational and Period Effects." *American Journal of Political Science* 23:78-96.
- Abramson, Paul R., and John H. Aldrich. 1982. "The Decline of Electoral Participation in America." *The American Political Science Review* 76(3): 502-521.
- Achen, Christopher H. 1975. "Mass Political Attitudes and the Survey Response." American Political Science Review 69(4): 1218-1231.

- Achen, Christopher H. 2005. "Let's Put Garbage-Can Regressions and Garbage-Can Probits Where They Belong." Conflict Management and Peace Science 22(Winter): 327-339.
- Aldrich, John H. 1983. "A Downsian Spatial Model with Party Activism." *American Political Science Review* 77(4): 974-990.
- Aldrich, John H. 1995. Why Parties?: The Origin and Transformation of Political Parties in America. Chicago: University of Chicago Press.
- Arceneaux, Kevin. 2002. "Direct Democracy and the Link between Public Opinion and State Abortion Policy." *State Politics & Policy Quarterly* 2(4): 372–87.
- Baer, Denise. 1995. "Contemporary Strategy and Agenda Setting." In *Campaigns and Elections American Style*, ed. James A. Thurber and Candice J. Nelson. Boulder, CO: Westview Press.
- Baker, Tod A., Robert P. Steed & Laurence W. Moreland. 1991. "Preachers and Politics: Jesse Jackson, Pat Robertson and the 1988 Presidential Nomination Campaign in South Carolina" in *The Bible And the Ballot Box: Religion and Politics in the 1988 Election*. Ed. Guth, James L. and John C. Green. Westview Press: Boulder, CO.
- Bartels, Larry M. 1992. "The Impact of Electioneering in the United States." In *Electioneering: A Comparative Study of Continuity and Change*, ed. David Butler and Austin Ranney. Oxford: Clarendon Press.
- Bartels, Larry M. 1993. "Messages Received: The Political Impact of Media Exposure." American Political Science Review 87: 267-85.

- Bartels, Larry M. 2000. "Partisanship and Voting Behavior, 1952-1996." *American Journal of Political Science* 44(1): 35-50.
- Bartels, Larry M. 2003. "Democracy with Attitudes." In *Electoral Democracy*, ed. Michael MacKuen and George Rabinowitz. Ann Arbor: University of Michigan Press, 48-82.
- Baum, Lawrence. 1988. "Measuring Policy Change in the US Supreme Court." *American Political Science Review* 82(3): 905-12.
- Bell, Derrick A.Jr., 1978. "The Referendum: Democracy's Barrier to Racial Equality." Washington Law Review 54.
- Berelson, Bernard R., Paul F. Lazarsfeld, and William N. McPhee. 1954. *Voting*. Chicago: University of Chicago Press.
- Berry, Williams D., Evan J, Ringquist, Richard C. Fording and Russell L. Hanson. 1998."Measuring Citizen and Government Ideology in the American States." *American Journal of Political Science* 42(1): 327-48.

Blumberg, Paul. 1990. "The Politics of Ignorance." Unpub. paper, Queens College.

- Bowler, Shaun, and Todd Donovan. 2002. "Democracy, Institutions and Attitudes about Citizen Influence on Government." *British Journal of Political Science* 32:371-390.
- Bowler, Shaun, Gary Segura, and Stephen Nicholson. 2006. "Earthquakes and Aftershocks: Race, Direct Democracy, and Partisan Change." *American Journal of Political Science* 50: 146-159.

- Brace Paul and Brent D. Boyea. 2008. "State Public Opinion, the Death Penalty, and the Practice of Electing Judges." *American Journal of Political Science* 52(2): 360-372.
- Brace, Paul, Laura Langer and Melinda G. Hall. 2000. "Measuring The Preferences of State Supreme Court Judges." *Journal of Politics* 62(2): 376-413.
- Brewer, M. B. 1979. In-group bias in the minimal inter-group situation: A cognitive motivational analysis. *Psychological Bulletin*, *86*, 307–324.
- Britt, Chester L. 1998. "Race, Religion, and Support for the Death Penalty: A Research Note." Justice Quarterly 15(1): 175 – 191.
- Broder, David S. 2000. *Democracy Derailed: Initiative Campaigns and the Power of Money*. New York: Harcourt.
- Brody, Richard A., and Lawrence S. Rothenberg. 1988. "The Instability of Partisanship: An Analysis of the 1980 Presidential Election." *British Journal of Political Science* 18:445-65.
- Burden, Barry C. 2004. "An Alternative Account of the 2004 Presidential Election." *The Forum* 2(4). Available at: http://www.bepress.com/forum/vol2/iss4/art2 [Accessed September 30, 2009].
- Burden, Barry C. 2005. "Institutions and Policy Representation in the States." *State Politics & Policy Quarterly* 5(4): 373–93.
- Burnham, Walter Dean. 1989. "The Reagan Heritage." In *The Election of 1988: Reports and Interpretations*, ed. Gerald M. Pomper et al. Chatham, N.J.: Chatham House.

- Caldeira, Gregory A. 1991. "Courts and Public Opinion." In American Courts: A Critical Assessment, eds. John B. Gates and Charles A. Johnson. Washington:Congressional Quarterly, Inc.
- Calfano, Brian R. and Paul A. Djupe. 2009. "God Talk: Religious Cues and Electoral Support." *Political Research Quarterly* 62(2): 329-39.
- Campbell, Angus, Phillip E. Converse, Warren E. Miller and Donald E. Stokes. 1960. *The American Voter*. New York: Wiley.
- Campbell, David E., and J. Quin Monson. 2008. "The Religion Card: Gay Marriage and the 2004 Presidential Election." *Public Opinion Quarterly* 72(3):399-419.
- Cassel, Carol A. 1993. "A Test of Converse's Theory of Party Support." *The Journal of Politics* 55(3): 664-681.
- Cassel, Carol A. 1999. "Voluntary associations, churches, and social participation theories of turnout." *Social Science Quarterly* 80: 504-514.
- Claassen, Ryan L., and Andrew Povtak. 2010. "The Christian Right Thesis: Explaining Longitudinal Change in Participation among Evangelical Christians." *Journal of Politics* 72:2-15.
- Claggett, William. 1981. "Partisan Acquisition Versus Partisan Intensity: Life-Cycle, Generation, and Period Effects, 1952-1976." American Journal of Political Science 25(2): 193-214.
- Clark, Paul and Marick F. Masters. 2001. "Competing Interest Groups and Union Members' Voting." *Social Science Quarterly* 38:104-23.

- Committee on Political Parties, American Political Science Association. 1950. "Toward a More Responsible Two-Party System." *American Political Science Review* 44(3, Part 2, Supplement): v-96.
- Converse, Philip E. 1964. "The Nature of Belief Systems in Mass Publics." In *Ideology and Discontent*, David E. Apter, ed. New York: The Free Press.
- Converse, Philip E. 1969. "Of Time and Partisan Stability." *Comparative Political Studies* 2(2): 139 -171.
- Converse, Philip E. 1976. *The Dynamics of Party Support: Cohort-Analyzing Party Identification*. Beverly Hills: Sage.
- Cook, Elizabeth Adell, Ted G. Jelen, and Clyde Wilcox. 1992. *Between Two Absolutes: Public Opinion and the Politics of Abortion*. Boulder, CO: Westview.
- Dahl, Robert A. 1956. *A Preface to Democratic Theory*. Chicago: The University of Chicago Press.
- Dahl, Robert A. 1998. On Democracy. New Haven, CT: Yale University Press.
- Dao, James. 2004. "Flush with Victory, grass-roots crusader against same-sex marriage think big." *New York Times*, November 26, A-28.
- Darmofal, David. 2005. "Elite Cues and Citizen Disagreement with Expert Opinion." *Political Research Quarterly* 58(3): 381-395.
- Delli Carpini, Michael X., and Scott Keeter. 1997. What Americans Know about Politics and Why it Matters. New Haven, CT: Yale University Press.
- DeNardo, James. 1980. "Turnout and the vote: the joke's on the Democrats." *American Political Science Review* 74:406-20.

- Devins, Neal, and Nicole Mansker. 2010. "Public Opinion and State Supreme Courts." University of Pennsylvania Journal of Constitutional Law 13: 455.
- DiMaggio, Paul, John H. Evans, and Bethany Bryson. 1996. "Have Americans' Social Attitudes Become More Polarized?" American Journal of Sociology 102: 690-755.
- Domke, David and Kevin Coe. 2008. *The God Strategy*. Oxford : Oxford University Press.
- Donovan, Todd, Caroline J. Tolbert and Daniel A. Smith. 2008. "Priming Presidential Votes by Direct Democracy." *Journal of Politics* 70: 1217-1231.
- Dougherty, Kevin D., Byron R. Johnson, and Edward C. Polson. 2007. "Recovering the Lost: Remeasuring U.S. Religious Affiliation." *Journal for the Scientific Study of Religion* 46: 483-99.
- Downs, Anthony. 1957. An Economic Theory of Democracy. New York: Harper & Row.
- Druckman, Daniel.1994. "Nationalism, Patriotism, and Group Loyalty: A Social Phscyhological Perspective." *Mershon International Studies Review* 38: 43-68.
- Egan, Patrick J. 2013. Partisan Priorities: How Issue Ownership Drives and Distorts American Politics. Cambridge University Press.
- Ensley, Michael J. and Erik P. Bucy. 2010. "Do Candidate Positions Matter? The Effect of the Gay Marriage Question on Gubernatorial Elections." *American Politics Research* 38(1): 142-164.
- Erikson, Robert S. 1972. "Malapportionment, Gerrymandering, and Party Fortunes in Congressional Elections." *American Political Science Review* 66:1234-45.

- Erikson, Robert S., Michael B. MacKuen, and James A. Stimson. 2002. *The Macro Polity*. Cambridge: Cambridge University Press.
- Erickson, Robert S., Gerald C. Wright and John P. McIver. 1993. Statehouse Democracy:
 Public Opinion and Policy in the American States. New York: Cambridge
 University Press.
- Epstein, Lee, Valerie Hoekstra, Jeffry A. Segal and Harold J. Spaeth. 1998. "Do Political Preferences Change? A Longitudinal Study of U.S. Supreme Court Justices." *Journal of Politics* 60(3): 801-18.
- Epstein, Lee, Andrew Martin, Kevin M. Quinn and Jeffery Segal. 2007. "Ideological Drift among Supreme Court Justices: Who, When and How Important?" *Northwestern University Law Review* 101(1): 1483-1541.
- Evans, John H. 2003. "Have Americans' Attitudes Become More Polarized? An Update." Social Science Quarterly 84: 71-90.
- Fastnow, Chris, J. Tobin Grant, and Thomas J. Rudolph. 1999. "Holy roll calls: Religious tradition and voting behavior in the U.S. House. *Social Science Quarterly* 80: 687-701.
- Fenno, Richard F.1978. *Home Style: House Members in their Districts*. New York: Little, Brown.
- Finkel, Steven E. 1993. "Reexamining the 'Minimal Effects' Model in Recent Presidential Elections." *Journal of Politics* 55:1-21.
- Fiorina, Morris P. 1981. *Retrospective Voting in American National Elections*. New Haven: Yale University Press.
- Fiorina, Morris P. 2010. *Culture War? The Myth of a Polarized America*, 3rd ed. New York: Pearson Longman.
- Flemming, Roy B. and B. Dan Wood. 1997. "The Public and the Supreme Court: Individual Justice Responsiveness to American Policy Moods." *American Journal* of Political Science 41(2) 468-98.
- Fowler, Robert Booth, Allen D. Hertzke, Laura R. Olson and Kevin R. den Dulk. 2010.
 Religion and Politics in America: Faith, Culture, and Strategic Choices, 4th ed.
 Boulder, CO: Westview Press.
- Franklin, Charles H., and John E. Jackson. 1983. "The Dynamics of Party Identification." *American Political Science Review* 77(4): 957-973.
- Friedman, Lawrence M. 1998. *American Law: An Introduction*. 2nd ed. New York: W. W. Norton & Company.
- Frisch, Morton J. 1985. Selected Writings and Speeches of Alexander Hamilton.Washington, D.C.: American Enterprise Institute.
- Gallup, George. 1938. "Testing Public Opinion." Public Opinion Quarterly. Special Supplement: Public Opinion in a Democrac. 2(January):8-14.
- Gallup, George and Rae Saul F. 1940. *The Pulse of Democracy*. New York: Simon and Schuster.
- Gay, David A., Christopher G. Ellison, and Daniel A. Powers. 1996. "In Search of Denominational Subcultures: Religious Affiliation and 'Pro-Family' Issues Revisited." *Review of Religious Research* 38: 3-17.

- Gerber, A. S. Green, D. P. Shachar, R. 2003. "Voting May Be Habit-Forming: Evidence from a Randomized Field Experiment." American Journal of Political Science 47(3): 540-550.
- Gibson, James L. 2008. "Challenges to the Impartiality of State Supreme Courts: Legitimacy Theory and "New-Style" Judicial Campaigns." *American Journal of Political Science* 102(1): 59-75.
- Giles, Michael W., Bethany Blackstone and Richard L. Vining, Jr. 2008. "The Supreme Court in American Democracy: Unraveling the Linkages between Public Opinion and Judicial Decision Making." *Journal of Politics* 70(2): 293-306.
- Gorski, Philip S. and Ates Altinordu. 2008. "After Secularization?" Annual Review of Sociology 34: 55-85.
- Green, Donald Philip, and Bradley Palmquist. 1990. "Of Artifacts and Partisan Instability." *American Journal of Political Science* 34(3): 872-902.
- Green, Donald, Bradley Palmquist, and Eric Schickler. 2002. Partisan Hearts and Minds: Political Parties and the Social Identities of Voters. New Haven: Yale University Press.
- Green, John C. 2007. *The Faith Factor: How Religion Influences American Elections*. Westport, Conn: Praeger Publishers.

Green, John C., James L. Guth & Cleveland R. Fraser. 1991. "Apostles and Apostates?
Religion and Politics Among Party Activists" in in *The Bible And the Ballot Box: Religion and Politics in the 1988 Election*. Ed. Guth, James L. and John C. Green.
Westview Press: Boulder, CO.

- Green, John C., James L. Guth, Corwin E. Smidt, and Lyman A. Kellstedt, ed. 1996.
 Religion and the Culture Wars: Dispatches from the Front. Lanham, MD:
 Rowman and Littlefield.
- Greene, Steven. 1999. "Understanding Party Identification: A Social Identity Approach." *Political Psychology* 20(2): 393-403.
- Greene, Steven. 2004. "Social Identity Theory and Party Identification." *Social Science Quarterly* 85(1): 136-153.
- Griffin, John D., and Brian Newman. 2005. "Are voters better represented?" *Journal of Politics* 67 (4): 1206–27.
- Guth, James L., Lyman A. Kellstedt, John C. Green and Corwin E. Smidt. 2002. "A
 Distant Thunder? Religion Mobilization in the 2000 Elections." In *Interest Group Politics*. 6th ed., ed. Allan J. Cigler and Burdett A. Loomis, 157-181. Washington D.C.
- Haider-Markel, Donald P., and Mark Joslyn. 1996. "The Politics of Gay and Lesbian Rights: Expanding the Scope of the Conflict." *Journal of Politics* 58:332-349.
- Haider-Markel, Donald P., Alana Querze, and Kara Lindaman. 2007. "Lose, Win or Draw? A Reexamination of Direct Democracy and Minority Rights." *Political Research Quarterly* 60: 304-314.
- Hall, Melinda G. 2001. "State Supreme Courts in American Democracy: Probing the Myths of Judicial Reform." *American Political Science Review* 95(2): 315-330.
- Harris, Fredrick C. 1995. "Something Within: Religion as a Mobilizer of African-American Political Activism." *Journal of Politics* 56(1):42-68.

Harris, Fredrick C. 1999. Something Within: Religion in African-American Political Activism. New York: Oxford University Press.

Haynie, Stacia L. 2003. Judging in Black and White. New York: Peter Lang.

- Hetherington, Marc J. 2001. "Resurgent Mass Partisanship: The Role of Elite Polarization." *American Political Science Review* 95(3): 619-631.
- Hill, David B., and Carol A. Cassel. 1983. "Comment on Abramson and Aldrich." *American Political Science Review* 77:1011-12.
- Hill, Kim Quaile, Jan E. Leighley, and Angela Hinton-Andersson. 1995. "Lower-class mobilization and policy linkage in the U.S. states." *American Journal of Political Science* 39 (1): 75–86.
- Hillygus, D. Sunshine, and Todd G. Shields. 2005. "Moral Issues and Voter Decision Making in the 2004 Presidential Election." PS: Political Science & Politics 38(02):201-209.

Holbrook, Thomas W. 1996. Do Campaigns Matter? Thousand Oaks, Calif: Sage.

- Holbrook, Thomas M., and Scott D. McClurg. 2005. "The Mobilization of Core Supporters: Campaigns, Turnout, and Electoral Composition in United States Presidential Elections." *American Journal of Political Science* 49(4): 689-703.
- Hougland, J.G. and J. A. Christenson. 1983. "Religion and Politics: The Relationship of Religious Participation to Political Efficacy and Involvement." *Sociology and Social Research* 67:405-420.
- Huber, Gregory A. and Sanford C. Gordon. 2004. "Accountability and Coercion: Is Justice Blind When it Runs for Office?" *American Journal of Political Science* 48(2): 247-63.

- Hurley, Patricia A. and Kim Quaile Hill. 2003. "Beyond the Demand-Input Model: A Theory of Representational Linkages." *Journal of Politics* 65(2): 304-26.
- Huckfeldt, Robert. 1979. "Political Participation and the Neighborhood Context." American *Journal of Political Science* 23 (3): 579-591.
- Huckfeldt, Robert. 1983. "Social Contexts, Social Networks, and Urban Neighborhoods: Environmental Constraints on Friendship Choice." *American Journal of Sociology* 89 (3): 651-669.
- Huckfeldt, Robert. 1984. "Political Loyalties and Social Class Ties: The Mechanisms of Contextual Influence." *American Journal of Political Science* 28 (2): 399-417.
- Huckfeldt, Robert. 2007. "Information, Persuasion, and Political Communication
 Networks." In *The Oxford Handbook of Political Behavior*, Russell J. Dalton and
 Hans-Dieter Klingemann, ed. Oxford: Oxford University Press.
- Huckfeldt, Robert, Paul A. Beck, Russell J. Dalton, and Jeffrey Levine. 1995. "Political Environments, Cohesive Social Groups, and the Communication of Public Opinion." *American Journal of Political Science* 39 (4): 1025-1054.
- Huckfeldt, Robert, Paul A. Beck, Rusell J. Dalton, Jeffrey Levine, and William Morgan.
 1998. "Ambiguity, Distorted Messages, and Nested Environmental Effects on Political Communication." *Journal of Politics* 60 (4): 996-1020.
- Huckfeldt, Robert, Paul E. Johnson, and John Sprague. 2004. Political Disagreement: The Survival of Diverse Opinions within Communication Networks. Cambridge: Cambridge University Press.

- Huckfeldt, Robert, Eric Plutzer, and John Sprague. 1993. "Alternative Contexts of Political Behavior: Churches, Neighborhoods, and Individuals." *Journal of Politics* 55:365-381.
- Huckfeldt, Robert, and John Sprague. 1995. Citizens, Politics, and Social
 Communication: Information and Influence in an Election Campaign. Cambridge,
 UK: Cambridge University Press.
- Huddy, Leonie. 2001. "From Social to Political Identity: A Critical Examination of Social Identity Theory." *Political Psychology* 22(1): 127-56.
- Hughes, D. L. and Charles W. Peek. 1986. "Ladies Against Women: Explaining the Political Participation of Traditional and Modern-Role Females." *Political Behavior* 8:158-169.
- Hunter, James Davison. 1991. *Culture Wars: The Struggle to Redefine America*. New York: Basic Books.
- Initiatives and Referendum Institute. 2000. *Election Post Election Report*. Washington, DC: Citizen Lawmaker Press.
- Iyengar, Shanto, and Donald Kinder. 1987. *News that Matters*. Chicago: Chicago University Press.
- Jelen, Ted G. 1993. The Political World of Clergy. Westport, CT: Praeger.
- Jerit, Jennifer, Jason Barabas, and Toby Bolsen. 2006. "Citizens, Knowledge, and the Information Environment." *American Journal of Political Science* 50(2): 266-282.
- Jones-Correa, Michael A., and David L. Leal. 2001. "Political Participation: Does Religion Matter?" *Political Research Quarterly* 54:751-770.

Keith, Bruce E. 1992. The Myth of the Independent voter. University of California Press.

- Kelly, Caroline. 1988. "Intergroup Differentiation in a Political Context." *British Journal* of Social Psychology 27: 319-32.
- Kelly, Caroline. 1989. "Political Identity and Perceived Intragroup Homogeneity." *British Journal of Social Psychology* 28: 239-250.
- Kelly, Caroline. 1990a. "Social Identity and Intergroup Perceptions in Minority-Majority Contexts."*Human Relations* 43: 583-599.
- Kelly, Caroline. 1990b. "Social Identity and Levels of Influence: When a political minority fails." *British Journal of Social Psychology* 29: 289-301.
- Kenny, Chris B. 1992. "Political Participation and Effects from the Social Environment." *American Journal of Political Science* 36 (1): 259-267.
- Key, V.O. 1942. Politics, Parties and Pressure Groups. New York: Thomas Y. Crowell.
- Key, V.O. 1949. Southern Politics in State and Nation. New York: Knopf.
- Key, V.O. 1961. Public Opinion and American Democracy. New York: Knopf.
- Kohut, Andrew, John C. Green, Scott Keeter, and Robert Toth. 2000. *The Diminishing Divide: Religion's Changing Role in American Politics*. Washington D.C.:
 Brookings Institution Press.
- Krosnick, J. A. 1990. "Government policy and citizen passion: A study of issue publics in contemporary America." *Political Behavior*, 12, 59-92.
- Kuklinski, James H., Paul J. Quirk, Jennifer Jerit, David Schwieder, and Robert Rich.
 2000. "Misinformation and the Currency of Citizenship." *Journal of Politics* 62: 790-816.

- Lau, Richard R., and David P. Redlawsk. 2001. "Advantages and Disadvantages of Cognitive Heuristics in Political Decision Making." *American Journal of Political Science* 45(4): 951-971.
- Layman, Geoffrey, C. 1999. "Culture Wars' in the American Party System: Religious and Cultural Change among Partisan Activists since 1972." *American Politics Quarterly* 27: 89-121.
- Layman, Geoffrey C. 2001. *The Great Divide: Religious and Cultural Conflict in American Party Politics*. New York: Columbia University Press.
- Layman, Geoffrey C. 2010. "Religion and Party Activists: A "Perfect Storm" of Polarization or a Recipe for Pragmatism?" In Alan Wolfe and Ira Katznelson, eds. *Religion and Democracy in the United States: Danger or Opportunity?* Princeton, NJ/Thousand Oaks, CA: Princeton University Press and Russell Sage, pp. 212-54.
- Layman,Geoffrey C., and John C. Green. 2006. "Wars and Rumours of Wars: The Contexts of Cultural Conflict in American Political Behaviour." *British Journal of Political Science* 36(1):61-89.
- Leege, David C. and Lyman A. Kellstedt. 1993. *Rediscovering the Religious Factor in American Politics*. New York: M. E. Sharpe.
- Leip, David. Dave Leip's Atlas of U.S. Presidential Elections. http://www.uselectionatlas.org (accessed on November 1, 2009).
- Levine, J., Carmines E., and Huckfeldt, R. 1997. "The Rise of Ideology in the Post-New Deal Party System, 1972-92." *American Politics Quarterly* 25: 19-34.
- Lewis, Gregory B. 2005. "Same-Sex Marriage and the 2004 Presidential Election." *PS: Political Science & Politics* 38(02):195-199.

- Lewis-Beck, Michael S., William G. Jacoby, Helmut Norpoth and Herbert F. Weisberg. 2008. *The American Voter Revisited*. Ann Arbor: University of Michigan Press.
- Lippman, Walter. 1922. *Public Opinion*. New York: Harcourt, Brace and Company.
 Lodge, Milton, Kathleen McGraw, and Patrick Stroh. 1989. "An ImpressionDriven Model of Candidate Evaluation." *American Political Science Review* 83(2): 399-419.
- Long, J. Scott and Jeremy Freese. 2006. *Regression Models for Categorical Dependent Variables Using Stata, Second Edition*. College Station, TX: Stata Press.
- Lupia, Arthur. 1994. "Shortcuts Versus Encyclopedias: Information and Voting Behavior in California Insurance Reform Elections." *American Political Science Review* 88(1): 63-76.
- Lupia, Arthur, and Mathew D. McCubbins. 1998. *The Democratic Dilemma: Can Citizens Learn What They Need to Know?* Cambridge: Cambridge University Press.
- Luskin, Robert C. 1987. "Measuring Political Sophistication." *American Journal of Political Science* 31: 856-99.
- Macaluso, Theodore F., and John Wanat. 1979. "Voting Turnout Religiosity." *Polity* 12:158-169.
- MacKuen, Michael B., Robert S. Erickson and James A. Stimson. 1989 "Macropartisanship" *American Political Science Review* 83(Dec): 1125-42.
- Markus, Gregory B., and Philip E. Converse. 1979. "A Dynamic Simultaneous Equation Model of Electoral Choice." *American Political Science Review* 73:1055-70.

- Mayhew, Professor David R. 1974. *Congress: The Electoral Connection*. New Haven: Yale University Press.
- Mcclurg, Scott D. 2003. "Social Networks and Political Participation: The Role of Social Interaction in Explaining Political Participation." *Political Research Quarterly* 56(4): 449–64.
- McGuire, Kevin T. and James A. Stimson. 2004. "The Least Dangerous Branch Revisited: New Evidence on Supreme Court Responsiveness to Public Opinion." *Journal of Politics* 66(4) 1018-1035.
- Milbrath, Lester W., and M. L. Goel. 1977. *Political Participation*. 2nd ed. Chicago: Rand McNally.
- Miller, Warren E. and Martin P. Watenberg. 1983. "Measuring Party Identification: Independent or No Partisan Preference?" *American Journal of Political Science* 27:106-21.
- Mishler, William and Reginald S. Sheehan. 1993. "The Supreme Court as a Countermajoritarian Institution? The Impact of Public Opinion on Supreme Court Decisions." *American Political Science Review* 87(1): 87-101.
- Murphy, Walter F. 1964. *Elements of Judicial Strategy*. Chicago: University of Chicago Press.
- Mutz, Diana C. 1998. Impersonal Influence: How Perceptions of Mass Collectives Affect Political Attitudes. Cambridge: Cambridge University Press.
- Mutz, Diana C., and Jeffery J. Mondak. 2006. "The Workplace as a Context for Cross-Cutting Political Discourse." *Journal of Politics* 68(1): 140-155.

- Nicholson, Stephen. P. 2005. Voting the Agenda: Candidates, Elections, and Ballot Propositions. Princeton, NJ: Princeton University Press.
- Nie, Norman H., and Kristi Andersen. 1974. "Mass Belief Systems Revisited: Political Change and Attitude Structure." *Journal of Politics* 36(3): 540-591.
- Niemi, Richard G. 1986. "The Dynamics of Public Opinion." In *Political Science: The Science of Politics*, ed. Herbert F. Weisberg. New York: Agathon.
- Niemi, Richard G., G. Bingham Powell, Jr., Harold W. Stanley, and C. Lawrence Evans. 1985. "Testing the Converse Partisanship Model with New Electorates." *Comparative Political Studies* 18:300-22.
- Norpoth, Helmut and Jerrold C. Rusk. 1982. "Partisan Dealignment in the American Electorate: Itemizing the Deductions since 1964." *American Political Science Review* 76: 522-37.

Padover, Saul K. 1939. Democracy. New York: The Greenwood Press.

- Page, Benjamin I., and Calvin C. Jones. 1979. "Reciprocal Effects of Policy Preferences, Party Loyalties and the Vote." *American Political Science Review* 73(4): 1071-1089.
- Pateman, Carole. 1970. *Participation and Democratic Theory*. Cambridge: Cambridge University Press.
- Peterson, Steven. 1990. Political Behavior: Patterns in Everyday Life. Newbury Park, CA: Sage.
- Petrocik, John R. 1996. "Issue Ownership in Presidential Elections, with a 1980 Case Study." *American Journal of Political Science* 40:825-850.

- Pew Forum on Religion & Public Life. 2008. U.S. Religious Landscape Survey. http://religions.pewforum.org/>.
- Pitkin, Hannah F. 1967. *The Concept of Representation*. Berkeley: The University of California Press.
- Prior, Markus. 2007. Post-Broadcast Democracy: How Media Choice Increases Inequality in Political Involvement and Polarizes Elections. Cambridge: Cambridge University Press.
- Putnam, Robert D. 1995. "Bowling Alone: America's Declining Social Capital." *Journal* of Democracy 6:65-78.
- Putnam, Robert D. 2000. *Bowling Alone: The Collapse and Revival of American Community*. New York: Simon Schuster.
- Putnam, Robert D. and David E. Campbell. 2010. *American Grace: How Religion Divides and Unites Us.* New York: Simon & Schuster Press.
- Rahn, Wendy M. 1993. "The Role of Partisan Stereotypes in Information Processing about Political Candidates." *American Journal of Political Science* 37:472-496.
- Ranney, Austin. 1972. "Turnout and Representation in Presidential Primary Elections." *The American Political Science Review* 66(1): 21-37.
- Rapoport, Ronald B., Walter J. Stone and Alan I. Abramowitz. 1991. "Do Endorsements Matter? Group Influence in the 1984 Democratic Caucuses." *American Political Science Review* 85: 193-203.
- Regnerus, Mark D., David Sikkink, and Christian Smith. 1999. "Voting with the Christian Right: Contextual and Individual Patterns of Electoral Influence." *Social Forces* 77(4): 1375–1401.

- Riker, William H. 1982. Liberalism against Populism: A Confrontation between the Theory of Democracy and the Theory of Social Choice. Prospect Heights, IL: Waveland Press.
- Romer, Daniel, Kate Kenski, Kenneth Winneg, Christopher Adasiewicz, and Kathleen Hall Jamieson. 2006. *Capturing Campaign Dynamics, 2000 and 2004: The National Annenberg Election Survey*. University of Pennsylvania Press.
- Rosenberg, Gerald N. 1991. *The Hallow Hope: Can Courts Bring About Social Change?* Chicago: University of Chicago Press.
- Rosenstone, Steven J., and John Mark Hansen. 1993. *Mobilization, Participation, and Democracy in America*. New York: Longman.

Schattschneider, E. E. 1959. Party Government. New York: Rinehart.

- Schattschneider, E.E. 1960. *The Semi-Sovereign People: A Realist's View of Democracy in America*. New York: Holt, Rinehart.
- Schumpeter, Joseph A. 1942. *Capitalism, Socialism and Democracy*. New York: Harper and Brothers.
- Segal, Jeffery A., and Harold J. Spaeth. 1993. Supreme Court and the Attitudinal Model. New York: Cambridge University Press.
- Segal, Jeffery A., and Harold J. Spaeth. 2002. *Supreme Court and the Attitudinal Model Revisited*. Cambridge: Cambridge University Press.
- Shaw, Daron R. 1999. "A Study of Presidential Campaign Event Effects from 1952 to 1992." *Journal of Politics* 61: 387-422.
- Shively, W. Phillips. 1972. "Voting Stability and the Nature of Party Attachments in the Weimar Re public." *American Political Science Review* 66:1203-25.

- Shively, W. Phillips. 1990. *The Craft of Political Research*. 3d ed. Englewood Cliffs, NJ: Prentice-Hall.
- Smidt, Corwin E. 1999. "Religion and Civic Engagement: A Comparative Analysis." Annals of the American Academy of Political Science 565:176-192
- Smidt, Corwin E. 2003. Religion as Social Capital: Producing the Common Good. Waco, Texas: Baylor University Press.
- Smidt, Corwin E., John C. Green, Lyman Kellstedt, and James L. Guth. 2007. "What Does the Lord Require? Evangelicals and the 2004 Presidential Vote." In *Religion and the Bush Presidency*, Mark J. Rozell and Gleaves Whitney, eds., 31-47. New York: Palgrave Macmillan.
- Smidt, Corwin E., Lyman A. Kellstedt, and James L Guth. 2009. "The Role of Religion in American Politics: Explanatory Theories and Associated Analytical and Measurement Issues." In *The Oxford Handbook of Religion and American Politics*, Smidt, Corwin E., Lyman A. Kellstedt, and James L Guth, ed. Oxford: Oxford University Press.
- Smith, Daniel A., Matthew DeSantis, and Jason Kassel. 2006. "Same-Sex Marriage Ballot Measures and the 2004 Presidential Election." *State & Local Government Review* 38(2):78 91.
- Smith, Daniel A, and C. Tolbert. 2001. "The Initiatives to Party: Partisanship and Ballot Initiatives in California," *Party Politics* 85:1177-1192.
- Smith, Daniel A, and C. Tolbert. 2007. "The Instrumental and Educative Effects of Ballot Measures: Research on Direct Democracy in the American States." *State Politics and Policy Quarterly* 7(4):416–45.

- Smith, Daniel A., Caroline J. Tolbert and Daniel C. Bowen. 2008. "The Educative Effects of Direct Democracy: A Research Primer for Legal Scholars." University of Colorado Law Review 78:101-124.
- Smith, Lauren E. and Laura R. Olson. 2013. "Attitudes about Socio moral issues Among Religious and Secular Youth." *Politics and Religion Journal* 7(2): 285-314.
- Smith, Lauren E., Laura R. Olson and Jeffery A. Fine. 2010. "Substantive Religious Representation in the U.S. Senate." *Political Research Quarterly* 63(1): 68-82.
- Smith, Lauren E., and Lee Demetrius Walker. 2013. "Belonging, Believing, and Group Behavior Religiosity and Voting in American Presidential Elections." *Political Research Quarterly* 66(2): 399–413.
- Smith, Mark A. 2001. "The Contingent Effects of Ballot Initiatives and Candidates Races on Turnout." *Journal of Political Science* 45:700-706
- Songer, Donald R.and Kelley A. Crews-Meyer. 2000. "Does Judge Gender Matter? Decision Making in State Supreme Courts." *Social Science Quarterly* 81(3): 750-762.
- Songer, Donald R. and Susan J. Tabrizi. 2000. "The Religious Right in Court: The Decision Making of Christian Evangelicals in State Supreme Courts." *Journal of Politics* 61(2): 507-26.
- Steensland, Brian, Jerry Z. Park, Mark D. Regnerus, Lynn D. Robinson, W. Bradford
 Wilcox, and Robert D. Woodberry. 2000. "The Measure of American Religion
 Towards Improving the State of the Art." *Social Forces* 79:291-318.

- Tajfel, Henri. 1978. Social categorization, social identity, and social comparisons. In Henri Tajfel ed., *Differentiation between social groups* (pp. 27-60). London: Academic Press.
- Tajfel, Henri. 1981. *Human groups and social categories*. Cambridge: Cambridge University Press.
- Tate, Neal. 1981. "Personal Attribute Model of the Voting Behavior of US Supreme Court Justices." *American Journal of Political Science* 75: 355-67.

Teixeira, Ruy A. 1992. The disappearing American voter. Brookings Institution Press.

- The Pew Forum of Religion and Life Death Penalty Survey. 2009. Accessed on April 2, 2010. Accessed at http://pewforum.org/Death-Penalty/An-Enduring-Majority-Americans-Continue-to-Support-the-Death-Penalty.aspx
- Tilley, James R. 2003. "Party Identification in Britain: Does Length of Time in the Electorate Affect Strength of Partisanship?." *British Journal of Political Science* 33(2): 332-344.
- Tocqueville, Alexis de. [1835-40] 1969. *Democracy in America*. Edited by J.P Mayer and translated by George Lawrence. Garden City, NY: Doubleday.
- Tolbret, Catherine J., John A. Gremmel, and Daniel E. Smith. 2001. "The Effects of Ballot Initiatives on Voter Turnout in the American States." *American Politics Research* 29:625-648.
- Truman, David B. 1951. *The Governmental Process: Political Interests and Public Opinion*. New York: Alfred A. Knopf.

- Turner, J. C., Hogg, M. A., Oakes, P. J., Reicher, S. D., & Wetherell, M. 1987. *Rediscovering the social group: A self categorization theory*. New York: Blackwell.
- Ulmer, Sidney S. 1973. "The longitudinal Behavior of Hugo Lafayette Black:Parabolic Support for Civil Liberities, 1937-1971." *Florida State University Law Review* 1(1): 131-53.
- Verba, Sidney, Kay Lehman Schlozman and Henry E. Brady. 1995. Voice and Equality: Civic Voluntarism in American Politics. Cambridge, MA: Harvard University Press.
- Wald, Kenneth D and Allison Calhoun-Brown. 2011. *Religion and Politics in the United States, 6th ed.* New York: Rowman and Littlefield Publishers.
- Wald, Kenneth D., Dennis E. Owen, and Samuel S. Hill Jr. 1988. "Churches as Political Communities." American Political Science Review 82: 531-548.
- Wald, Kenneth D., Dennis E. Owen, and Samuel S. Hill Jr. 1990. "Political Cohesion in Churches." *Journal of Politics* 52: 197-2 15.
- Walsh, Katherine Cramer. 2004. *Talking About Politics: Informal Groups and Social Identity in American Life*. Chicago: University of Chicago Press.
- Wattenberg, Martin P. 1996. *The Decline of American Political Parties: 1952-1994*.Cambridge: Harvard University Press.
- Weisberg, Herbert F. 1980. "A Multidimensional Conceptualization of Party Identification." *Political Behavior* 2(1): 33-60.
- Widner, Jennifer A. 2001. *Building the Rule of Law*. New York: W.W. Norton & Company.

Wielhouwer, Peter W. 2009. "Religion and American Political Participation." In *The* Oxford Handbook of Religion and American Politics eds. Corwin E. Smidt,

Lyman A. Kellstedt and James L. Guth. Oxford: Oxford University Press.

- Wilcox, Clyde, and Carin Larson. 2006. *Onward Christian soldiers? The Religious Right in American politics*. 3rd ed. Boulder, CO: Westview.
- Wilcox, Clyde and Carin Robinson. 2011. Onward Christian Soldiers?: The Religious Right in American Politics. Boulder, CO: Westview Press.
- Wilson, Bertha. 1990. "Will Women Judges Really Make a Difference?" Osgood Hall Law Journal 507.
- Wolfinger, Raymond E., and Steven J. Rosenstone. 1980. *Who Votes?* New Haven, CT: Yale University Press.
- Wood, Gordon S. 1969. The Creation of the American Republic 1776-1787. Chapel Hill, NC: The University of North Carolina Press.
- Wood, Gordon S. 1991. *The Radicalism of the American Revolution*. New York: Vintage Books.
- Wuthnow, Robert. 1999. Mobilizing Civic Engagement: The Changing Impact of
 Religious Involvement. In *Civic Engagement in American Democracy*, edited by
 Theda Skocpol and Morris P. Fiornia. Washington D.C." Brookings Institution
 Press.
- Young, Robret L. 1992. "Religious Orientation, Race and Support for the Death Penalty." Journal for the Scientific Study of Religion 31(1): 76-87.
- Zaller, John R. 1992. *The Nature and Origins of Mass Opinion*. Cambridge: Cambridge University Press.

Zorn, Christopher. 2006. "Comparing GEE and Robust Standard Errors for Conditionally Dependent Data." *Political Research Quarterly* 59(3): 329-41.